ATTACHMENT A: PROFESSIONAL SERVICES CONTRACT

Purchase Requisition Number:	
Contract Number:	
Vendor Number:	

MILWAUKEE BOARD OF SCHOOL DIRECTORS PROFESSIONAL SERVICES CONTRACT MODIFICATION AND EXTENSION AGREEMENT

THIS MODIFICATION AND EXTENSION AGREEMENT ("Extension Agreement") is made as of the 28th day of October, 2021, by and between MILWAUKEE BOARD OF SCHOOL DIRECTORS ON BEHALF OF MILWAUKEE PUBLIC SCHOOLS (hereinafter "MPS") and CONSTELLATION NEWENERGY – GAS DIVISION, LLC (as successor by merger with Constellation Energy Services Natural Gas, LLC (hereinafter "Supplier").

WHEREAS, on May 4, 2016, MPS and Supplier entered into Professional Services Contract with a term of July 1, 2016 through June 30, 2019 ("Contract"); and

WHEREAS, on April 12, 2019, MPS and Supplier modified and extended the Contract for a single additional three-year term, from July 1, 2019 through June 30, 2022 ("Contract"); and

WHEREAS, the Contract consisted of (1) MPS' Professional Services Contract ("PSC"), (2) the Supplier's NAESB Base Contract for Sale and Purchase of Natural Gas ("NAESB Agreement"), and (3) Supplier's Retail Special Provisions to the NAESB Base Contract for Sale and Purchase of Natural Gas ("Special Provisions"); and

WHEREAS, Section 2 of the Contract provides for up to three additional one-year extensions subject to funding and an annual performance review; and

WHEREAS, the parties wish to modify and extend the Contract for a single twenty-one month term, from July1, 2022 through March 31, 2024, under the same terms and conditions as set forth in the original Contract, except for those specifically modified by this Extension Agreement;

NOW, THEREFORE, the parties hereto agree as follows (all capitalized terms used but not defined herein shall have the meaning set forth in the Contract):

1. Section 2 of the Contract is deleted in its entirety and replaced with the following:

"Subject to Section 12 of the NAESB Agreement, the Term of this Contract shall be July 1, 2022 through March 31, 2024.

No work shall commence before a Supplier receives a fully executed Contract and has been given approval to proceed. Any work performed by the Supplier prior to obtaining a fully-executed contract with approval to proceed shall not be compensated pursuant to this contract. Any continuation of the contract beyond this term must be set forth in writing and signed by authorized signatories on behalf of the Party for which it is executed."

2. Section 7 of the Contract is deleted in its entirety and replaced with the following:

"Supplier understands and agrees that financial responsibility for claims or damages to any person, or to Supplier's employees and agents, shall rest with the Supplier. Supplier and its subSuppliers

Purchase Requisition Number:	
Contract Number:	
Vendor Number:	

shall effect and maintain any insurance coverage, including, but not limited to, Workers' Compensation, Employers' Liability, General Liability, Contractual Liability, Professional Liability, Automobile Liability and Umbrella Liability to support such financial obligations. The indemnification obligation, however, shall not be reduced in any way by existence or non-existence, limitation, amount or type of damages, compensation, or benefits payable under Workers' Compensation laws or other insurance provisions.

The minimum limits of insurance required of the Supplier by MPS shall be:

Workers' Compensation Statutory Limits

Employers' Liability \$100,000 per occurrence

General Liability \$1,000,000 per occurrence/\$2,000,000 aggregate

Professional Liability \$1,000,000 per occurrence Auto Liability \$1,000,000 per occurrence Umbrella (excess) Liability \$1,000,000 per occurrence

The Milwaukee Board of School Directors shall be named as an additional insured under Supplier's and subSuppliers' general liability insurance and umbrella liability insurance. Evidence of all required insurances of Supplier shall be given to MPS. The certificate of insurance or policies of insurance evidencing all coverages shall include a statement that MPS shall be afforded a thirty (30) day written notice of cancellation, non-renewal or material change by any of Supplier's insurers providing the coverages required by MPS for the duration of this contract."

- 3. The parties hereby extend the Term of the Contract, as modified by this Extension Agreement, for a term of July 1, 2022 through March 31, 2024 ("Extension Term").
- 4. The NAESB Agreement to be in effect during the Extension Term shall be the NAESB Agreement attached hereto as Appendix A and incorporated into the Contract by reference ("NAESB Agreement in Extension Term").
- 5. The Special Provisions to be in effect during the Extension Term shall be the Special Provisions attached hereto as Appendix B and incorporated into the Contract by reference ("Special Provisions in Extension Term").
- 6. The Contract Quantity per year is 857,700 dekatherms during the Extension Term and is estimated not to exceed \$7,000,000 per year without Board approval and is based on historical usage and consideration of worst case scenarios. MPS shall be responsible for payment of actual volumes which may be higher or lower based on actual use by MPS and will be purchased or sold at prevailing market rates. All invoicing and payment obligations during the Extension Term shall be governed by the NAESB Agreement in Extension Term and Special Provisions in Extension Term. The total cost of transportation, daily balancing and basis during the Extension Term is NYMEX plus \$1.02 per dekatherm (Dth).

Purchase Requisition Number:	
Contract Number:	
Vendor Number:	

IN WITNESS WHEREOF, the parties hereto have executed this Extension Agreement on the date first written above.

Milwaukee Public Schools
Dated:
By:
(Principal or Administrator Signature)
By:
Keith P. Posley, Ed.D.
Superintendent of Schools
By:
Robert E. Peterson, President
Milwaukee Board of School Directors



Account Manager:

Sherron, Lori

(667) 313-1941

lori.sherron@constellation.com

DEAL NO. TBD

	Transaction Confirmation
for Sale and Purchase	mation dated is made a part of, and pursuant to terms of, the Base Contract of Gas dated 5/4/2016 between Buyer and Seller identified herein (the"NAESB").
Special Condition: This with the Professional S	Confirmation is not effective and binding upon the parties hereto unless signed concurrently Services Contract Modification and Extension Agreement dated effective
Trade Date:	TBD
Buyer:	Milwaukee Board of School Directors on behalf of Milwaukee Public Schools
<u>Seller</u> :	Constellation NewEnergy - Gas Division, LLC (CNEGAS)
Facility Name:	Milwaukee Public Schools (inc) Attachment A
Delivery Period:	7/1/2022 - 6/30/2025 (inclusive) $3/31/24$
Nature of Obligaton:	7300 MMBtu/day Firm with balance as Secondary Firm
	"Firm" means deliveries and receipts may not be interrupted without liability except for reasons of Force Majeure regardless of the capacity type Constellation uses to receive such gas.
	"Secondary Firm" means deliveries and receipts will be on a best-efforts basis up to Customer's maximum daily quantity and performance may be interrupted without liability to the extent that one or more of the following conditions are present: (i) Force Majeure; (ii) curtailment by the local distribution company owning and/or controlling and maintaining the distribution system required for delivery of gas to the Facility(ies) (the "Utility"); (iii) curtailment of supply by a natural gas supplier; (iv) curtailment of storage by a storage provider; (v) curtailment of transportation by a gas gathering or pipeline company, or Utility (each a "Transporter"), transporting gas for CNEG or Customer downstream or upstream of the Delivery Point(s), including, but not limited to, transportation between secondary firm points; (vi) recall of transportation capacity release by its releaser; or (vii) curtailment of gas production behind a specific meter.
Deal Type:	Physical Basis
Contract Quantity/Pri	ice: NYMEX Last Day Settle ("NYM LDS") plus Market Basis per dekatherm. Either or both of the price components may be fixed for specific quantities and delivery months with mutual agreement of the Parties. The Basis is fixed at per dekatherm.

© 2018 Constellation Energy Resources, LLC. The offerings described herein are those of either Constellation NewEnergy-Gas Division, LLC or Constellation NewEnergy, Inc., affiliates of each other and ultimate subsidiaries of Exelon Corporation. Brand names and product names are trademarks or service marks of their respective holders. All rights reserved. Errors and omissions excepted.

Page 1 of 5



An Exelon Company

	Year 2022	Year 2023	Year 2024	Year 2025
Contract Month	Fixed Amount (MMBtu)	Fixed Amount (MMBtu)	Fixed Amount (MMBtu)	Fixed Amount (MMBtu)
January	(MINIDO)	155.000.0	155,000.0	455,000.0
February		430.000:0	130,000.0	130,000.0
March		110.000.0	110,000.0	110,000.07
April		-70.000.0	~70,000.0	70,000.0
		30.000.0	30.000.0	30,000.0
May		6,000.0	6,000.0	6,000.0
July	4.000.0	4.000.0	4,000.0	
August	4.000.0	4,000.0	4,000.0	
September	7.500.0	-7,500.0	7,500.0	
October	40.000.0	40,000.0	40,000.07	
November	1		100,000.0	
December	130,000.0	1.30,000.0	130,000.0	

Volume: 1 MMBtu=10 THERMS

Price: Per MMBtu price / 10=per THERM price

Jul - 5600	Apr	64,700
Aug - 5100	May	28,000
Sep-8500	Tim	9,000
1.6 - 46,000	50.	
Nav - 95,800		
Du 162,000		
Tan 165,000		
Feb 149,000		
Mar 117,000		

The price referenced herein is inclusive of fuel to the delivery point.

Buyer acknowledges and understands that, in connection with this transaction, Seller may be paying a fee to a third party. The fee, if any, would be included in Buyer's price and would be charged by a broker/agent representing Buyer, a broker/agent representing Seller, or an association/referrer who helped facilitate this transaction.

The price referenced herein represents basis only and does not include any applicable commodity price. For the fixed quantities and Delivery Period shown herein, if Customer has not otherwise locked the commodity price for a specific month, the commodity price shall be the NYMEX last day settle price for that month.

Incremental Pricing: For all additional quantities required by Customer, in excess of the fixed quantities listed herein in combination with any other nominated quantities, or for gas quantities bought back from Customer by Constellation due to Customer's requirements being less than the total nominated volumes each month, the price for such additional purchases or buybacks shall be based on the then current market price, as determined by Constellation in its reasonable discretion.

If Constellation and Customer have previously entered into any commodity transaction(s) for the Facility (the "Previous Transaction") that overlap one or more of the delivery months covered in this Transaction Confirmation (the "Overlapping Month(s)"), the incremental gas pricing in this Transaction Confirmation shall apply to the Overlapping Months instead of any conflicting incremental gas pricing associated with the Previous Transaction.

<u>Allocation:</u> Constellation will allocate the Contract Quantity listed herein among the Customer's separate facilities or meters at a later date.

Pipeline: ANR

ANR Contract(s): 114007, 134880, 13877, 134890

Delivery Point(s): MILWAUKEE

Utility: WG

Attachment A is attached and made part of this Transaction Confirmation.

1. A Balancing Service charge of \$0.00 per dekatherm applies in addition to the Price.



An Exelon Company

Buyer and Seller shall use commercially reasonable efforts to provide each other with sufficient notice to allow timely and accurate nomination in accordance with the nomination requirements of that Transporter, as they may change from time to time due to Constraint Day requirements or otherwise. Buyer accepts the following responsibilities: Buyer hereby appoints Seller as its agent for the purposes of balancing, including the receipt of current and historic usage, storage, billing and transportation data from transporter. Buyer agrees to assist Seller in the timely collection of data directly and through Transporter, notify Seller of usage expectations and any material changes to those usage expectations, and comply with any daily limitations concerning use. If usage data is unavailable electronically, Buyer shall obtain usage data manually and deliver that information weekly via facsimile or electronic mail or as requested by Seller.

- 2. Seller agrees to monitor and review all information provided to the Seller and Buyer and Transporter regarding Buyer's gas use. Based on such information, Seller shall use commercially reasonable effort to keep Buyer within the balancing parameters established by Transporter. Seller shall pass through to Buyer and Buyer shall pay for any imbalance charges and/or cashout resulting from Buyer's failure to (i) fulfill its obligations set forth in paragraph 1 above, (ii) maintain telemetry, and/or (iii) meet daily limitations with respect to use.
- 3. Mid-month variations to delivered quantity. In the event that the quantity of natural gas that can be delivered to Buyer (without the Transporter assessing imbalance charges or cashout) is less than the quantity agreed to be purchased for the relevant period, Buyer shall remain liable for the entire quantity It agreed to purchase; however, Seller will provide a credit at prevailing market rates for that portion of the purchased quantity that is not delivered. In the event that Buyer's expected use will exceed the quantity of gas purchased by Buyer and/or available to Buyer without incurring imbalance charges for the relevant period, Seller will use commercially reasonable efforts to provide incremental supply, plus fuel (if applicable) at prevailing market rates. Credits for reductions in delivered quantity and/or charges for incremental supply shall appear as separate line items on Seller's invoice to Buyer, including applicable price and quantity.
- 4. Telemetry Obligations. Buyer acknowledges and agrees that it shall be responsible for installing and maintaining a meter telephone line. Buyer shall also be responsible for any charges imposed by a contractor for repairing the meter telephone line. As a convenience to Buyer, Seller may arrange for telemetry service or repairs to be provided and Buyer agrees it shall be responsible for any costs associated with such service and repairs.

"Constraint Day" means any period of time during which a Transporter issues balancing order, operational now orders, operational matching orders (or the equivalent), or winter splits or deems the day a Critical Gas Day (or the equivalent) and thereby restricts deliveries by seller or use by Buyer, requires a specific delivery by Seller or use by Buyer, or otherwise requires an affirmative action by either Party due to Adverse operational conditions experienced by the Transporter.

"Daily Limitations" means Transporter enforced restrictions or mandates concerning quantities of gas delivered and/or used, including those imposed due to Constraint Day conditions and/or under ordinary operating conditions.

- Invoices. Invoices shall be prepared based upon quantities used.
- 6. Consolidated Billing: Seller shall include the Transporter distribution fees on the monthly invoice. Past due and other charges due Transporter, that apply to service prior to the effective date of the Confirmation will not be included in Seller's invoice to Buyer.

All other terms and conditions of this Transaction Confirmation shall remain unchanged.

© 2018 Constellation Energy Resources, LLC. The offerings described herein are those of either Constellation NewEnergy-Gas Division, LLC or Constellation NewEnergy, Inc., affiliates of each other and ultimate subsidiaries of Exelon Corporation. Brand names and product names are trademarks or service marks of their respective holders. All rights reserved. Errors and omissions excepted. DEAL NO. TBD Page 3 of 5



Default Service: To To the extent the facility continues to receive gas from Constellation beyond the Delivery Period, all such gas will be considered month to month purchases and will be priced at the Market Price.

To the extent that Seller is able to continue to provide Firm level of service after the initial Delivery Period, at commercially reasonable terms, Seller will offer such service to Buyer subject to an agreement by both Parties on pricing, term, credit provisions, and other provisions.

(SIGNATURE BLOCKS FOLLOW ON NEXT PAGE)



Please return via the DocuSign E-Signature process or email to constellation Constellation Constellation Constellation Constellation Constellation Constellation Constellation Constellation Constellation Confirmations@Constellation.com.

Milwaukee Board of School Directors on behalf of Milwaukee Public Schools

Ву:

Name:

Nancy H. Fischer

Title:

Vice President- Retail Ops

Date:

By:

Name:

Title:

Date:

Customer ID PA-229686 Contract No. Deal No. TBD

PR#:	
Contract Number:	
RFP Number (if applicable):	NATURAL GAS

MILWAUKEE BOARD OF SCHOOL DIRECTORS PROFESSIONAL SERVICES CONTRACT

This Professional Services Contract is being entered into this 4th day of May, 2016, by and between Constellation Energy Services – Natural Gas, LLC ("Supplier") and Milwaukee Board of School Directors ("MPS"). The "Contract" between the parties consists of this Professional Services Contract ("PSC"), Supplier's NAESB Base Contract for Sale and Purchase of Natural Gas dated May 4, 2016 (hereinafter "NAESB Agreement") and Supplier's Retail Special Provisions to the NAESB Base Contract for Sale and Purchase of Natural Gas (hereinafter "Special Provisions").

1. SCOPE OF SERVICES

Supplier shall sell, and MPS shall purchase, natural gas.

MPS solicited proposals via Request for Proposal ("RFP"). The RFP and Supplier's response to the RFP dated March 30, 2016 are incorporated herein by reference.

Supplier shall provide, at its own expense, all personnel required to perform the services under this Contract. All gas purchases and sales shall be governed by the NAESB Agreement.

2. TERM

Subject to Section 12 of the NAESB Agreement, the Term of this Contract shall be July 1, 2016 through June 30, 2019, with three possible, one-year extensions subject to funding and an annual performance review.

No work shall commence before a Supplier receives a fully executed Contract and has been given approval to proceed. Any work performed by the Supplier prior to obtaining a fully-executed contract with approval to proceed shall not be compensated pursuant to this contract. Any continuation of the contract beyond this term must be set forth in writing and signed by the original signatories to the contract.

3. COMPENSATION

All invoicing and payment obligations shall be governed by the NAESB Agreement and Special Provisions, and consistent with Supplier's response to the RFP dated March 30, 2016. The total cost of transportation, daily balancing and basis is NYMEX plus \$0.785 per dekatherm (Dth). The Contract Quantity per year is 857,700 dekatherms for the Term of this Contract and is estimated not to exceed \$6,000,000 per year without board approval and is based on historical usage and consideration of worst case scenarios. MPS shall be responsible for payment of actual volumes which may be higher or lower based on actual use by MPS and will be purchased or sold at prevailing market rates.

4. NON APPROPRIATION OF FUNDS

This Contract is contingent upon the appropriation of sufficient funds by appropriate MPS officials. In the event that MPS fails to appropriate monies with respect to natural gas service, the Contract shall be terminated with notice from MPS to Supplier and all terminated Riders or Transactions shall be treated as Terminated Transactions arising under an Event of Default and Supplier shall be deemed to be the Non-Defaulting Party for purposes of the termination payment calculated in accordance with Section 10.3.1 of the NAESB Agreement.

5. NON-DISCRIMINATION

In the performance of work under this Contract, Supplier shall not discriminate in any way against any employee or applicant for employment because of race, religion, color, national origin, ancestry, physical handicap, mental disability, medical condition, marital status, age (over 40) or sex. This prohibition includes but is not be limited to employment; promotions, demotions and transfers; recruitment; advertising; layoff or termination; rates of pay or other forms of compensation; and selection for training, including apprenticeships. Supplier is required to include a similar provision in all subcontracts to this contract. For purposes of this Contract, Supplier, as a subsidiary of Exelon Corporation, is subject to Exelon Corporation's Equal Opportunity/Affirmative Action program and written policy.

If MPS determines Supplier has violated this non-discrimination policy, MPS may terminate this Contract without liability for undelivered services or materials. MPS may also deem the Supplier ineligible to participate in future contracts with MPS.

6. BACKGROUND CHECKS

A criminal information background check is required for all persons providing services under this Contract, including volunteers, that: (1) provide services in MPS facility(ies) on a regular and ongoing basis or more than 5 hours per week; and (2) come into contact with or have access to MPS students with or without the presence of an MPS teacher or MPS supervisor.

The purpose of this check is to ensure there is nothing that would render the person(s) unfit to perform services under this Contract where there is contact and or access to MPS students. MPS will, in its sole discretion, determine whether there is anything in a background check that would render a person unfit to work in an MPS facility with contact or access to MPS students. MPS shall perform background checks in the state(s) in which the individual resided for at least 6 months in the last 5 years, and was 18 years old or older at the time.

Supplier may perform its own criminal background checks through the Wisconsin Department of Justice Crime Information Bureau ("CIB"). Supplier shall provide the completed criminal background checks at least 10 days prior to any services being performed pursuant to this contract.

MPS will perform the necessary background investigation at the rate of \$10.00 per person. In the event Supplier chooses this option, Supplier may contact the Office of Employment Relations at 475-8280 to obtain the necessary forms. Please note that all forms must be filled out and submitted at least 30 days prior to the commencement of the services.

All background checks must be completed prior to the commencement of services under this contract. MPS will NOT be responsible for the payment of any services rendered by Supplier before the completion of these criminal information background checks.

7. INSURANCE AND PROOF OF FINANCIAL RESPONSIBILITY

Supplier understands and agrees that financial responsibility for claims or damages to any person, or to Supplier's employees and agents, shall rest with the Supplier. Supplier and its subSuppliers shall effect and maintain any insurance coverage, including, but not limited to, Workers' Compensation, Employers' Liability, General Liability, Contractual Liability, Professional Liability, Automobile Liability and Umbrella Liability to support such financial obligations. The indemnification obligation, however, shall not be reduced in any way by existence or non-existence, limitation, amount or type of damages, compensation, or benefits payable under Workers' Compensation laws or other insurance provisions.

The minimum limits of insurance required of the Supplier by MPS shall be:

Workers' Compensation
Employers' Liability
General Liability
Professional Liability
Auto Liability
S1,000,000 per occurrence
4,000,000 per occurrence
\$1,000,000 per occurrence
\$1,000,000 per occurrence
\$1,000,000 per occurrence
\$1,000,000 per occurrence
Umbrella (excess) Liability
\$1,000,000 per occurrence

MPS shall be named as an additional insured under Supplier's and subSuppliers' general liability insurance and umbrella liability insurance. Evidence of all required insurances of Supplier shall be given to MPS. The certificate of insurance or policies of insurance evidencing all coverages shall include a statement that MPS shall be afforded a thirty (30) day written notice of cancellation, non-renewal or material change by any of Supplier's insurers providing the coverages required by MPS for the duration of this contract.

8. IRREPARABLE HARM

It is mutually agreed the breach of this Contract on Supplier's part shall result in irreparable and continuing damage to MPS for which money damages may not provide adequate relief. Therefore, the breach of this Contract on Supplier's part shall entitle MPS to both preliminary and permanent injunctive relief and money damages insofar as they can be determined under the circumstances.

9. BREACH BY A PARTY

A failure of either party to fulfill its material obligations under this Contract shall constitute an Event of Default under the NAESB Agreement. If either party experiences an Event of Default under the NAESB Agreement, such Event of Default shall be a failure to fulfill its material obligations under this Contract. Parties agree that the process for termination for cause shall be governed by the Event of Default provision of the NAESB Agreement.

10. TERMINATION BY MPS

MPS further reserves the right to terminate this Contract at any time for any reason by giving Supplier written notice by Registered or Certified Mail of such termination. MPS will attempt to give Supplier 30 days' notice, but reserves the right to give

immediate notice. In the event of said termination, Supplier shall reduce its activities hereunder, as mutually agreed to, upon receipt of said notice. Upon said termination, Supplier shall be paid for all services rendered through the date of termination, including any early termination penalties pursuant to Section 10.3.1 of the NAESB Agreement if less than 30 days' notice is provided.

11. INDEPENDENT SUPPLIER

Supplier agrees and stipulates that in performing this Contract, it is acting as an Independent Supplier, and that no relationship of employer and employee, partnership or joint venture is created by this Contract. Supplier has exclusive control over work hours, location, and other details of such services, and MPS' sole interest is to ensure that said service shall be performed and rendered in a competent, safe, efficient, timely and satisfactory manner in accordance with the terms of this Contract.

Supplier has the sole obligation to provide for and pay any contribution or taxes required by federal, state or local authorities imposed on or measured by income. Supplier specifically covenant not to file any complaint, charge, or claim with any local, state or federal agency or court in which Supplier claims to be or to have been an employee of MPS during the period of time covered by this Contract and that if any such agency or court assumes jurisdiction of any complaint, charge or claim against MPS on Supplier's behalf, Supplier will request such agency or court to dismiss such matter. MPS shall not be charged any obligation or responsibility whatsoever of extending any fringe benefits which may be extended to MPS employees, including any insurance, or pension plans.

Supplier further agrees that MPS is not to be charged with the obligation or responsibility of extending any fringe benefits such as hospital, medical and life insurance, or pension plans which may be extended to employees of MPS from time-to-time.

12. ASSIGNMENT LIMITATION

This Contract shall be binding upon and inure to the benefit of the parties and their successors and assigns; provided, however, that neither party shall assign its obligations hereunder without the prior written consent of the other, such consent not to be unreasonably withheld, conditioned or delayed.

13. PROHIBITED PRACTICES

- A. Supplier during the period of this contract shall not hire, retain or use for compensation any member, officer, or employee of MPS to perform services under this Contract, or any other person who, to the knowledge of Supplier, has a conflict of interest.
- B. Supplier hereby attests it is familiar with MPS's Code of Ethics, providing in pertinent part, "[a]n employee of Milwaukee Public Schools may not accept any gift or gratuity in excess of \$25.00 annually from any person, persons, group or any firm which does business with or is attempting to do business with MPS."
- C. Supplier shall adhere to the MPS' Livable Wage Policy that requires all Suppliers to pay their employees a minimum of \$7.70 per hour.
- D. No current or former MPS employee may perform services on a professional services contract without the prior written consent of the Director of Human Relations of her designee.

14. NOTICES

Notices are provided for in Section 9 of the NAESB Agreement.

15. WAIVER

The waiver or failure of either Party to exercise in any respect any rights provided for in this Contract shall not be deemed a waiver of any further right under this Contract.

16. INTEGRATION/SEVERABILITY

This Contract, the NAESB Agreement and Special Provisions constitute the entire Contract among the Parties with respect to the subject matter hereof and supersede all prior proposals, negotiations, conversations, discussions and Contracts among the Parties concerning the subject matter hereof. No amendment or modification of any provision of this Contract shall be effective unless the same shall be in writing and signed by both Parties.

If any term or provision of this Contract should be declared invalid by a court of competent jurisdiction or by operation of law, the remaining terms and provisions of this Contract shall be interpreted as if such invalid Contracts or covenants were not contained herein.

17. ORDER OF PRIORITY

In the event of a conflict between the documents which make up this Contract, the terms of the NAESB Agreement or Special Provisions shall control. Specific exceptions to this paragraph are noted.

18. CHOICE OF LAW & FORUM

With respect to any conflict in the terms regarding choice of law, venue or construction of this Contract, the terms of the PSC shall control. The state and federal courts of Wisconsin shall be the sole forum for all disputes arising of this contract. Venue shall be the Milwaukee County Circuit Court or the Eastern District of Wisconsin. The validity, construction, enforcement and effect of this Contract shall be governed solely by the laws of the State of Wisconsin.

19. TIMING

Time is of the essence in this Contract.

20. CERTIFICATION REGARDING DEBARMENT OR SUSPENSION

Supplier certifies that neither Supplier or its principals; its subcontractors or their principals; the sub-recipients (if applicable) or their principals are suspended, debarred, proposed for debarment, voluntarily excluded from covered transactions, or otherwise disqualified by any federal department or agency from doing business with the Federal Government pursuant to Executive Orders 12549 and 12689. Supplier specifically covenants that neither the Supplier or its principals, its sub-Suppliers or their principals, or the sub-recipients (if applicable) or their principals are included on the Excluded Parties List System ("EPLS") maintained by the General Services Administration ("GSA").

21. STUDENT DATA

Supplier acknowledges that student data is protected by both federal and state law. See Wis. Stat. § 118.125; 20 U.S.C. § 1232g(b); 34 C.F.R. § 99.1 et seq. If MPS determines that Supplier has disclosed any student record information in violation of either federal or state law, without prejudice to any other rights or remedies the MPS may have, MPS shall be entitled to immediately terminate this and every other existing Contract without further liability. Moreover, MPS may bar Supplier from future MPS contracts for varying periods up to and including permanent debarment.

22. MPS LOGO/PUBLICITY

No Supplier shall use the MPS Logo in its literature or issue a press release about the subject of this Contract without prior written notice to and written approval of MPS' Director of Communication and Public Affairs.

23. PUBLIC RECORDS

Both parties understand that the Board is bound by the Wisconsin Public Records Law, and as such, all of the terms of this Contract are subject to and conditioned on the provisions of Wis. Stat. § 19.21, et seq. Supplier acknowledges that it is obligated to assist the Board in retaining and producing records that are subject to Wisconsin Public Records Law, and that the failure to do so shall constitute a material breach of this Contract, and that the Supplier must defend and hold the Board harmless from liability under the law. Except as otherwise authorized, those records shall be maintained for a period of seven years after receipt of final payment under this Contract.

24. HUB REQUIREMENT

The HUB requirement on this contract is 0%.

25. COUNTERPARTS

This Contract may be executed and delivered in counterparts (including facsimile transmission), each of which will be deemed an original and all of which together will constitute one and the same instrument. By way of clarification however, this Contract will not be effective unless both Parties have executed a counterpart.

26. AUTHORITY

By signing below, each individual additionally warrants that he or she is authorized to sign this Contract on behalf of the Party for which it was executed.

IN WITNESS WHEREOF, the parties here to have executed this Contract on the day, month and year first above written.

Constellation Energy Services - Natural Gas, LLC	Milwaukee Board Of School Directors
Dated: 6/6/6	Dated: 05-27-16
Of By: Dand I Donat	By: Marin Hammer (Principal or Administrator Signature)

Address: 9960 Corporate Campus Drive, Suite 2000 Louisville, KY 40223	By:
Phone Number: (502) 426-4500	Darienne B. Driver, Ed.D.
Tax Id or SS	Superintendent of Schools By:
Budget Code:	Mark Sain, President
1.5	Milwaukee Board of School Directors

Base Contract for Sale and Purchase of Natural Gas

This Base Contract is entered into as of the following date: May 4, 2016 ("Effective Date")

The parties to this Base Contract are the following:

PARTY A		PARTY B		
Constellation Energy Services – Natural Gas, LLC ("CESNG" or "Seller")	PARTY NAME	Milwaukee Board of School Directors on behalf of Milwaukee Public Schools ("Buyer")		
9960 Corporate Campus Drive, Suite 2000 Louisville, KY 40223	ADDRESS	5225 W. Vliet Street Milwaukee, WI 53208		
	BUSINESS WEBSITE			
CESNG19189	CONTRACT NUMBER			
	D-U-N-S® NUMBER			
□ US FEDERAL:	TAX ID NUMBERS	US FEDERAL:		
OTHER:	JURISDICTION OF	OTHER:		
Wisconsin	ORGANIZATION			
□ Corporation	COMPANY TYPE	☐ Corporation ☐ LLC ☐ Limited Partnership ☐ Partnership ☐ LLP ☐ Other:		
	GUARANTOR			
COL	(IF APPLICABLE) NTACT INFORMAT			
9960 Corporate Campus Drive, Suite 2000, Louisville, KY 40223 ATTN: Contracts Department TEL#: (502) 426-4500 FAX#: (502) 214-6381	COMMERCIAL (if different from Contract/Legal notices)	ATTN: TEL#: FAX#: EMAIL:		
9960 Corporate Campus Drive, Suite 2000, Louisville, KY 40223 ATTN: Contracts Department TEL#: (502) 426-4500 FAX#: (502) 214-6381	SCHEDULING (if different from Contract/Legal notices)	ATTN: TEL#: FAX#: EMAIL:		
9960 Corporate Campus Drive, Suite 2000, Louisville, KY 40223 ATTN: Contracts Department TEL#: (502) 426-4500 FAX#: (502) 214-6381	CONTRACT AND LEGAL NOTICES	1224 North 11 th Street, Milwaukee, WI 53233 ATTN: John Linn TEL#: (414)-283-4600 FAX#: (414) 283-4682 EMAIL: linnja@milwaukee.k12.wi.us		
9960 Corporate Campus Drive, Suite 2000, Louisville, KY 40223 ATTN: Contracts Department TEL#: (502) 426-4500 FAX#: (502) 214-6381	CREDIT (if different from Contract/Legal notices)	n/a ATTN: TEL#: FAX#: EMAIL:		
9960 Corporate Campus Drive, Suite 2000, Louisville, KY 40223 ATTN: Contracts Department TEL#: (502) 426-4500 FAX#: (502) 214-6381	TRANSACTION CONFIRMATIONS (if different from Contract/Legal notices)	ATTN: TEL#: FAX#: EMAIL:		
ACCOUNTING INFORMATION				
ATTN: see invoice TEL#: FAX#: EMAIL:	• INVOICES • PAYMENTS • SETTLEMENTS	ATTN: TEL#: FAX#: EMAIL:		
BANK: see invoice ABA:	WIRE TRANSFER NUMBERS (IF APPLICABLE)	BANK: N/A ABA: ACCT: OTHER DETAILS:		
BANK: see invoice ABA:	ACH NUMBERS (IF APPLICABLE)	BANK: N/A ABA: ACCT: OTHER DETAILS:		
ATTN: see invoice Address:	CHECKS (IF APPLICABLE)	ATTN: N/A ADDRESS:		

Base Contract for Sale and Purchase of Natural Gas

(Continued)

This Base Contract incorporates by reference for all purposes the General Terms and Conditions for Sale and Purchase of Natural Gas published by the North American Energy Standards Board. The parties hereby agree to the following provisions offered in said General Terms and Conditions. In the event the parties fail to check a box, the specified default provision shall apply. Select the appropriate box(es) from each section:

Transaction Procedure Section 2.7 Confirm Deadline Section 2.8 Confirming Party	OR	Oral (default) Written 2 Business Days after receipt (default) Business Days after receipt Seller (default) Buyer	Section 10.2 Additional Events of Default		No Additional Events of Default (default) Indebtedness Cross Default Party A: Party B: Transactional Cross Default Specified Transactions:
Performance	⊠ OR □	Cover Standard (default) Spot Price Standard	Section 10.3.1 Early Termination Damages	⊠ OR	Early Termination Damages Apply (default) Early Termination Damages Do Not Apply
Note: The following immediately preced Section 2.31 Spot Price Publication		Gas Daily Midpoint (default)	Section 10.3.2 Other Agreement Setoffs	⊠ OR	Other Agreement Setoffs Apply (default) Bilateral (default) Triangular
Section 6 Taxes	⊠ OR □	Buyer Pays At and After Delivery Point (default) Seller Pays Before and At Delivery Point	a		Other Agreement Setoffs Do Not Apply
Section 7.2 Payment Date	×	 25th Day of Month following Month of delivery (default) 10 days following date of invoice 	Section 15.5 Choice Of Law		Wisconsin
Section 7.2 Method of Payment		Wire transfer (default) Automated Clearinghouse Credit (ACH) Check	Section 15.10 Confidentiality	⊠ OR □	Confidentiality applies (default) Confidentiality does not apply
Section 7.7 Netting	OR	Netting applies (default) Netting does not apply umber of sheets attached 3			

IN WITNESS WHEREOF, the parties hereto have executed this Base Contract as of the Effective Date.

	CONSTELLATION ENERGY SERVICES - NATURAL GAS, LLC	PARTY NAME	MILWAUKEE BOARD OF SCHOOL DIRECTORS ON BEHALF OF MILWAUKEE PUBLIC SCHOOLS
W. 088	By: Dand J Donat	SIGNATURE	By: A DO
	David + Donat	PRINTED NAME	Dariegne B. Driver, Ed D.
	Vice President Retail Operations	TITLE	Superintendent of Schools
			By: That a. R.
			Mark A. Sain, President
			Milwaukee Board of School Directors

General Terms and Conditions Base Contract for Sale and Purchase of Natural Gas

SECTION 1. PURPOSE AND PROCEDURES

1.1. These General Terms and Conditions are intended to facilitate purchase and sale transactions of Gas on a Firm or Interruptible basis. "Buyer" refers to the party receiving Gas and "Seller" refers to the party delivering Gas. The entire agreement between the parties shall be the Contract as defined in Section 2.9.

The parties have selected either the "Oral Transaction Procedure" or the "Written Transaction Procedure" as indicated on the Base Contract.

Oral Transaction Procedure:

1.2. The parties will use the following Transaction Confirmation procedure. Any Gas purchase and sale transaction may be effectuated in an EDI transmission or telephone conversation with the offer and acceptance constituting the agreement of the parties. The parties shall be legally bound from the time they so agree to transaction terms and may each rely thereon. Any such transaction shall be considered a "writing" and to have been "signed". Notwithstanding the foregoing sentence, the parties agree that Confirming Party shall, and the other party may, confirm a telephonic transaction by sending the other party a Transaction Confirmation by facsimile, EDI or mutually agreeable electronic means within three Business Days of a transaction covered by this Section 1.2 (Oral Transaction Procedure) provided that the failure to send a Transaction Confirmation shall not invalidate the oral agreement of the parties. Confirming Party adopts its confirming letterhead, or the like, as its signature on any Transaction Confirmation as the identification and authentication of Confirming Party. If the Transaction Confirmation contains any provisions other than those relating to the commercial terms of the transaction (i.e., price, quantity, performance obligation, delivery point, period of delivery and/or transportation conditions), which modify or supplement the Base Contract or General Terms and Conditions of this Contract (e.g., arbitration or additional representations and warranties), such provisions shall not be deemed to be accepted pursuant to Section 1.3 but must be expressly agreed to by both parties; provided that the foregoing shall not invalidate any transaction agreed to by the parties.

Written Transaction Procedure:

- 1.2. The parties will use the following Transaction Confirmation procedure. Should the parties come to an agreement regarding a Gas purchase and sale transaction for a particular Delivery Period, the Confirming Party shall, and the other party may, record that agreement on a Transaction Confirmation and communicate such Transaction Confirmation by facsimile, EDI or mutually agreeable electronic means, to the other party by the close of the Business Day following the date of agreement. The parties acknowledge that their agreement will not be binding until the exchange of nonconflicting Transaction Confirmations or the passage of the Confirm Deadline without objection from the receiving party, as provided in Section 1.3.
- 1.3. If a sending party's Transaction Confirmation is materially different from the receiving party's understanding of the agreement referred to in Section 1.2, such receiving party shall notify the sending party via facsimile, EDI or mutually agreeable electronic means by the Confirm Deadline, unless such receiving party has previously sent a Transaction Confirmation to the sending party. The failure of the receiving party to so notify the sending party in writing by the Confirm Deadline constitutes the receiving party's agreement to the terms of the transaction described in the sending party's Transaction Confirmation. If there are any material differences between timely sent Transaction Confirmations governing the same transaction, then neither Transaction Confirmation shall be binding until or unless such differences are resolved including the use of any evidence that clearly resolves the differences in the Transaction Confirmations. In the event of a conflict among the terms of (i) a binding Transaction Confirmation pursuant to Section 1.2, (ii) the oral agreement of the parties which may be evidenced by a recorded conversation, where the parties have selected the Oral Transaction Procedure of the Base Contract, (iii) the Base Contract, and (iv) these General Terms and Conditions, the terms of the documents shall govern in the priority listed in this sentence.
- 1.4. The parties agree that each party may electronically record all telephone conversations with respect to this Contract between their respective employees, without any special or further notice to the other party. Each party shall obtain any necessary consent of its agents and employees to such recording. Where the parties have selected the Oral Transaction Procedure in Section 1.2 of the Base Contract, the parties agree not to contest the validity or enforceability of telephonic recordings entered into in accordance with the requirements of this Base Contract.

SECTION 2. DEFINITIONS

The terms set forth below shall have the meaning ascribed to them below. Other terms are also defined elsewhere in the Contract and shall have the meanings ascribed to them herein.

- 2.1. "Additional Event of Default" shall mean Transactional Cross Default or Indebtedness Cross Default, each as and if selected by the parties pursuant to the Base Contract.
- 2.2. "Affiliate" shall mean, in relation to any person, any entity controlled, directly or indirectly, by the person, any entity that controls, directly or indirectly, the person or any entity directly or indirectly under common control with the person. For this purpose, "control" of any entity or person means ownership of at least 50 percent of the voting power of the entity or person.
- 2.3. "Alternative Damages" shall mean such damages, expressed in dollars or dollars per MMBtu, as the parties shall agree upon in the Transaction Confirmation, in the event either Seller or Buyer fails to perform a Firm obligation to deliver Gas in the case of Seller or to receive Gas in the case of Buyer.

Page 1 of 10

- 2.4. "Base Contract" shall mean a contract executed by the parties that incorporates these General Terms and Conditions by reference; that specifies the agreed selections of provisions contained herein; and that sets forth other information required herein and any Special Provisions and addendum(s) as identified on page one.
- 2.5. "British thermal unit" or "Btu" shall mean the International BTU, which is also called the Btu (IT).
- 2.6. "Business Day(s)" shall mean Monday through Friday, excluding Federal Banking Holidays for transactions in the U.S.
- 2.7. "Confirm Deadline" shall mean 5:00 p.m. in the receiving party's time zone on the second Business Day following the Day a Transaction Confirmation is received or, if applicable, on the Business Day agreed to by the parties in the Base Contract; provided, if the Transaction Confirmation is time stamped after 5:00 p.m. in the receiving party's time zone, it shall be deemed received at the opening of the next Business Day.
- 2.8. "Confirming Party" shall mean the party designated in the Base Contract to prepare and forward Transaction Confirmations to the other party.
- 2.9. "Contract" shall mean the legally-binding relationship established by (i) the Base Contract, (ii) any and all binding Transaction Confirmations and (iii) where the parties have selected the Oral Transaction Procedure in Section 1.2 of the Base Contract, any and all transactions that the parties have entered into through an EDI transmission or by telephone, but that have not been confirmed in a binding Transaction Confirmation, all of which shall form a single integrated agreement between the parties.
- 2.10. "Contract Price" shall mean the amount expressed in U.S. Dollars per MMBtu to be paid by Buyer to Seller for the purchase of Gas as agreed to by the parties in a transaction.
- 2.11. "Contract Quantity" shall mean the quantity of Gas to be delivered and taken as agreed to by the parties in a transaction.
- 2.12. "Cover Standard", as referred to in Section 3.2, shall mean that if there is an unexcused failure to take or deliver any quantity of Gas pursuant to this Contract, then the performing party shall use commercially reasonable efforts to (i) if Buyer is the performing party, obtain Gas, (or an alternate fuel if elected by Buyer and replacement Gas is not available), or (ii) if Seller is the performing party, sell Gas, in either case, at a price reasonable for the delivery or production area, as applicable, consistent with: the amount of notice provided by the nonperforming party; the immediacy of the Buyer's Gas consumption needs or Seller's Gas sales requirements, as applicable; the quantities involved; and the anticipated length of failure by the nonperforming party.
- 2.13. "Credit Support Obligation(s)" shall mean any obligation(s) to provide or establish credit support for, or on behalf of, a party to this Contract such as cash, an irrevocable standby letter of credit, a margin agreement, a prepayment, a security interest in an asset, guaranty, or other good and sufficient security of a continuing nature.
- 2.14. "Day" shall mean a period of 24 consecutive hours, coextensive with a "day" as defined by the Receiving Transporter in a particular transaction.
- 2.15. "Delivery Period" shall be the period during which deliveries are to be made as agreed to by the parties in a transaction.
- 2.16. "Delivery Point(s)" shall mean such point(s) as are agreed to by the parties in a transaction.
- 2.17. "EDI" shall mean an electronic data interchange pursuant to an agreement entered into by the parties, specifically relating to the communication of Transaction Confirmations under this Contract.
- 2.18. "EFP" shall mean the purchase, sale or exchange of natural Gas as the "physical" side of an exchange for physical transaction involving gas futures contracts. EFP shall incorporate the meaning and remedies of "Firm", provided that a party's excuse for nonperformance of its obligations to deliver or receive Gas will be governed by the rules of the relevant futures exchange regulated under the Commodity Exchange Act.
- 2.19. "Firm" shall mean that either party may interrupt its performance without liability only to the extent that such performance is prevented for reasons of Force Majeure; provided, however, that during Force Majeure interruptions, the party invoking Force Majeure may be responsible for any Imbalance Charges as set forth in Section 4.3 related to its interruption after the nomination is made to the Transporter and until the change in deliveries and/or receipts is confirmed by the Transporter.
- 2.20. "Gas" shall mean any mixture of hydrocarbons and noncombustible gases in a gaseous state consisting primarily of methane.
- 2.21, "Guarantor" shall mean any entity that has provided a guaranty of the obligations of a party hereunder.
- 2.22. "Imbalance Charges" shall mean any fees, penalties, costs or charges (in cash or in kind) assessed by a Transporter for failure to satisfy the Transporter's balance and/or nomination requirements.
- 2.23. "Indebtedness Cross Default" shall mean if selected on the Base Contract by the parties with respect to a party, that it or its Guarantor, if any, experiences a default, or similar condition or event however therein defined, under one or more agreements or instruments, individually or collectively, relating to indebtedness (such indebtedness to include any obligation whether present or future, contingent or otherwise, as principal or surety or otherwise) for the payment or repayment of borrowed money in an aggregate amount greater than the threshold specified in the Base Contract with respect to such party or its Guarantor, if any, which results in such indebtedness becoming immediately due and payable.
- 2.24. "Interruptible" shall mean that either party may interrupt its performance at any time for any reason, whether or not caused by an event of Force Majeure, with no liability, except such interrupting party may be responsible for any Imbalance Charges as set forth in Section 4.3 related to its interruption after the nomination is made to the Transporter and until the change in deliveries and/or receipts is confirmed by Transporter.

Page 2 of 10

- 2.25. "MMBtu" shall mean one million British thermal units, which is equivalent to one dekatherm.
- 2.26. "Month" shall mean the period beginning on the first Day of the calendar month and ending immediately prior to the commencement of the first Day of the next calendar month.
- 2.27. "Payment Date" shall mean a date, as indicated on the Base Contract, on or before which payment is due Seller for Gas received by Buyer in the previous Month.
- 2.28. "Receiving Transporter" shall mean the Transporter receiving Gas at a Delivery Point, or absent such receiving Transporter, the Transporter delivering Gas at a Delivery Point.
- 2.29. "Scheduled Gas" shall mean the quantity of Gas confirmed by Transporter(s) for movement, transportation or management.
- 2.30. "Specified Transaction(s)" shall mean any other transaction or agreement between the parties for the purchase, sale or exchange of physical Gas, and any other transaction or agreement identified as a Specified Transaction under the Base Contract.
- 2.31. "Spot Price" as referred to in Section 3.2 shall mean the price listed in the publication indicated on the Base Contract, under the listing applicable to the geographic location closest in proximity to the Delivery Point(s) for the relevant Day; provided, if there is no single price published for such location for such Day, but there is published a range of prices, then the Spot Price shall be the average of such high and low prices. If no price or range of prices is published for such Day, then the Spot Price shall be the average of the following: (i) the price (determined as stated above) for the first Day for which a price or range of prices is published that next precedes the relevant Day; and (ii) the price (determined as stated above) for the first Day for which a price or range of prices is published that next follows the relevant Day.
- 2.32. "Transaction Confirmation" shall mean a document, similar to the form of Exhibit A, setting forth the terms of a transaction formed pursuant to Section 1 for a particular Delivery Period.
- 2.33. "Transactional Cross Default" shall mean if selected on the Base Contract by the parties with respect to a party, that it shall be in default, however therein defined, under any Specified Transaction.
- 2.34. "Termination Option" shall mean the option of either party to terminate a transaction in the event that the other party fails to perform a Firm obligation to deliver Gas in the case of Seller or to receive Gas in the case of Buyer for a designated number of days during a period as specified on the applicable Transaction Confirmation.
- 2.35. "Transporter(s)" shall mean all Gas gathering or pipeline companies, or local distribution companies, acting in the capacity of a transporter, transporting Gas for Seller or Buyer upstream or downstream, respectively, of the Delivery Point pursuant to a particular transaction.

SECTION 3. PERFORMANCE OBLIGATION

3.1. Seller agrees to sell and deliver, and Buyer agrees to receive and purchase, the Contract Quantity for a particular transaction in accordance with the terms of the Contract. Sales and purchases will be on a Firm or Interruptible basis, as agreed to by the parties in a transaction.

The parties have selected either the "Cover Standard" or the "Spot Price Standard" as indicated on the Base Contract.

Cover Standard:

The sole and exclusive remedy of the parties in the event of a breach of a Firm obligation to deliver or receive Gas shall be recovery of the following: (i) in the event of a breach by Seller on any Day(s), payment by Seller to Buyer in an amount equal to the positive difference, if any, between the purchase price paid by Buyer utilizing the Cover Standard and the Contract Price, adjusted for commercially reasonable differences in transportation costs to or from the Delivery Point(s), multiplied by the difference between the Contract Quantity and the quantity actually delivered by Seller for such Day(s) excluding any quantity for which no replacement is available; or (ii) in the event of a breach by Buyer on any Day(s), payment by Buyer to Seller in the amount equal to the positive difference, if any, between the Contract Price and the price received by Seller utilizing the Cover Standard for the resale of such Gas, adjusted for commercially reasonable differences in transportation costs to or from the Delivery Point(s), multiplied by the difference between the Contract Quantity and the quantity actually taken by Buyer for such Day(s) excluding any quantity for which no sale is available; and (iii) in the event that Buyer has used commercially reasonable efforts to replace the Gas or Seller has used commercially reasonable efforts to sell the Gas to a third party, and no such replacement or sale is available for all or any portion of the Contract Quantity of Gas, then in addition to (i) or (ii) above, as applicable, the sole and exclusive remedy of the performing party with respect to the Gas not replaced or sold shall be an amount equal to any unfavorable difference between the Contract Price and the Spot Price, adjusted for such transportation to the applicable Delivery Point, multiplied by the quantity of such Gas not replaced or sold. Imbalance Charges shall not be recovered under this Section 3.2, but Seller and/or Buyer shall be responsible for Imbalance Charges, if any, as provided in Section 4.3. The amount of such unfavorable difference shall be payable five Business Days after presentation of the performing party's invoice, which shall set forth the basis upon which such amount was calculated.

Spot Price Standard:

3.2. The sole and exclusive remedy of the parties in the event of a breach of a Firm obligation to deliver or receive Gas shall be recovery of the following: (i) in the event of a breach by Seller on any Day(s), payment by Seller to Buyer in an amount equal to the difference between the Contract Quantity and the actual quantity delivered by Seller and received by Buyer for such Day(s), multiplied by the positive difference, if any, obtained by subtracting the Contract Price from the Spot Price; or (ii) in the event of a breach by Buyer on any Day(s), payment by Buyer to Seller in an amount equal to the difference between the Contract Quantity and the actual quantity delivered by Seller and received by Buyer for such Day(s), multiplied by the positive difference, if any,

obtained by subtracting the applicable Spot Price from the Contract Price. Imbalance Charges shall not be recovered under this Section 3.2, but Seller and/or Buyer shall be responsible for Imbalance Charges, if any, as provided in Section 4.3. The amount of such unfavorable difference shall be payable five Business Days after presentation of the performing party's invoice, which shall set forth the basis upon which such amount was calculated.

- 3.3. Notwithstanding Section 3.2, the parties may agree to Alternative Damages in a Transaction Confirmation executed in writing by both parties.
- 3.4. In addition to Sections 3.2 and 3.3, the parties may provide for a Termination Option in a Transaction Confirmation executed in writing by both parties. The Transaction Confirmation containing the Termination Option will designate the length of nonperformance triggering the Termination Option and the procedures for exercise thereof, how damages for nonperformance will be compensated, and how liquidation costs will be calculated.

SECTION 4. TRANSPORTATION, NOMINATIONS, AND IMBALANCES

- 4.1. Seller shall have the sole responsibility for transporting the Gas to the Delivery Point(s). Buyer shall have the sole responsibility for transporting the Gas from the Delivery Point(s).
- 4.2. The parties shall coordinate their nomination activities, giving sufficient time to meet the deadlines of the affected Transporter(s). Each party shall give the other party timely prior Notice, sufficient to meet the requirements of all Transporter(s) involved in the transaction, of the quantities of Gas to be delivered and purchased each Day. Should either party become aware that actual deliveries at the Delivery Point(s) are greater or lesser than the Scheduled Gas, such party shall promptly notify the other party.
- 4.3. The parties shall use commercially reasonable efforts to avoid imposition of any Imbalance Charges. If Buyer or Seller receives an invoice from a Transporter that includes Imbalance Charges, the parties shall determine the validity as well as the cause of such Imbalance Charges. If the Imbalance Charges were incurred as a result of Buyer's receipt of quantities of Gas greater than or less than the Scheduled Gas, then Buyer shall pay for such Imbalance Charges or reimburse Seller for such Imbalance Charges paid by Seller. If the Imbalance Charges were incurred as a result of Seller's delivery of quantities of Gas greater than or less than the Scheduled Gas, then Seller shall pay for such Imbalance Charges or reimburse Buyer for such Imbalance Charges paid by Buyer.

SECTION 5. QUALITY AND MEASUREMENT

All Gas delivered by Seller shall meet the pressure, quality and heat content requirements of the Receiving Transporter. The unit of quantity measurement for purposes of this Contract shall be one MMBtu dry. Measurement of Gas quantities hereunder shall be in accordance with the established procedures of the Receiving Transporter.

SECTION 6. TAXES

The parties have selected either "Buyer Pays At and After Delivery Point" or "Seller Pays Before and At Delivery Point" as indicated on the Base Contract.

Buyer Pays At and After Delivery Point:

Seller shall pay or cause to be paid all taxes, fees, levies, penalties, licenses or charges imposed by any government authority ("Taxes") on or with respect to the Gas prior to the Delivery Point(s). Buyer shall pay or cause to be paid all Taxes on or with respect to the Gas at the Delivery Point(s) and all Taxes after the Delivery Point(s). If a party is required to remit or pay Taxes that are the other party's responsibility hereunder, the party responsible for such Taxes shall promptly reimburse the other party for such Taxes. Any party entitled to an exemption from any such Taxes or charges shall furnish the other party any necessary documentation thereof.

Seller Pays Before and At Delivery Point:

Seller shall pay or cause to be paid all taxes, fees, levies, penalties, licenses or charges imposed by any government authority ("Taxes") on or with respect to the Gas prior to the Delivery Point(s) and all Taxes at the Delivery Point(s). Buyer shall pay or cause to be paid all Taxes on or with respect to the Gas after the Delivery Point(s). If a party is required to remit or pay Taxes that are the other party's responsibility hereunder, the party responsible for such Taxes shall promptly reimburse the other party for such Taxes. Any party entitled to an exemption from any such Taxes or charges shall furnish the other party any necessary documentation thereof.

SECTION 7. BILLING, PAYMENT, AND AUDIT

- 7.1. Seller shall invoice Buyer for Gas delivered and received in the preceding Month and for any other applicable charges, providing supporting documentation acceptable in industry practice to support the amount charged. If the actual quantity delivered is not known by the billing date, billing will be prepared based on the quantity of Scheduled Gas. The invoiced quantity will then be adjusted to the actual quantity on the following Month's billing or as soon thereafter as actual delivery information is available.
- 7.2. Buyer shall remit the amount due under Section 7.1 in the manner specified in the Base Contract, in immediately available funds, on or before the later of the Payment Date or 10 Days after receipt of the invoice by Buyer; provided that if the Payment Date is not a Business Day, payment is due on the next Business Day following that date. In the event any payments are due Buyer hereunder, payment to Buyer shall be made in accordance with this Section 7.2.
- 7.3. In the event payments become due pursuant to Sections 3.2 or 3.3, the performing party may submit an invoice to the nonperforming party for an accelerated payment setting forth the basis upon which the invoiced amount was calculated. Payment from the nonperforming party will be due five Business Days after receipt of invoice.
- 7.4. If the invoiced party, in good faith, disputes the amount of any such invoice or any part thereof, such invoiced party will pay such amount as it concedes to be correct; provided, however, if the invoiced party disputes the amount due, it must provide supporting

Page 4 of 10

documentation acceptable in industry practice to support the amount paid or disputed without undue delay. In the event the parties are unable to resolve such dispute, either party may pursue any remedy available at law or in equity to enforce its rights pursuant to this Section.

- 7.5. If the invoiced party fails to remit the full amount payable when due, interest on the unpaid portion shall accrue from the date due until the date of payment at a rate equal to the lower of (i) the then-effective prime rate of interest published under "Money Rates" by The Wall Street Journal, plus two percent per annum; or (ii) the maximum applicable lawful interest rate.
- 7.6. A party shall have the right, at its own expense, upon reasonable Notice and at reasonable times, to examine and audit and to obtain copies of the relevant portion of the books, records, and telephone recordings of the other party only to the extent reasonably necessary to verify the accuracy of any statement, charge, payment, or computation made under the Contract. This right to examine, audit, and to obtain copies shall not be available with respect to proprietary information not directly relevant to transactions under this Contract. All invoices and billings shall be conclusively presumed final and accurate and all associated claims for under- or overpayments shall be deemed waived unless such invoices or billings are objected to in writing, with adequate explanation and/or documentation, within two years after the Month of Gas delivery. All retroactive adjustments under Section 7 shall be paid in full by the party owing payment within 30 Days of Notice and substantiation of such inaccuracy.
- 7.7. Unless the parties have elected on the Base Contract not to make this Section 7.7 applicable to this Contract, the parties shall net all undisputed amounts due and owing, and/or past due, arising under the Contract such that the party owing the greater amount shall make a single payment of the net amount to the other party in accordance with Section 7; provided that no payment required to be made pursuant to the terms of any Credit Support Obligation or pursuant to Section 7.3 shall be subject to netting under this Section. If the parties have executed a separate netting agreement, the terms and conditions therein shall prevail to the extent inconsistent herewith.

SECTION 8. TITLE, WARRANTY, AND INDEMNITY

- 8.1. Unless otherwise specifically agreed, title to the Gas shall pass from Seller to Buyer at the Delivery Point(s). Seller shall have responsibility for and assume any liability with respect to the Gas prior to its delivery to Buyer at the specified Delivery Point(s). Buyer shall have responsibility for and assume any liability with respect to said Gas after its delivery to Buyer at the Delivery Point(s).
- 8.2. Seller warrants that it will have the right to convey and will transfer good and merchantable title to all Gas sold hereunder and delivered by it to Buyer, free and clear of all liens, encumbrances, and claims. EXCEPT AS PROVIDED IN THIS SECTION 8.2 AND IN SECTION 15.8, ALL OTHER WARRANTIES, EXPRESS OR IMPLIED, INCLUDING ANY WARRANTY OF MERCHANTABILITY OR OF FITNESS FOR ANY PARTICULAR PURPOSE, ARE DISCLAIMED.
- 8.3. Seller agrees to indemnify Buyer and save it harmless from all losses, liabilities or claims including reasonable attorneys' fees and costs of court ("Claims"), from any and all persons, arising from or out of claims of title, personal injury (including death) or property damage from said Gas or other charges thereon which attach before title passes to Buyer. Buyer agrees to indemnify Seller and save it harmless from all Claims, from any and all persons, arising from or out of claims regarding payment, personal injury (including death) or property damage from said Gas or other charges thereon which attach after title passes to Buyer.
- 8.4. The parties agree that the delivery of and the transfer of title to all Gas under this Contract shall take place within the Customs Territory of the United States (as defined in general note 2 of the Harmonized Tariff Schedule of the United States 19 U.S.C. §1202, General Notes, page 3); provided, however, that in the event Seller took title to the Gas outside the Customs Territory of the United States, Seller represents and warrants that it is the importer of record for all Gas entered and delivered into the United States, and shall be responsible for entry and entry summary filings as well as the payment of duties, taxes and fees, if any, and all applicable record keeping requirements.
- 8.5. Notwithstanding the other provisions of this Section 8, as between Seller and Buyer, Seller will be liable for all Claims to the extent that such arise from the failure of Gas delivered by Seller to meet the quality requirements of Section 5.

SECTION 9. NOTICES

- 9.1. All Transaction Confirmations, invoices, payment instructions, and other communications made pursuant to the Base Contract ("Notices") shall be made to the addresses specified in writing by the respective parties from time to time.
- 9.2. All Notices required hereunder shall be in writing and may be sent by facsimile or mutually acceptable electronic means, a nationally recognized overnight courier service, first class mail or hand delivered.
- 9.3. Notice shall be given when received on a Business Day by the addressee. In the absence of proof of the actual receipt date, the following presumptions will apply. Notices sent by facsimile shall be deemed to have been received upon the sending party's receipt of its facsimile machine's confirmation of successful transmission. If the day on which such facsimile is received is not a Business Day or is after five p.m. on a Business Day, then such facsimile shall be deemed to have been received on the next following Business Day. Notice by overnight mail or courier shall be deemed to have been received on the next Business Day after it was sent or such earlier time as is confirmed by the receiving party. Notice via first class mail shall be considered delivered five Business Days after mailing.
- 9.4. The party receiving a commercially acceptable Notice of change in payment instructions or other payment information shall not be obligated to implement such change until ten Business Days after receipt of such Notice.

SECTION 10. FINANCIAL RESPONSIBILITY

10.1. If either party ("X") has reasonable grounds for insecurity regarding the performance of any obligation under this Contract (whether or not then due) by the other party ("Y") (including, without limitation, the occurrence of a material change in the

creditworthiness of Y or its Guarantor, if applicable), X may demand Adequate Assurance of Performance. "Adequate Assurance of Performance" shall mean sufficient security in the form, amount, for a term, and from an issuer, all as reasonably acceptable to X, including, but not limited to cash, a standby irrevocable letter of credit, a prepayment, a security interest in an asset or guaranty. Y hereby grants to X a continuing first priority security interest in, lien on, and right of setoff against all Adequate Assurance of Performance in the form of cash transferred by Y to X pursuant to this Section 10.1. Upon the return by X to Y of such Adequate Assurance of Performance, the security interest and lien granted hereunder on that Adequate Assurance of Performance shall be released automatically and, to the extent possible, without any further action by either party.

- 10.2. In the event (each an "Event of Default") either party (the "Defaulting Party") or its Guarantor shall: (i) make an assignment or any general arrangement for the benefit of creditors; (ii) file a petition or otherwise commence, authorize, or acquiesce in the commencement of a proceeding or case under any bankruptcy or similar law for the protection of creditors or have such petition filed or proceeding commenced against it; (iii) otherwise become bankrupt or insolvent (however evidenced); (iv) be unable to pay its debts as they fall due; (v) have a receiver, provisional liquidator, conservator, custodian, trustee or other similar official appointed with respect to it or substantially all of its assets; (vi) fail to perform any obligation to the other party with respect to any Credit Support Obligations relating to the Contract; (vii) fail to give Adequate Assurance of Performance under Section 10.1 within 48 hours but at least one Business Day of a written request by the other party; (viii) not have paid any amount due the other party hereunder on or before the second Business Day following written Notice that such payment is due; or ix) be the affected party with respect to any Additional Event of Default; then the other party (the "Non-Defaulting Party") shall have the right, at its sole election, to immediately withhold and/or suspend deliveries or payments upon Notice and/or to terminate and liquidate the transactions under the Contract, in the manner provided in Section 10.3, in addition to any and all other remedies available hereunder.
- 10.3. If an Event of Default has occurred and is continuing, the Non-Defaulting Party shall have the right, by Notice to the Defaulting Party, to designate a Day, no earlier than the Day such Notice is given and no later than 20 Days after such Notice is given, as an early termination date (the "Early Termination Date") for the liquidation and termination pursuant to Section 10.3.1 of all transactions under the Contract, each a "Terminated Transaction". On the Early Termination Date, all transactions will terminate, other than those transactions, if any, that may not be liquidated and terminated under applicable law ("Excluded Transactions"), which Excluded Transactions must be liquidated and terminated as soon thereafter as is legally permissible, and upon termination shall be a Terminated Transaction and be valued consistent with Section 10.3.1 below. With respect to each Excluded Transaction, its actual termination date shall be the Early Termination Date for purposes of Section 10.3.1.

The parties have selected either "Early Termination Damages Apply" or "Early Termination Damages Do Not Apply" as indicated on the Base Contract.

Early Termination Damages Apply:

10.3.1. As of the Early Termination Date, the Non-Defaulting Party shall determine, in good faith and in a commercially reasonable manner, (i) the amount owed (whether or not then due) by each party with respect to all Gas delivered and received between the parties under Terminated Transactions and Excluded Transactions on and before the Early Termination Date and all other applicable charges relating to such deliveries and receipts (including without limitation any amounts owed under Section 3.2), for which payment has not yet been made by the party that owes such payment under this Contract and (ii) the Market Value, as defined below, of each Terminated Transaction. The Non-Defaulting Party shall (x) liquidate and accelerate each Terminated Transaction at its Market Value, so that each amount equal to the difference between such Market Value and the Contract Value, as defined below, of such Terminated Transaction(s) shall be due to the Buyer under the Terminated Transaction(s) if such Market Value exceeds the Contract Value and to the Seller if the opposite is the case; and (y) where appropriate, discount each amount then due under clause (x) above to present value in a commercially reasonable manner as of the Early Termination Date (to take account of the period between the date of liquidation and the date on which such amount would have otherwise been due pursuant to the relevant Terminated Transactions).

For purposes of this Section 10.3.1, "Contract Value" means the amount of Gas remaining to be delivered or purchased under a transaction multiplied by the Contract Price, and "Market Value" means the amount of Gas remaining to be delivered or purchased under a transaction multiplied by the market price for a similar transaction at the Delivery Point determined by the Non-Defaulting Party in a commercially reasonable manner. To ascertain the Market Value, the Non-Defaulting Party may consider, among other valuations, any or all of the settlement prices of NYMEX Gas futures contracts, quotations from leading dealers in energy swap contracts or physical gas trading markets, similar sales or purchases and any other bona fide third-party offers, all adjusted for the length of the term and differences in transportation costs. A party shall not be required to enter into a replacement transaction(s) in order to determine the Market Value. Any extension(s) of the term of a transaction to which parties are not bound as of the Early Termination Date (including but not limited to "evergreen provisions") shall not be considered in determining Contract Values and Market Values. For the avoidance of doubt, any option pursuant to which one party has the right to extend the term of a transaction shall be considered in determining Contract Values and Market Values. The rate of interest used in calculating net present value shall be determined by the Non-Defaulting Party in a commercially reasonable manner.

Early Termination Damages Do Not Apply:

10.3.1. As of the Early Termination Date, the Non-Defaulting Party shall determine, in good faith and in a commercially reasonable manner, the amount owed (whether or not then due) by each party with respect to all Gas delivered and received between the parties under Terminated Transactions and Excluded Transactions on and before the Early Termination Date and all other applicable charges relating to such deliveries and receipts (including without limitation any amounts owed under Section 3.2), for which payment has not yet been made by the party that owes such payment under this Contract.

The parties have selected either "Other Agreement Setoffs Apply" or "Other Agreement Setoffs Do Not Apply" as indicated on the Base Contract.

Other Agreement Setoffs Apply:

Bilateral Setoff Option:

10.3.2. The Non-Defaulting Party shall net or aggregate, as appropriate, any and all amounts owing between the parties under Section 10.3.1, so that all such amounts are netted or aggregated to a single liquidated amount payable by one party to the other (the "Net Settlement Amount"). At its sole option and without prior Notice to the Defaulting Party, the Non-Defaulting Party is hereby authorized to setoff any Net Settlement Amount against (i) any margin or other collateral held by a party in connection with any Credit Support Obligation relating to the Contract; and (ii) any amount(s) (including any excess cash margin or excess cash collateral) owed or held by the party that is entitled to the Net Settlement Amount under any other agreement or arrangement between the parties.

Triangular Setoff Option:

10.3.2. The Non-Defaulting Party shall net or aggregate, as appropriate, any and all amounts owing between the parties under Section 10.3.1, so that all such amounts are netted or aggregated to a single liquidated amount payable by one party to the other (the "Net Settlement Amount"). At its sole option, and without prior Notice to the Defaulting Party, the Non-Defaulting Party is hereby authorized to setoff (i) any Net Settlement Amount against any margin or other collateral held by a party in connection with any Credit Support Obligation relating to the Contract; (ii) any Net Settlement Amount against any amount(s) (including any excess cash margin or excess cash collateral) owed by or to a party under any other agreement or arrangement between the parties; (iii) any Net Settlement Amount owed to the Non-Defaulting Party against any amount(s) (including any excess cash margin or excess cash collateral) owed by the Non-Defaulting Party or its Affiliates to the Defaulting Party under any other agreement; (iv) any Net Settlement Amount owed to the Defaulting Party against any amount(s) (including any excess cash margin or excess cash collateral) owed by the Defaulting Party to the Non-Defaulting Party against any amount(s) (including any excess cash margin or excess cash collateral) owed by the Defaulting Party or its Affiliates to the Non-Defaulting Party under any other agreement or arrangement.

Other Agreement Setoffs Do Not Apply:

- 10.3.2. The Non-Defaulting Party shall net or aggregate, as appropriate, any and all amounts owing between the parties under Section 10.3.1, so that all such amounts are netted or aggregated to a single liquidated amount payable by one party to the other (the "Net Settlement Amount"). At its sole option and without prior Notice to the Defaulting Party, the Non-Defaulting Party may setoff any Net Settlement Amount against any margin or other collateral held by a party in connection with any Credit Support Obligation relating to the Contract.
- 10.3.3. If any obligation that is to be included in any netting, aggregation or setoff pursuant to Section 10.3.2 is unascertained, the Non-Defaulting Party may in good faith estimate that obligation and net, aggregate or setoff, as applicable, in respect of the estimate, subject to the Non-Defaulting Party accounting to the Defaulting Party when the obligation is ascertained. Any amount not then due which is included in any netting, aggregation or setoff pursuant to Section 10.3.2 shall be discounted to net present value in a commercially reasonable manner determined by the Non-Defaulting Party.
- 10.4. As soon as practicable after a liquidation, Notice shall be given by the Non-Defaulting Party to the Defaulting Party of the Net Settlement Amount, and whether the Net Settlement Amount is due to or due from the Non-Defaulting Party. The Notice shall include a written statement explaining in reasonable detail the calculation of the Net Settlement Amount, provided that failure to give such Notice shall not affect the validity or enforceability of the liquidation or give rise to any claim by the Defaulting Party against the Non-Defaulting Party. The Net Settlement Amount as well as any setoffs applied against such amount pursuant to Section 10.3.2, shall be paid by the close of business on the second Business Day following such Notice, which date shall not be earlier than the Early Termination Date. Interest on any unpaid portion of the Net Settlement Amount as adjusted by setoffs, shall accrue from the date due until the date of payment at a rate equal to the lower of (i) the then-effective prime rate of interest published under "Money Rates" by The Wall Street Journal, plus two percent per annum; or (ii) the maximum applicable lawful interest rate.
- 10.5. The parties agree that the transactions hereunder constitute a "forward contract" within the meaning of the United States Bankruptcy Code and that Buyer and Seller are each "forward contract merchants" within the meaning of the United States Bankruptcy Code.
- 10.6. The Non-Defaulting Party's remedies under this Section 10 are the sole and exclusive remedies of the Non-Defaulting Party with respect to the occurrence of any Early Termination Date. Each party reserves to itself all other rights, setoffs, counterclaims and other defenses that it is or may be entitled to arising from the Contract.
- 10.7. With respect to this Section 10, if the parties have executed a separate netting agreement with close-out netting provisions, the terms and conditions therein shall prevail to the extent inconsistent herewith.

SECTION 11. FORCE MAJEURE

- 11.1. Except with regard to a party's obligation to make payment(s) due under Section 7, Section 10.4, and Imbalance Charges under Section 4, neither party shall be liable to the other for failure to perform a Firm obligation, to the extent such failure was caused by Force Majeure. The term "Force Majeure" as employed herein means any cause not reasonably within the control of the party claiming suspension, as further defined in Section 11.2.
- 11.2. Force Majeure shall include, but not be limited to, the following: (i) physical events such as acts of God, landslides, lightning, earthquakes, fires, storms or storm warnings, such as hurricanes, which result in evacuation of the affected area, floods, washouts, explosions, breakage or accident or necessity of repairs to machinery or equipment or lines of pipe; (ii) weather related events affecting an entire geographic region, such as low temperatures which cause freezing or failure of wells or lines of pipe;

Page 7 of 10

- (iii) interruption and/or curtailment of Firm transportation and/or storage by Transporters; (iv) acts of others such as strikes, lockouts or other industrial disturbances, riots, sabotage, insurrections or wars, or acts of terror; and (v) governmental actions such as necessity for compliance with any court order, law, statute, ordinance, regulation, or policy having the effect of law promulgated by a governmental authority having jurisdiction. Seller and Buyer shall make reasonable efforts to avoid the adverse impacts of a Force Majeure and to resolve the event or occurrence once it has occurred in order to resume performance.
- 11.3. Neither party shall be entitled to the benefit of the provisions of Force Majeure to the extent performance is affected by any or all of the following circumstances: (i) the curtailment of interruptible or secondary Firm transportation unless primary, in-path, Firm transportation is also curtailed; (ii) the party claiming excuse failed to remedy the condition and to resume the performance of such covenants or obligations with reasonable dispatch; or (iii) economic hardship, to include, without limitation, Seller's ability to sell Gas at a higher or more advantageous price than the Contract Price, Buyer's ability to purchase Gas at a lower or more advantageous price than the Contract Price, or a regulatory agency disallowing, in whole or in part, the pass through of costs resulting from this Contract; (iv) the loss of Buyer's market(s) or Buyer's inability to use or resell Gas purchased hereunder, except, in either case, as provided in Section 11.2; or (v) the loss or failure of Seller's gas supply or depletion of reserves, except, in either case, as provided in Section 11.2. The party claiming Force Majeure shall not be excused from its responsibility for Imbalance Charges.
- 11.4. Notwithstanding anything to the contrary herein, the parties agree that the settlement of strikes, lockouts or other industrial disturbances shall be within the sole discretion of the party experiencing such disturbance.
- 11.5. The party whose performance is prevented by Force Majeure must provide Notice to the other party. Initial Notice may be given orally; however, written Notice with reasonably full particulars of the event or occurrence is required as soon as reasonably possible. Upon providing written Notice of Force Majeure to the other party, the affected party will be relieved of its obligation, from the onset of the Force Majeure event, to make or accept delivery of Gas, as applicable, to the extent and for the duration of Force Majeure, and neither party shall be deemed to have failed in such obligations to the other during such occurrence or event.
- 11.6. Notwithstanding Sections 11.2 and 11.3, the parties may agree to alternative Force Majeure provisions in a Transaction Confirmation executed in writing by both parties.

SECTION 12. TERM

This Contract may be terminated on 30 Day's written Notice, but shall remain in effect until the expiration of the latest Delivery Period of any transaction(s). The rights of either party pursuant to Section 7.6, Section 10, Section 13, the obligations to make payment hereunder, and the obligation of either party to indemnify the other, pursuant hereto shall survive the termination of the Base Contract or any transaction.

SECTION 13. LIMITATIONS

FOR BREACH OF ANY PROVISION FOR WHICH AN EXPRESS REMEDY OR MEASURE OF DAMAGES IS PROVIDED, SUCH EXPRESS REMEDY OR MEASURE OF DAMAGES SHALL BE THE SOLE AND EXCLUSIVE REMEDY. A PARTY'S LIABILITY HEREUNDER SHALL BE LIMITED AS SET FORTH IN SUCH PROVISION, AND ALL OTHER REMEDIES OR DAMAGES AT LAW OR IN EQUITY ARE WAIVED. IF NO REMEDY OR MEASURE OF DAMAGES IS EXPRESSLY PROVIDED HEREIN OR IN A TRANSACTION, A PARTY'S LIABILITY SHALL BE LIMITED TO DIRECT ACTUAL DAMAGES ONLY. SUCH DIRECT ACTUAL DAMAGES SHALL BE THE SOLE AND EXCLUSIVE REMEDY, AND ALL OTHER REMEDIES OR DAMAGES AT LAW OR IN EQUITY ARE WAIVED. UNLESS EXPRESSLY HEREIN PROVIDED, NEITHER PARTY SHALL BE LIABLE FOR CONSEQUENTIAL, INCIDENTAL, PUNITIVE, EXEMPLARY OR INDIRECT DAMAGES, LOST PROFITS OR OTHER BUSINESS INTERRUPTION DAMAGES, BY STATUTE, IN TORT OR CONTRACT, UNDER ANY INDEMNITY PROVISION OR OTHERWISE. IT IS THE INTENT OF THE PARTIES THAT THE LIMITATIONS HEREIN IMPOSED ON REMEDIES AND THE MEASURE OF DAMAGES BE WITHOUT REGARD TO THE CAUSE OR CAUSES RELATED THERETO, INCLUDING THE NEGLIGENCE OF ANY PARTY, WHETHER SUCH NEGLIGENCE BE SOLE, JOINT OR CONCURRENT, OR ACTIVE OR PASSIVE. TO THE EXTENT ANY DAMAGES REQUIRED TO BE PAID HEREUNDER ARE LIQUIDATED, THE PARTIES ACKNOWLEDGE THAT THE DAMAGES ARE DIFFICULT OR IMPOSSIBLE TO DETERMINE, OR OTHERWISE OBTAINING AN ADEQUATE REMEDY IS INCONVENIENT AND THE DAMAGES CALCULATED HEREUNDER CONSTITUTE A REASONABLE APPROXIMATION OF THE HARM OR LOSS.

SECTION 14. MARKET DISRUPTION

If a Market Disruption Event has occurred then the parties shall negotiate in good faith to agree on a replacement price for the Floating Price (or on a method for determining a replacement price for the Floating Price) for the affected Day, and if the parties have not so agreed on or before the second Business Day following the affected Day then the replacement price for the Floating Price shall be determined within the next two following Business Days with each party obtaining, in good faith and from nonaffiliated market participants in the relevant market, two quotes for prices of Gas for the affected Day of a similar quality and quantity in the geographical location closest in proximity to the Delivery Point and averaging the four quotes. If either party fails to provide two quotes then the average of the other party's two quotes shall determine the replacement price for the Floating Price. "Floating Price" means the price or a factor of the price agreed to in the transaction as being based upon a specified index. "Market Disruption Event" means, with respect to an index specified for a transaction, any of the following events: (a) the failure of the index to announce or publish information necessary for determining the Floating Price; (b) the failure of trading to commence or the permanent discontinuation or material suspension of trading on the exchange or market acting as the index; (c) the temporary or permanent discontinuance or unavailability of the index, (d) the temporary or permanent closing of any exchange acting as the index; or (e) both parties agree that a material change in the formula for or the method of determining the Floating Price has occurred. For the purposes of the calculation of a replacement price for the Floating Price, all numbers shall be rounded to three decimal places. If the fourth decimal number is five or greater, then the third decimal number shall be increased by one and if the fourth decimal number is less than five, then the third decimal number shall remain unchanged.

SECTION 15. MISCELLANEOUS

- 15.1. This Contract shall be binding upon and inure to the benefit of the successors, assigns, personal representatives, and heirs of the respective parties hereto, and the covenants, conditions, rights and obligations of this Contract shall run for the full term of this Contract. No assignment of this Contract, in whole or in part, will be made without the prior written consent of the non-assigning party (and shall not relieve the assigning party from liability hereunder), which consent will not be unreasonably withheld or delayed; provided, either party may (i) transfer, sell, pledge, encumber, or assign this Contract or the accounts, revenues, or proceeds hereof in connection with any financing or other financial arrangements, or (ii) transfer its interest to any parent or Affiliate by assignment, merger or otherwise without the prior approval of the other party. Upon any such assignment, transfer and assumption, the transferor shall remain principally liable for and shall not be relieved of or discharged from any obligations hereunder.
- 15.2. If any provision in this Contract is determined to be invalid, void or unenforceable by any court having jurisdiction, such determination shall not invalidate, void, or make unenforceable any other provision, agreement or covenant of this Contract.
- 15.3. No waiver of any breach of this Contract shall be held to be a waiver of any other or subsequent breach.
- 15.4. This Contract sets forth all understandings between the parties respecting each transaction subject hereto, and any prior contracts, understandings and representations, whether oral or written, relating to such transactions are merged into and superseded by this Contract and any effective transaction(s). This Contract may be amended only by a writing executed by both parties.
- 15.5. The interpretation and performance of this Contract shall be governed by the laws of the jurisdiction as indicated on the Base Contract, excluding, however, any conflict of laws rule which would apply the law of another jurisdiction.
- 15.6. This Contract and all provisions herein will be subject to all applicable and valid statutes, rules, orders and regulations of any governmental authority having jurisdiction over the parties, their facilities, or Gas supply, this Contract or transaction or any provisions thereof.
- 15.7. There is no third party beneficiary to this Contract.
- 15.8. Each party to this Contract represents and warrants that it has full and complete authority to enter into and perform this Contract. Each person who executes this Contract on behalf of either party represents and warrants that it has full and complete authority to do so and that such party will be bound thereby.
- 15.9. The headings and subheadings contained in this Contract are used solely for convenience and do not constitute a part of this Contract between the parties and shall not be used to construe or interpret the provisions of this Contract.
- 15.10. Unless the parties have elected on the Base Contract not to make this Section 15.10 applicable to this Contract, neither party shall disclose directly or indirectly without the prior written consent of the other party the terms of any transaction to a third party (other than the employees, lenders, royalty owners, counsel, accountants and other agents of the party, or prospective purchasers of all or substantially all of a party's assets or of any rights under this Contract, provided such persons shall have agreed to keep such terms confidential) except (i) in order to comply with any applicable law, order, regulation, or exchange rule, (ii) to the extent necessary for the enforcement of this Contract, (iii) to the extent necessary to implement any transaction, (iv) to the extent necessary to comply with a regulatory agency's reporting requirements including but not limited to gas cost recovery proceedings; or (v) to the extent such information is delivered to such third party for the sole purpose of calculating a published index. Each party shall notify the other party of any proceeding of which it is aware which may result in disclosure of the terms of any transaction (other than as permitted hereunder) and use reasonable efforts to prevent or limit the disclosure. The existence of this Contract is not subject to this confidentiality obligation. Subject to Section 13, the parties shall be entitled to all remedies available at law or in equity to enforce, or seek relief in connection with this confidentiality obligation. The terms of any transaction hereunder shall be kept confidential by the parties hereto for one year from the expiration of the transaction.

In the event that disclosure is required by a governmental body or applicable law, the party subject to such requirement may disclose the material terms of this Contract to the extent so required, but shall promptly notify the other party, prior to disclosure, and shall cooperate (consistent with the disclosing party's legal obligations) with the other party's efforts to obtain protective orders or similar restraints with respect to such disclosure at the expense of the other party.

Page 9 of 10

- 15.11. The parties may agree to dispute resolution procedures in Special Provisions attached to the Base Contract or in a Transaction Confirmation executed in writing by both parties
- 15.12. Any original executed Base Contract, Transaction Confirmation or other related document may be digitally copied, photocopied, or stored on computer tapes and disks (the "Imaged Agreement"). The Imaged Agreement, if introduced as evidence on paper, the Transaction Confirmation, if introduced as evidence in automated facsimile form, the recording, if introduced as evidence in its original form, and all computer records of the foregoing, if introduced as evidence in printed format, in any judicial, arbitration, mediation or administrative proceedings will be admissible as between the parties to the same extent and under the same conditions as other business records originated and maintained in documentary form. Neither Party shall object to the admissibility of the recording, the Transaction Confirmation, or the Imaged Agreement on the basis that such were not originated or maintained in documentary form. However, nothing herein shall be construed as a waiver of any other objection to the admissibility of such evidence.

DISCLAIMER: The purposes of this Contract are to facilitate trade, avoid misunderstandings and make more definite the terms of contracts of purchase and sale of natural gas. Further, NAESB does not mandate the use of this Contract by any party. NAESB DISCLAIMS AND EXCLUDES, AND ANY USER OF THIS CONTRACT ACKNOWLEDGES AND AGREES TO NAESB'S DISCLAIMER OF, ANY AND ALL WARRANTIES, CONDITIONS OR REPRESENTATIONS, EXPRESS OR IMPLIED, ORAL OR WRITTEN, WITH RESPECT TO THIS CONTRACT OR ANY PART THEREOF, INCLUDING ANY AND ALL IMPLIED WARRANTIES OR CONDITIONS OF TITLE, NON-INFRINGEMENT, MERCHANTABILITY, OR FITNESS OR SUITABILITY FOR ANY PARTICULAR PURPOSE (WHETHER OR NOT NAESB KNOWS, HAS REASON TO KNOW, HAS BEEN ADVISED, OR IS OTHERWISE IN FACT AWARE OF ANY SUCH PURPOSE), WHETHER ALLEGED TO ARISE BY LAW, BY REASON OF CUSTOM OR USAGE IN THE TRADE, OR BY COURSE OF DEALING. EACH USER OF THIS CONTRACT ALSO AGREES THAT UNDER NO CIRCUMSTANCES WILL NAESB BE LIABLE FOR ANY DIRECT, SPECIAL, INCIDENTAL, EXEMPLARY, PUNITIVE OR CONSEQUENTIAL DAMAGES ARISING OUT OF ANY USE OF THIS CONTRACT.

EXHIBIT A - FORM OF TRANSACTION CONFIRMATION*

*The actual Transaction Confirmation ent and other local requirements and terms a only and CESNG reserves the right to alte	pplicable to the Facility.	This Form of Transacti	on Confirmation is for illustrative purpose						
This Transaction Confirmation is delivered pursuant to and in accordance with a natural gas supply agreement effective, 20 (the "Master Agreement"), by and between Constellation Energy Services – Natural Gas, LLC ("CESNG") and ("Buyer"), and is subject to and made part of the terms and conditions of such Master Agreement. Capitalized terms used herein but not defined will have the meanings ascribed to them in the Master Agreement.									
Trade Date:									
Facility Name:									
Delivery Period:									
<u>Deal Type</u> :									
Nature of Parties' Obligation:									
Contract Quantity/Price:									
Month/Yr	Contract Quanti	ty (in MMBtu)	Contract Price						
Delivery Point(s):									
<u>Utility</u> :									
This Transaction Confirmation docum It is binding and shall be deemed acc in writing within two (2) business days CONSTELLATION ENERGY SERVICES	epted as the definitive of CESNG's execution	expression of the Tra							
Ву:		By: SAMPLE							
Name: Title		Name: NOT FOR E							
Date:	.	Date:							

TRANSACTION CONFIRMATION

Page 1 of 2

This Transaction Confirmation dated May 4, 2016 is made a part of, and pursuant to terms of, the Base Contract for Sale and Purchase of Gas dated May 4, 2016 between Buyer and Seller identified herein (the "NAESB").

GELLED.

Constellation Energy Services - Natural Gas, LLC

BUYER:

Milwaukee Board of School Directors on behalf of Milwaukee Public Schools

ACCOUNTS:

Check here if Attachment A is attached and made a part of this Transaction Confirmation.

1716 Lawrence Drive PO Box 6025

De Pere, WI 54115-6025 Attn: Account Management E-mail: amwi@constellation.com Phone: 888-367-4403 (Option 2)

Fax: 443-213-6116

Attn: John Linn

1124 N. 11th St. Milwaukee WI 53233

E-mail: linnja@milwaukee.k12.wi.us

Phone: 414-283-4703 Fax: 414-283-4682

Agent / Assoc: Julie Cowling

Account Executive:

Delivery Period: July 1, 2016 to June 30, 2019; provided however, the Delivery Period shall begin on the later of, (i) the day after the first meter read during the period noted above, or (ii) upon acceptance by the utility of Seller as Buyer's gas supplier.

Contract Quantity (in dekatherms): see table

Month	Quantity	Month	Quantity	Month	Quantity	<u>Month</u>	Quantity
Jul	5,600	Oct	46,000	Jan	165,000	Apr	66,700
Aug	5,100	Nov	95,800	Feb	149,000	May	28,000
Sep	8,500	Dec	162,000	Mar	117,000	Jun	9,000

Price: NYMEX Last Day Settle ("NYM LDS") plus Market Basis per dekatherm. Either or both of the price components may be fixed for specific quantities and delivery months with mutual agreement of the Parties. The Basis is fixed at \$0.755 per dekatherm.

Performance Obligation: Up to 7,300 dkthm/day shall be Firm. The remainder shall be Interruptible

Delivery Point: Buyer's Transporter Citygate
Buyer's Transporter: WE Energies

- SPECIAL PROVISIONS -

Balancing Services.

1. A Balancing Service charge of \$0.03 per dekatherm applies in addition to the Price.

Buyer and Seller shall use commercially reasonable efforts to provide each other with sufficient notice to allow timely and accurate nominations in accordance with the nomination requirements of that Transporter, as they may change from time to time due to Constraint Day requirements or otherwise. Buyer accepts the following responsibilities: Buyer hereby appoints Seller as its agent for the purposes of balancing, including the receipt of current and historic Usage, storage, billing and transportation data from Transporter. Buyer agrees to assist Seller in the timely collection of data directly and through Transporter, notify Seller of usage expectations and any material changes to those usage expectations, and comply with any Daily Limitations concerning use. If Usage data is unavailable electronically, Buyer shall obtain usage data manually and deliver that information weekly via facsimile or electronic mail, or as requested by Seller.

- 2. Seller agrees to monitor and review all information provided to the Seller by Buyer and Transporter regarding Buyer's gas Use. Based on such information, Seller shall use commercially reasonable effort to keep Buyer within the balancing parameters established by Transporter. Seller shall pass through to Buyer and Buyer shall pay for any Imbalance Charges and/or Cashout resulting from Buyer's failure to (i) fulfill its obligations set forth in paragraph 1 above, (ii) maintain telemetry, and/or (iii) meet Daily Limitations with respect to Use.
- 3. Mid-month variations to delivered quantity. In the event that the quantity of natural gas that can be delivered to Buyer (without the Transporter assessing Imbalance Charges or Cashout) is less than the quantity agreed to be purchased for the relevant period, Buyer shall remain liable for the entire quantity it agreed to purchase, however, Seller will provide a credit at prevailing market rates for that portion of the purchased quantity that is not delivered. In the event that Buyer's expected Use will exceed the quantity of gas purchased by Buyer and/or available to Buyer without incurring Imbalance Charges for the relevant period, Seller will use commercially reasonable efforts to provide incremental supply, plus Fuel (if applicable) at prevailing market rates. Credits for reductions in delivered quantity and/or charges for incremental supply shall appear as separate line items on Seller's invoice to Buyer, including applicable price and quantity.

EXHIBIT A

TRANSACTION CONFIRMATION

Page 2 of 2

4. **Telemetry Obligations**. Buyer acknowledges and agrees that it shall be responsible for installing and maintaining a meter telephone line. Buyer shall also be responsible for any charges imposed by a contractor for repairing the meter telephone line. As a convenience to Buyer, Seller may arrange for telemetry service or repairs to be provided and Buyer agrees it shall be responsible for any costs associated with such service and repairs.

"Constraint Day" means any period of time during which a Transporter issues balancing orders, operational flow orders, operational matching orders (or the equivalent), or winter splits or deems the day a Critical Gas Day (or the equivalent) and thereby restricts deliveries by Seller or use by Buyer, requires a specific delivery by Seller or use by Buyer, or otherwise requires an affirmative action by either Party due to adverse operational conditions experienced by the Transporter.

"Daily Limitations" means Transporter enforced restrictions or mandates concerning quantities of gas delivered and/or used, including those imposed due to Constraint Day conditions and/or under ordinary operating conditions.

- 5. Invoices. Invoices shall be prepared based upon quantities used.
- 6. Consolidated Billing: Seller shall include the Transporter distribution fees on the monthly invoice. Past due and other charges due Transporter, that apply to service prior to the effective date of the Confirmation will not be included in Seller's invoice to Buyer. Seller may discontinue this service at any time with notice to Buyer.
- 7. Term: This Transaction Confirmation is effective for the Delivery Period as it may be extended or renewed as described below.
- 8. Extending Delivery Period: Should the Parties agree to a fixed price for any pricing component for a delivery month (or billing cycle, as applicable) that occurs after the current Delivery Period, then the Delivery Period of this Transaction Confirmation shall be extended to the latest occurring delivery month (or billing cycle, as applicable) with a fixed price or fixed price component.
- 9. Renewal: The Parties agree that this Transaction Confirmation shall automatically renew for successive twelve (12) month periods (each constituting a "Renewal Delivery Period") unless written notice pursuant to Section 9 of the Agreement terminating the Transaction Confirmation at the end of a Delivery Period is given by either Party thirty (30) days prior to the end of the Delivery Period. The Contract Quantity during any Renewal Delivery Period shall be the Contract Quantity stated in the table above for the respective month. The Price for the Contract Quantity during any Renewal Delivery Period shall be NYM LDS plus Market Basis.

All other terms and conditions of this Transaction Confirmation shall remain unchanged.

Special Condition: This Confirmation is not effective and binding upon the parties hereto unless signed concurrently with the Base Contract for Sale and Purchase of Gas (NAESB) dated effective May 4, 2016

Seller: Constellation Energy Services – Natural Gas, LLC

Buyer: Milwaukee Board of School Directors on behalf of Milwaukee Public Schools

Signature:

By: David T. Dona +

Title: Ville President Refail Operations

Date: Mark A. Sain

Title: President

Date: President

Date: President

Date: President

This Transaction Confirmation dated May 4, 2016 is made a part of, and pursuant to terms of, the Base Contract for Sale and Purchase of Gas dated May 4, 2016 between Buyer and Seller identified herein (the "NAESB").

SELLER:

Constellation Energy Services - Natural Gas, LLC

BUYER:

Milwaukee Board of School Directors on behalf of Milwaukee Public Schools

ACCOUNTS:

☑ Check here if Attachment A is attached and made a part of this Transaction Confirmation.

1716 Lawrence Drive PO Box 6025 De Pere, WI 54115-6025 Attn: Account Management

E-mail: amwi@constellation.com Phone: 888-367-4403 (Option 2)

Fax: 443-213-6116

Attn: John Linn

1124 N. 11th St.

Milwaukee WI 53233

E-mail: linnja@milwaukee.k12.wi.us

Phone: 414-283-4703 Fax: 414-283-4682

Agent / Assoc: Julie Cowling

Account Executive:

Delivery Period: July 1, 2016 to June 30, 2019; provided however, the Delivery Period shall begin on the later of, (i) the day after the first meter read during the period noted above, or (ii) upon acceptance by the utility of Seller as Buyer's gas supplier.

Contract Quantity (in dekatherms): see table

Month	Quantity	antity <u>Month</u> <u>Quantity</u> <u>Month</u> <u>Quantity</u>		Quantity	<u>Month</u>	Quantity	
Jul	5,600	Oct	46,000	Jan	165,000	Apr	66,700
Aug	5,100	Nov	95,800	Feb	149,000	May	28,000
Sep	8,500	Dec	162,000	Mar	117,000	Jun	9,000

Price: NYMEX Last Day Settle ("NYM LDS") plus Market Basis per dekatherm. Either or both of the price components may be fixed for specific quantities and delivery months with mutual agreement of the Parties. The Basis is fixed at \$0.755 per dekatherm.

Performance Obligation: Up to 7,300 dkthm/day shall be Firm. The remainder shall be Interruptible

Delivery Point: Buyer's Transporter Citygate

Buyer's Transporter: WE Energies

- SPECIAL PROVISIONS -

Balancing Services.

1. A Balancing Service charge of \$0.03 per dekatherm applies in addition to the Price.

Buyer and Seller shall use commercially reasonable efforts to provide each other with sufficient notice to allow timely and accurate nominations in accordance with the nomination requirements of that Transporter, as they may change from time to time due to Constraint Day requirements or otherwise. Buyer accepts the following responsibilities: Buyer hereby appoints Seller as its agent for the purposes of balancing, including the receipt of current and historic Usage, storage, billing and transportation data from Transporter. Buyer agrees to assist Seller in the timely collection of data directly and through Transporter, notify Seller of usage expectations and any material changes to those usage expectations, and comply with any Daily Limitations concerning use. If Usage data is unavailable electronically, Buyer shall obtain usage data manually and deliver that information weekly via facsimile or electronic mail, or as requested by Seller.

- 2. Seller agrees to monitor and review all information provided to the Seller by Buyer and Transporter regarding Buyer's gas Use. Based on such information, Seller shall use commercially reasonable effort to keep Buyer within the balancing parameters established by Transporter. Seller shall pass through to Buyer and Buyer shall pay for any Imbalance Charges and/or Cashout resulting from Buyer's failure to (i) fulfill its obligations set forth in paragraph 1 above, (ii) maintain telemetry, and/or (iii) meet Daily Limitations with respect to Use.
- 3. Mid-month variations to delivered quantity. In the event that the quantity of natural gas that can be delivered to Buyer (without the Transporter assessing Imbalance Charges or Cashout) is less than the quantity agreed to be purchased for the relevant period, Buyer shall remain liable for the entire quantity it agreed to purchase, however, Seller will provide a credit at prevailing market rates for that portion of the purchased quantity that is not delivered. In the event that Buyer's expected Use will exceed the quantity of gas purchased by Buyer and/or available to Buyer without incurring Imbalance Charges for the relevant period, Seller will use commercially reasonable efforts to provide incremental supply, plus Fuel (if applicable) at prevailing market rates. Credits for reductions in delivered quantity and/or charges for incremental supply shall appear as separate line items on Seller's invoice to Buyer, including applicable price and quantity.

TRANSACTION CONFIRMATION

Page 2 of 2

4. **Telemetry Obligations**. Buyer acknowledges and agrees that it shall be responsible for installing and maintaining a meter telephone line. Buyer shall also be responsible for any charges imposed by a contractor for repairing the meter telephone line. As a convenience to Buyer, Seller may arrange for telemetry service or repairs to be provided and Buyer agrees it shall be responsible for any costs associated with such service and repairs.

"Constraint Day" means any period of time during which a Transporter issues balancing orders, operational flow orders, operational matching orders (or the equivalent), or winter splits or deems the day a Critical Gas Day (or the equivalent) and thereby restricts deliveries by Seller or use by Buyer, requires a specific delivery by Seller or use by Buyer, or otherwise requires an affirmative action by either Party due to adverse operational conditions experienced by the Transporter.

"Daily Limitations" means Transporter enforced restrictions or mandates concerning quantities of gas delivered and/or used, including those imposed due to Constraint Day conditions and/or under ordinary operating conditions.

5. Invoices. Invoices shall be prepared based upon quantities used.

Seller: Constellation Energy Services - Natural Gas, LLC

- 6. Consolidated Billing: Seller shall include the Transporter distribution fees on the monthly invoice. Past due and other charges due Transporter, that apply to service prior to the effective date of the Confirmation will not be included in Seller's invoice to Buyer. Seller may discontinue this service at any time with notice to Buyer.
- 7. Term: This Transaction Confirmation is effective for the Delivery Period as it may be extended or renewed as described below.
- 8. **Extending Delivery Period:** Should the Parties agree to a fixed price for any pricing component for a delivery month (or billing cycle, as applicable) that occurs after the current Delivery Period, then the Delivery Period of this Transaction Confirmation shall be extended to the latest occurring delivery month (or billing cycle, as applicable) with a fixed price or fixed price component.
- 9. Renewal: The Parties agree that this Transaction Confirmation shall automatically renew for successive twelve (12) month periods (each constituting a "Renewal Delivery Period") unless written notice pursuant to Section 9 of the Agreement terminating the Transaction Confirmation at the end of a Delivery Period is given by either Party thirty (30) days prior to the end of the Delivery Period. The Contract Quantity during any Renewal Delivery Period shall be the Contract Quantity stated in the table above for the respective month. The Price for the Contract Quantity during any Renewal Delivery Period shall be NYM LDS plus Market Basis.

All other terms and conditions of this Transaction Confirmation shall remain unchanged.

Special Condition: This Confirmation is not effective and binding upon the parties hereto unless signed concurrently with the Base Contract for Sale and Purchase of Gas (NAESB) dated effective May 4, 2016

Buyer: Milwaukee Board of School Directors on behalf of

Signature: Donat Signature: By: David T. Donat By: Title: Vice President Retail Operations Title: Date: 6/6/16 Date:

RETAIL SPECIAL PROVISIONS

TO THE BASE CONTRACT FOR SALE AND PURCHASE OF NATURAL GAS (2006) BETWEEN CONSTELLATION ENERGY SERVICES – NATURAL GAS, LLC ("CESNG" or "Seller") AND MILWAUKEE BOARD OF SCHOOL DIRECTORS ON BEHALF OF MILWAUKEE PUBLIC SCHOOLS ("Buyer") DATED MAY 4, 2016

The following special provisions to the Base Contract and General Terms and Conditions shall supplement and form part of the Base Contract and General Terms and Conditions between the parties. In the event of any conflict or inconsistency between such special provisions and the Base Contract or General Terms and Conditions, the special provisions shall govern.

SECTION 1. PURPOSE AND PROCEDURES

Section 1.1 is deleted in its entirety and replaced with the following:

- 1.1 These General Terms and Conditions are intended to facilitate purchase and sale transactions of Gas (each a "Transaction") for one or more of Buyer's accounts or facilities (each a "Facility") on a Firm, Secondary Firm or Interruptible basis. The terms of any Transaction between CESNG and Customer shall be as set forth on the applicable TC, but CESNG will procure gas from its suppliers on terms CESNG deems appropriate. The parties may further agree to implement certain strategies or related services, the specifics of which will be set forth in a rider executed by the parties ("Rider"). The entire agreement between the parties shall be the Contract as defined in Section 2.9.
- 1.2 Section 1.2 "Oral Transaction Procedure" is amended by adding the phrase "or other electronic means of communication" immediately after the phrase "EDI transmission" in the second sentence.
- 1.3 The last sentence of Section 1.3 is deleted in its entirety and replaced with the following:

"In the event of a conflict among the terms of (i) a binding Transaction Confirmation pursuant to Section 1.2, (ii) a Rider, (iii) the oral agreement of the parties which may be evidenced by a recorded conversation, where the parties have selected the Oral Transaction Procedure of the Base Contract, (iv) the Base Contract, and (v) these General Terms and Conditions, the terms of the documents shall govern in the priority listed in this sentence."

The following shall be added to Section 1:

1.5 **Price Locks.** Unless otherwise set forth in a Transaction Confirmation or a Rider, Buyer can request to lock in the commodity price for any month(s) at any time during the Delivery Period, prior to 12:00 p.m. EST/EDT on the final day of NYMEX last day settlement for each applicable delivery month. Commodity purchase/sale prices exclude pipeline and utility distribution charges. Buyer also has the right to lock basis at a fixed price and at predetermined volumes. Basis includes interstate and intrastate pipeline transportation but does not include the commodity cost or the utility distribution charges.

SECTION 2. DEFINITIONS

Section 2.9 is deleted in its entirety and replaced with the following:

- 2.9 "Contract" shall mean the legally-binding relationship established by (i) the Milwaukee Board of School Directors Professional Services Contract, (ii) the Base Contract, (iii) any and all binding Transaction Confirmations, (iv) any and all Riders, and (v) where the parties have selected the Oral Transaction Procedure in Section 1.2 of the Base Contract, any and all transactions that the parties have entered into through an EDI transmission or by telephone, but that have not been confirmed in a binding Transaction Confirmation.
- 2.10 Section 2.10 "Contract Price" shall be amended by adding the following at the end of the section: "The Contract Price includes reimbursement to Seller for any production, severance, ad valorem or other Taxes owed with respect to Gas prior to delivery to Buyer, all of which Taxes shall be borne and paid exclusively by Seller." 2.12 Section 2.12 "Cover Standard" is amended by deleting the words "(or an alternate fuel if elected by Buyer and replacement Gas is not available)" in the third line of Section 2.12 ("Cover Standard").

Section 2.16 is deleted in its entirety and replaced with the following:

- 2.16 "Delivery Point" shall mean city gate interconnection between the Buyer's utility and the upstream Transporter or such other delivery point(s) as are agreed in a Transaction. However, to the extent that Buyer owned transportation capacity is used to deliver Gas to the Buyer, the Delivery Point shall be the receipt point of such capacity.
- 2.21 Section 2.21 "Guarantor" shall be deleted in its entirety and replaced with the following: "Guarantor shall mean any entity or individual that has provided a guaranty or surety of the obligations of a party hereunder."
- 2.32 Section 2.32 "Transaction Confirmation" shall be amended by adding the following at the end of the section: "The term "Transaction Confirmation" shall have the same meaning as "TC"."
- 2.35 Section 2.35 "Transporter" is amended by adding the word ", storage," after the word "gathering" on the first line of Section 2.35 and adding the words "or storing" after the word "transporting" on the second line of Section 2.35.

The following sections shall be added to Section 2:

- 2.36 "Costs" means, with respect to the Non-Defaulting Party, brokerage fees, commissions and other similar transaction costs and expenses reasonably incurred by such party as a result of the Event of Default except for such amounts already included in the Net Settlement Amount.
- 2.37 "Market Price" is a price comprised of (i) the spot commodity cost of Gas as determined by CESNG in its reasonable discretion, plus (ii) all related interstate and intrastate pipeline charges required to deliver Gas to the Delivery Point, plus (iii) a reasonable market based margin. Market Price does not include the cost of utility imposed charges, including but not limited to utility or pipeline balancing charges unless otherwise agreed upon.
- 2.38 "Secondary Firm" means deliveries and receipts of Gas under this Contract will be on a best-efforts basis up to Buyer's maximum daily quantity as set forth in a Transaction Confirmation, if any. Seller may interrupt performance without liability to the extent that one or more of the following conditions are present: (i) Force Majeure, (ii) curtailment by Buyer's local distribution company, (iii) curtailment of supply by a Gas supplier, (iv) curtailment of storage by a storage provider, (v) curtailment of transportation by a Transporter, including, but not limited to, transportation between secondary firm points, (vi) recall of transportation capacity release by its releaser, (vii) curtailment of Gas production behind a specific meter. If Seller interrupts for any other reason, Buyer's exclusive remedy shall be that Buyer may recover damages as provided in Section 3.2 of the Base Contract.

SECTION 3. PERFORMANCE OBLIGATION

- 3.1 Section 3.1 shall be amended by inserting ", Secondary Firm," before "Firm" in the second sentence.
- 3.2 Section 3.2 shall be amended by inserting " or Secondary Firm" after "Firm" in the first sentence.

SECTION 6. TAXES

Section 6 is deleted in its entirety and replaced with the following:

6. Buyer shall pay all federal, state, municipal and local taxes, duties, fees, levies, premiums or other charges imposed by any governmental authority, directly or indirectly, on or with respect to the Gas and related products and services provided under the Contract, including without limitation any production, severance or ad valorem taxes, and including any taxes enacted after the Effective Date (collectively, "Taxes"). The term "Taxes" shall include any amounts imposed on Buyer directly or on CESNG in its function as Buyer's supplier, and that are associated with the supply of Gas to Buyer (in which case the Buyer shall be responsible to reimburse CESNG for such amounts). If Buyer is exempt from any Taxes, Buyer shall provide CESNG with any state and/or local exemption certificate prior to the issue date of Buyer's first invoice. All Taxes invoiced to Buyer under this Contract will be included on the invoice or in the applicable fixed price as allowed by law.

SECTION 7. BILLING, PAYMENT AND AUDIT

Section 7.1 is deleted in its entirety and replaced with the following:

- 7.1 Buyer will be billed for natural gas usage and related products and services supplied under the Contract in one of the following ways based on availability and eligibility of each Facility, which may change from time to time: (i) Dual Billing: Buyer will receive two invoices, one from CESNG for the natural gas supply and one from the utility for the amounts payable by Buyer for services provided by the utility ("Delivery Charges"); (ii) CESNG Consolidated Billing: Buyer will receive one invoice from CESNG that includes both the natural gas supply charges and the Delivery Charges. In the case of CESNG Consolidated Billing, Buyer agrees that (i) Buyer remains exclusively liable to the utility for all Delivery Charges, (ii) CESNG has no obligation to review Delivery Charges for accuracy, (iii) should Buyer dispute Delivery Charges, that is a matter for Buyer and the utility to resolve without any involvement or obligation on the part of CESNG, and (iv) CESNG may withhold any payments due to the utility if Buyer fails to pay CESNG invoices in accordance with this Contract. If the actual quantity of Gas delivered is not known by the invoice date, invoices will be prepared based on estimated meter data or other estimated information. The invoiced quantity will then be adjusted to the actual quantity on the following Month's invoice or as soon thereafter as actual delivery information is available. Section 7.2 is deleted in its entirety and replaced with the following:
- 7.2 All amounts set forth in an invoice are payable to the party issuing the invoice and will include, in addition to the natural gas supply charges, Delivery Charges, and Taxes, all other amounts related to the purchase and delivery of natural gas. CESNG's invoices will be sent to Buyer in accordance with CESNG's normal billing cycle, as adjusted from time to time. CESNG's invoices are due and payable on the Payment Date set forth on the cover page of the Base Contract, or such other date as set forth in a Rider, without offset or reduction of any kind, to the address set forth on the invoice; provided that if the Payment Date is not a Business Day, payment is due on the next Business Day following that date.
- 7.4 Section 7.4 shall be deleted in its entirety.

SECTION 8. TITLE, WARRANTY, AND INDEMNITY

8.1 The last sentence of Section 8.1 is deleted in its entirety and replaced with the following:

"Buyer shall have responsibility for and assume any liability with respect to said Gas at and after its delivery to Buyer at the Delivery Point(s)."

SECTION 10. FINANCIAL RESPONSIBILITY

Section 10.1 will be deleted in its entirety and replaced with the following:

10.1 If CESNG has reasonable grounds: (i) to believe that Buyer's creditworthiness has become unsatisfactory; or (ii) for insecurity with respect to Buyer's performance under the Agreement, CESNG may demand, in writing, adequate assurance of future performance from Buyer in a form, in an amount, from an issuer, and for a term, all as reasonably satisfactory to CESNG ("Adequate Assurance of Performance"). To satisfy a demand, Buyer shall provide Adequate Assurance of Performance to CESNG within three (3) Business Days of the date of the written demand. If at any time CESNG requires Buyer to prepay for gas, then (i) CESNG shall be under no obligation to deliver gas if Buyer falls to pay any prepayment by its due date, and (ii) in the event the aggregate cost of any quantities of gas required by Buyer in a month in excess of the scheduled nominated quantities exceeds \$15,000.00, Buyer shall pay CESNG the aggregate cost of such quantities within three (3) Business Days of CESNG's request for payment.

The following sections shall be amended as follows:

- 10.2 Section 10.2 shall be amended by deleting the word "or" before "(ix)" in such Section; and adding the following immediately after the ";" in subclause (ix):
- "(x) make any representation or warranty herein which is false or misleading in any material respect when made or when deemed made or repeated; (xi) fail to perform any material covenant or obligation set forth in this Contract (except to the extent such failure constitutes a separate Event of Default, and except for such party's obligations to deliver or receive Gas (the exclusive remedy for which is provided in Section 3)) if such failure is not remedied within five (5) Business Days after receipt of written notice thereof; (xii) or consolidate or amalgamate with, or merge with or into, or transfer all or substantially all of its assets to, another entity and, at the time of such consolidation, amalgamation, merger or transfer, the resulting, surviving or transferee entity fails to assume all of the obligations of such party under the Contract; then"

Section 10.2 shall further be amended by adding the following immediately before the "." at the end of such Section:

"provided, that no suspension shall continue for more than ten (10) Business Days unless an Early Termination Date has been declared and the Defaulting Party has been given notice thereof in accordance with Section 10.3."

10.3.1 The following sentence shall be added at the end of Section 10.3.1: "In calculating its early termination damages hereunder, the Non-Defaulting Party may include its Costs."

SECTION 11. FORCE MAJEURE

11.6 Section 11.6 shall be deleted in its entirety.

SECTION 13. LIMITATIONS

Section 13 shall be deleted in its entirety and replaced with the following:

THIS SECTION DOES NOT WAIVE ANY LIMITATION ON INDEMNITY OR DAMAGES IMPOSED ON BUYER OR SELLER BY WISCONSIN STATE STATUTES. NOTWITHSTANDING ANY OTHER PROVISION OF THE CONTRACT TO THE CONTRARY, THE ENTIRE LIABILITY OF EACH PARTY FOR ANY AND ALL CLAIMS WILL BE LIMITED TO DIRECT ACTUAL DAMAGES ONLY, SUBJECT IN ALL CASES TO AN AFFIRMATIVE OBLIGATION ON THE PART OF EACH PARTY TO MITIGATE ITS DAMAGES, AND NEITHER PARTY WILL BE LIABLE FOR ANY CONSEQUENTIAL, EXEMPLARY, SPECIAL, INCIDENTAL OR PUNITIVE DAMAGES, INCLUDING, WITHOUT LIMITATION, LOST OPPORTUNITIES OR LOST PROFITS NOT CONTEMPLATED BY SECTION 10 ABOVE. The rights, powers, remedies and privileges provided in this Contract are cumulative and not exclusive of any rights, powers, remedies and privileges provided by law. Buyer acknowledges and agrees that the utility is exclusively responsible for the gas distribution and delivery system, that CESNG has no independent control over their systems and will have no liability for any of their acts or omissions.

SECTION 15. MISCELLANEOUS

The following sections shall be amended as follows:

Section 15.4 shall be deleted in its entirety and replaced with the following:

Supersedes Prior Agreements. The Contract embodies the parties' entire agreement and understanding and may not be contradicted by any prior or contemporaneous oral or written agreement. Notwithstanding any provision to the contrary in a prior agreement, the parties agree that this Base Contract shall supersede and replace all prior written and oral agreements or arrangements between the parties with respect to the purchase and sale of Gas and related services and that all transactions under any such prior agreement(s) are, as of the Effective Date, now governed solely by the terms of this Base Contract and shall be Transactions hereunder and a part of the single integrated agreement between the parties. This Contract may be amended only by a writing executed by both parties.

15.8 The following shall be added to the end of Section 15.8:

"On the effective date and the date of entering into each transaction, each party represents and warrants that: (i) it is duly organized, validly operating and in good standing under the laws of the jurisdiction of its formation; (ii) execution, delivery and performance of the Contract are duly authorized and do not violate any governing documents or any of its contracts or any applicable Law; (iii) it has the knowledge and experience to evaluate the merits and risks associated with the Contract; (iv) it is an "eligible contract participant" as that term is defined in the Commodity Exchange Act, as amended; (v) this Contract, each transaction hereunder, and each other document executed and delivered in accordance with this Contract constitutes a legally valid and binding obligation enforceable against it in accordance with the terms of said document, subject to any equitable defenses; (vi) it, or its credit support provider, if applicable, is not bankrupt and there are no proceedings pending or being contemplated by it, its credit support provider, if any, or, to its knowledge, threatened against it which would result in it being or becoming bankrupt and there is not pending or, to its knowledge, threatened against it, or its credit support provider, if any, any legal proceedings that could materially adversely affect its ability to perform its obligations under this Contract and each transaction hereunder; and (vii) no Event of Default with respect to it has occurred and is continuing and no such event or circumstance would occur as a result of its entering into or performing its obligations under this Contract and each transaction hereunder. The parties agree that this Contract constitutes a "qualified financial contract" as that term is defined in N.Y.G.O.L. §5-701(b) and a "master netting agreement" as defined in 11 U.S.C. 101(38A) or any successor provisions.

To the extent, if any, that a transaction does not qualify as a "first sale" as defined by the Natural Gas Act and §§ 2 and 601 of the Natural Gas Policy Act, each Party irrevocably waives its rights, including its rights under §§ 4-5 of the Natural Gas Act, unilaterally to seek or support a change to any terms and conditions of the Contract, including but not limited to the rate(s), charges, or classifications set forth therein. By this provision, each Party expressly waives its right to seek or support, either directly or indirectly, and by whatever means: (i) an order from the U.S. Federal Energy Regulatory Commission ("FERC") seeking to change any of the terms and conditions of the Contract agreed to by the Parties; and (ii) any refund from the other Party with respect to the Contract. Each Party further agrees that this waiver and covenant shall be binding upon it notwithstanding any regulatory or market changes that may occur after the date of the Contract or any transaction entered into between the Parties. Absent the agreement of both Parties to the proposed change, the standard of review for changes to any terms and conditions of the Contract proposed by (a) a Party, to the extent that the waiver set forth in this Section 15.8 is unenforceable or ineffective as to such Party due to a final determination being made under applicable law that precludes the Party from waiving its rights to seek or support changes from the FERC to the terms and conditions of this Contract, (b) a non-party, or (c) the FERC acting sua sponte, shall solely be the "public interest" application of the "just and reasonable" standard of review set forth in United Gas Pipe Line Co. v. Mobile Gas Service Corp., 350 U.S. 332 (1956) and Federal Power Commission v. Sierra Pacific Power Co., 350 U.S. 348 (1956) (the "Mobile-Sierra Doctrine"), as the MobileSierra Doctrine has been clarified by Morgan Stanley Capital Group, Inc. v. Public Util. Dist. No. 1 of Snohomish 128 S.Ct. 2733 (2008).

Each party represents to the other party (with such representations deemed to be repeated by each party on each date on which a relevant transaction is entered into) that, with respect to any commodity option transaction, it is: (1) a producer, processor, or commercial user of, or a merchant handling the commodity that is the subject of the option transaction; (2) offering or has been offered or is entering into the transaction solely for purposes related to its business as such; and (3) intends to make or take, as applicable, physical delivery of the commodity if the option is exercised.

Further, Buyer warrants, represents and covenants that: (a) the data given and representations made concerning its Facility(ies) are true and correct; (b) it is entering into this Contract to purchase its Gas requirements only and not for speculative or resale purposes, and that the natural gas purchased under this Contract will be consumed at the Facility(ies); (c) if any transaction hereunder gives Buyer the right to adjust the Contract Quantity, or to require CESNG to provide some other quantity of Gas, Buyer's election to exercise such right is based predominantly on supply and demand factors related to Buyer's business; (d) it is the party of record of the Facility(ies), or if it is not the party of record, it has the authority to enter into and bind its principal to the Contract; and (e) if Buyer is a Governmental Entity, it will not claim immunity on the grounds of sovereignty or similar grounds from enforcement of the Contract. For purposes of this Section, defenses identified in Wis. Stats. §§ 893.80 and 895.46 are not considered similar defenses to sovereign immunity. "Governmental Entity" means a municipality, county, governmental board, governmental department, commission, agency, bureau, administrative body, joint action agency, court or other similar political subdivision (including a public school district or special purpose district or authority), or public entity or instrumentality of the United States or one or more states."

- 15.10 Section 15.10 is amended by adding the words "and such party's affiliates" after the word "party" in the third line of such Section and by adding the parenthetical "(excluding the name of the other party)" after the word "information" in the sub-section (v).
- 15.11 Section 15.11 shall be deleted in its entirety and replaced with the following:

This Contract, and the rights and duties of the parties arising therefrom, shall be governed by, and interpreted and construed in accordance with, the law of the jurisdiction identified on the cover page of the Base Contract (without reference to choice of law doctrine). EACH PARTY HEREBY IRREVOCABLY WAIVES ANY AND ALL RIGHT TO TRIAL BY JURY

IN ANY PROCEEDING ARISING OUT OF OR RELATING TO THIS CONTRACT.

The following sections shall be added to Section 15:

- 15.13 Information and Authorization. Buyer shall designate CESNG to the utility and any other applicable Transporter(s) as an authorized recipient of Buyer's current and historical Gas billing and usage data. Buyer authorizes CESNG to take such actions it deems necessary to enroll the Buyer's accounts with the Utility as accounts to be served by CESNG solely for fulfilling CESNG's obligations under this Contract, including executing on Buyer's behalf any documents necessary to effectuate any account enrollment or election, undertaking the management of any storage or transportation capacity allocated to Buyer by the utility or other Transporter(s), disposing of storage balances, adding or deleting accounts as necessary, receiving usage nominations from the utility relating to Buyer's Gas requirements and other similar documents. Buyer shall take any actions and execute any documents as CESNG shall reasonably request in this regard.
- 15.14 Change in Law. CESNG may pass through or allocate, as the case may be, to Buyer any increase or decrease in CESNG's costs related to the natural gas and related products and services sold to Buyer that results from the implementation of new, or changes (including changes to transportation rates) to existing, Laws, or other requirements or changes in administration or interpretation of Laws or other requirements. "Law" means any law, rule, regulation, ordinance, statute, judicial decision, administrative order, Transporter business practices or protocol, Transporter tariff, or rule of any commission or agency with jurisdiction in the state in which the Facilities are located. Such adjusted amounts will be included in subsequent invoices to Buyer. 15.15 Transporter Restrictions. If either Party receives an operational flow order or any other usage or operating instructions, restrictions or similar notice (each an "OFO") from a Transporter requiring action to be taken in connection with the flow and/or consumption of gas covered by this Agreement, such party will use commercially reasonable efforts to notify the other party of such event in a timely

manner. Such notice may be given by electronic means. Each Party will take all commercially reasonable actions required by the OFO within the time prescribed. If OFO penalties, charges, fees, costs or expenses result from the actions or inactions of one Party, then such Party shall be solely responsible for any such penalties, charges, fees, costs or expenses. Both parties agree that OFOs may require one or both parties to buy or sell Gas quantities in the then-current market conditions, which may be appreciably higher or lower than the original pricing. If CEGS is responsible for nominations and balancing on a Transporter imposing an OFO or similar restriction, CEGS may (but is not required to) increase or decrease nominations, as appropriate, to avoid penalties.

IN WITNESS WHEREOF, the parties hereto have executed these Special Provisions as of the Effective Date.

CONSTELLATION ENERGY SERVICES - NATURAL GAS, LLC	MILWAUKEE BOARD OF SCHOOL DIRECTORS ON BEHALF OF
Name: David T Donat	MILWAUKÉE PUBLIC 80HOOLS By: Name: Darienne B. Driver, Ed D
	Title: Superintendent of Schools
	By: _ ha. S.
	Name: Mark A. Sain, President
	Title: Milwaukee Board of School Directors

Client#: 82241

EXELON

ACORD.

CERTIFICATE OF LIABILITY INSURANCE

DATE (MM/DD/YYYY) 6/03/2016

THIS CERTIFICATE IS ISSUED AS A MATTER OF INFORMATION ONLY AND CONFERS NO RIGHTS UPON THE CERTIFICATE HOLDER. THIS CERTIFICATE DOES NOT AFFIRMATIVELY OR NEGATIVELY AMEND, EXTEND OR ALTER THE COVERAGE AFFORDED BY THE POLICIES BELOW. THIS CERTIFICATE OF INSURANCE DOES NOT CONSTITUTE A CONTRACT BETWEEN THE ISSUING INSURER(S), AUTHORIZED REPRESENTATIVE OR PRODUCER, AND THE CERTIFICATE HOLDER.

IMPORTANT: If the certificate holder is an ADDITIONAL INSURED, the policy(ies) must be endorsed. If SUBROGATION IS WAIVED, subject to

the terms and conditions of the policy, certain policies m certificate holder in lieu of such endorsement(s).	nay require an endorsement. A statement on this certificate does not confer	ights to the						
PRODUCER	CONTACT Elizabeth Rodriguez	CONTACT Elizabeth Rodriguez						
P&C Chicago	PHONE (A/C, No, Ext): FAX (A/C, No): 312	-595-8169						
Mesirow Insurance Services	EMAIL ADDRESS: erodriguez@mesirowfinancial.com							
353 N. Clark Street	INSURER(S) AFFORDING COVERAGE	NAIC #						
Chicago, IL 60654	INSURER A : American Guarantee & Liability	26247						
INSURED	INSURER B: National Fire Ins Co of Hartfor	20478						
Constellation NewEnergy, Inc.	INSURER C:							
100 Constellation Way	INSURER D:							
Baltimore, MD 21202	INSURER E :							
	INSURER F:							
COVERAGES CERTIFICATE NUMBER	R: REVISION NUMBER:							
	ISTED BELOW HAVE BEEN ISSUED TO THE INSURED NAMED ABOVE FOR THE POI							
	OR CONDITION OF ANY CONTRACT OR OTHER DOCUMENT WITH RESPECT TO VIRANCE AFFORDED BY THE POLICIES DESCRIBED HEREIN IS SUBJECT TO ALL							
EXCLUSIONS AND CONDITIONS OF SUCH POLICIES. LIMITS								
INSR TYPE OF INSURANCE ADDLISUBR	POLICY EFF POLICY EXP							

R	TYPE OF INSURANCE	ADDL INSR	SUBR WVD	POLICY NUMBER	POLICY EFF (MM/DD/YYYY)	POLICY EXP (MM/DD/YYYY)	LIMIT	s
L	COMMERCIAL GENERAL LIABILITY						EACH OCCURRENCE	\$
L	CLAIMS-MADE OCCUR						DAMAGE TO RENTED PREMISES (Ea occurrence)	\$
							MED EXP (Any one person)	\$
L							PERSONAL & ADV INJURY	\$
ŀ	GEN'L AGGREGATE LIMIT APPLIES PER:					-	GENERAL AGGREGATE	\$
L	POLICY PRO- JECT LOC	ĺ				_	PRODUCTS - COMP/OP AGG	\$
1	OTHER:						COMPINED CONCLETIME	\$
	AUTOMOBILE LIABILITY			5082786753	11/01/2015	11/01/2016	COMBINED SINGLE LIMIT (Ea accident)	\$1,000,000
L	X ANY AUTO						BODILY INJURY (Per person)	\$
L	ALL OWNED SCHEDULED AUTOS						BODILY INJURY (Per accident)	\$
L	X HIRED AUTOS X AUTOS	1					PROPERTY DAMAGE (Per accident)	\$
1								\$
L	UMBRELLA LIAB OCCUR		1				EACH OCCURRENCE	\$
L	EXCESS LIAB CLAIMS-MADE						AGGREGATE	\$
1	DED RETENTION \$	ļ					loro loru	\$
	WORKERS COMPENSATION AND EMPLOYERS' LIABILITY Y/N		İ	WC931275104	01/05/2016	01/05/2017	X PER OTH- STATUTE · OTH-	
1	ANY PROPRIETOR/PARTNER/EXECUTIVE N	NIA					E.L. EACH ACCIDENT	\$1,000,000
ı	(Mandatory in NH) If yes, describe under				1		E.L. DISEASE - EA EMPLOYEE	\$1,000,000
1	DESCRIPTION OF OPERATIONS below	Ļ_					E.L. DISEASE - POLICY LIMIT	\$1,000,000
l								
ļ								
1		<u> </u>	<u> </u>					
	RIPTION OF OPERATIONS / LOCATIONS / VEHIO Professional Services Contract.	•	ACOR	O 101, Additional Remarks Schedule,	may be attached if mo	ore space is requ	ired)	
•	Froiessional Services Contract.							

RTIFICATE HOLDER	CANCELLATION
Milwaukee Board of School Directors 1124 N. 11th St.	SHOULD ANY OF THE ABOVE DESCRIBED POLICIES BE CANCELLED BEFOR THE EXPIRATION DATE THEREOF, NOTICE WILL BE DELIVERED IN ACCORDANCE WITH THE POLICY PROVISIONS.
Milwaukee, WI 53233	AUTHORIZED REPRESENTATIVE



CERTIFICATE OF LIABILITY INSURANCE

DATE (MM/DD/YYYY) 6/3/2016

THIS CERTIFICATE IS ISSUED AS A MATTER OF INFORMATION ONLY AND CONFERS NO RIGHTS UPON THE CERTIFICATE HOLDER. THIS CERTIFICATE DOES NOT AFFIRMATIVELY OR NEGATIVELY AMEND, EXTEND OR ALTER THE COVERAGE AFFORDED BY THE POLICIES BELOW. THIS CERTIFICATE OF INSURANCE DOES NOT CONSTITUTE A CONTRACT BETWEEN THE ISSUING INSURER(S), AUTHORIZED REPRESENTATIVE OR PRODUCER, AND THE CERTIFICATE HOLDER.

IMPORTANT: If the certificate holder is an ADDITIONAL INSURED, the policy(ies) must be endorsed. If SUBROGATION IS WAIVED, subject to the terms and conditions of the policy, certain policies may require an endorsement. A statement on this certificate does not confer rights to the certificate holder in lieu of such endorsement(s).

C	ertificate holder in lieu of such endors	cen seme	ain p ent(s)	ooncies may require an ei	ndorse	ment. A sta	tement on th	is certificate does not c	onfer r	ights to the	
PRO	DUCER		• •		CONTA NAME:	CT Jenna I	lanv				
Mike	Hogue				NAME: Jenna Hagy PHONE (A/C, No, Ext):405-235-6633 FAX (A/C, No):405-235-6634						
Artn	ur J. Gallagher Risk Management S 0 North Robinson Avenue	ervi	ces,	Inc.	E-Mail. Appress:Jenna Hagy@ajg.com						
Okla	shoma City OK 73103-4820				INSURER(S) AFFORDING COVERAGE					NAIC#	
										20281	
INSU	RED		INSURER B:								
Con	stellation Energy Services - Natural	Gas	s. LL	С	INSURER C:						
c/o	Exelon Corporation		•		INSURI						
Balt	9 Tangier Drive, Suite A imore MD 21220				INSURI				-		
					INSURI	***************************************					
	VERAGES CER	TIFK	CATE	NUMBER: 1643351167	,			REVISION NUMBER:			
TI	IS IS TO CERTIFY THAT THE POLICIES	OF	NSUF	RANCE LISTED BELOW HAY	/F REE	N ISSUED TO	THE INSURE	D NAMED AROVE FOR T	HE POL	JCY PERIOD	
1 IIN	DICATED. NOTWITHSTANDING ANY RE	OUIF	REME	NT. TERM OR CONDITION	OF AN	Y CONTRACT	OR OTHER	DOCUMENT WITH DECDE	CT TO 1	MALICH THIC	
E	ERTIFICATE MAY BE ISSUED OR MAY I CCLUSIONS AND CONDITIONS OF SUCH	POLI	CIES.	LIMITS SHOWN MAY HAVE	BEEN!	REDUCED BY	S DESCRIBEI PAID CLAIMS	D HEKEIN IS SUBJECT TO	J ALL I	THE TERMS,	
INSR LTR	TYPE OF INSURANCE	ADDL	SUBR WVD	POLICY NUMBER		POLICY EFF (MM/DD/YYYY)		LIMIT	·s		
Α	GENERAL LIABILITY	Υ	1110	35816976		8/1/2015	8/1/2016	EACH OCCURRENCE	\$1,000,	000	
	X COMMERCIAL GENERAL LIABILITY							DAMAGE TO RENTED			
İ	CLAIMS-MADE X OCCUR							PREMISES (Ea occurrence) MED EXP (Any one person)	\$1,000, \$10,000		
								PERSONAL & ADV INJURY	\$1,000,		
								GENERAL AGGREGATE	\$2,000,		
	GEN'L AGGREGATE LIMIT APPLIES PER:							PRODUCTS - COMP/OP AGG	\$2,000,		
	X POLICY PRO- JECT LOC							FRODUCTS - COMPTOP AGG	\$ \$	000	
A	AUTOMOBILE LIABILITY	Υ		74999277		8/1/2015	8/1/2016	COMBINED SINGLE LIMIT (Ea accident)	\$1,000,	000	
	X ANY AUTO							BODILY INJURY (Per person)	\$1,000,	000	
	ALL OWNED SCHEDULED AUTOS AUTOS							BODILY INJURY (Per accident)	\$		
	AUTOS AUTOS NON-OWNED AUTOS							PROPERTY DAMAGE (Per accident)	s		
								(Fer accide;(i)	\$		
Α	X UMBRELLA LIAB X OCCUR	Υ		79834313		8/1/2015	8/1/2016	EACH OCCURRENCE	\$1,000,	000	
	EXCESS LIAB CLAIMS-MADE							AGGREGATE	\$	000	
	DED RETENTION\$							7.00(120)(12	\$		
	WORKERS COMPENSATION AND EMPLOYERS' LIABILITY							WC STATU- OTH- TORY LIMITS ER	Ť		
	ANY PROPRIETOR/PARTNER/EXECUTIVE N	NΙA						E.L. EACH ACCIDENT	\$		
	(Mandatory in NH)	MIA						E.L. DISEASE - EA EMPLOYEE			
	If yes, describe under DESCRIPTION OF OPERATIONS below							E.L. DISEASE - POLICY LIMIT	\$		
	CRIPTION OF OPERATIONS / LOCATIONS / VEHICL	.ES (/	Attach .	ACORD 101, Additional Remarks S	Schedule	, if more space is	required)				
RE:	Professional Services Contract										
#42	ificate holder is included as Addition 02-0038 (8-89), Auto endorsement	181 II #16.	nsure .non	ed on the General Liabilit	ty, Aut	omobile and	Umbrella p	olicies, as per GL ende	orseme	ent	
cond	ditions and exclusions.	<i>n</i> -10	-02-0	232 (4-11), and Online	ia puis	suant to and	Subject to t	ne policy's terms, delin	mons,		
CEF	RTIFICATE HOLDER				CANO	ELLATION				·	
	Milwaukee Board of Schoo	l Din	ector	·c				ESCRIBED POLICIES BE C			
	1124 N. 11th St		00,01	Ĭ				EREOF, NOTICE WILL I	SE DEI	LIVERED IN	
	Milwaukee WI 53233										
					AUTHO	RIZED REPRESE	NTATIVE		,		
					0	CUI					
L	L				グウ	coll -					



CERTIFICATE OF LIABILITY INSURANCE

DATE (MM/DD/YYYY) 06/04/2016

THIS CERTIFICATE IS ISSUED AS A MATTER OF INFORMATION ONLY AND CONFERS NO RIGHTS UPON THE CERTIFICATE HOLDER. THIS CERTIFICATE DOES NOT AFFIRMATIVELY OR NEGATIVELY AMEND, EXTEND OR ALTER THE COVERAGE AFFORDED BY THE POLICIES BELOW. THIS CERTIFICATE OF INSURANCE DOES NOT CONSTITUTE A CONTRACT BETWEEN THE ISSUING INSURER(S), AUTHORIZED REPRESENTATIVE OR PRODUCER, AND THE CERTIFICATE HOLDER.

IMPORTANT: If the certificate holder is an ADDITIONAL INSURED, the policy(ies) must be endorsed. If SUBROGATION IS WAIVED, subject to the terms and conditions of the policy, certain policies may require an endorsement. A statement on this certificate does not confer rights to the certificate holder in lieu of such endorsement(s). PRODUCER Marsh USA Inc. CONTACT NAME: PHONE (A/C, No, Ext): E-MAIL 1717 Arch Street FAX (A/C, No): Philadelphia, PA 19103-2797 ADDRESS: INSURER(S) AFFORDING COVERAGE NAIC# S27339-NewEn-PROF-16-17 INSURER A : Lexington Insurance Company 19437 INSURED Constellation Energy Services-INSURER B: Natural Gas, LLC INSURER C: 124 W Broadway INSURER D Madison, WI 53716 INSURER E : INSURER F: COVERAGES CERTIFICATE NUMBER: CLE-005103281-01 **REVISION NUMBER:2** THIS IS TO CERTIFY THAT THE POLICIES OF INSURANCE LISTED BELOW HAVE BEEN ISSUED TO THE INSURED NAMED ABOVE FOR THE POLICY PERIOD INDICATED. NOTWITHSTANDING ANY REQUIREMENT, TERM OR CONDITION OF ANY CONTRACT OR OTHER DOCUMENT WITH RESPECT TO WHICH THIS CERTIFICATE MAY BE ISSUED OR MAY PERTAIN, THE INSURANCE AFFORDED BY THE POLICIES DESCRIBED HEREIN IS SUBJECT TO ALL THE TERMS, EXCLUSIONS AND CONDITIONS OF SUCH POLICIES. LIMITS SHOWN MAY HAVE BEEN REDUCED BY PAID CLAIMS. ADDL SUBR INSD WVD POLICY EFF POLICY EXP (MM/DD/YYYY) TYPE OF INSURANCE POLICY NUMBER LIMITS COMMERCIAL GENERAL LIABILITY EACH OCCURRENCE \$ CLAIMS-MADE DAMAGE TO RENTED PREMISES (Ea occurrence) MED EXP (Any one person) \$ PERSONAL & ADV INJURY \$ GEN'L AGGREGATE LIMIT APPLIES PER: GENERAL AGGREGATE \$ POLICY PRO-PRODUCTS - COMP/OP AGG S AUTOMOBILE LIABILITY COMBINED SINGLE LIMIT (Ea accident) \$ ANY AUTO BODILY INJURY (Per person) \$ SCHEDULED AUTOS NON-OWNED AUTOS ALL OWNED BODILY INJURY (Per accident) \$ PROPERTY DAMAGE (Per accident) HIRED AUTOS \$ ŝ **UMBRELLA LIAB** OCCUR EACH OCCURRENCE \$ **EXCESS LIAB** CLAIMS-MADE AGGREGATE \$ DED RETENTION \$ WORKERS COMPENSATION AND EMPLOYERS' LIABILITY STATUTE ANY PROPRIETOR/PARTNER/EXECUTIVE OFFICER/MEMBER EXCLUDED? (Mandatory in NH) E.L. EACH ACCIDENT N If yes, describe under DESCRIPTION OF OPERATIONS below E.L. DISEASE - EA EMPLOYEE \$ E.L. DISEASE - POLICY LIMIT Professional Liability 023058104 05/19/2016 05/19/2017 Limit 1,000,000 Architects / Engineers SIR 200,000 DESCRIPTION OF OPERATIONS / LOCATIONS / VEHICLES (ACORD 101, Additional Remarks Schedule, may be attached if more space is required) **CERTIFICATE HOLDER** CANCELLATION Milwaukee Board of School Directors SHOULD ANY OF THE ABOVE DESCRIBED POLICIES BE CANCELLED BEFORE 1124 N. 11th St. THE EXPIRATION DATE THEREOF, NOTICE WILL BE DELIVERED IN Milwaukee, WI 53233 ACCORDANCE WITH THE POLICY PROVISIONS. **AUTHORIZED REPRESENTATIVE** of Marsh USA Inc. Manashi Mukherjee Mariaoni Mukerenjee

Purchase Requisition Number:
Contract Number:
Vendor Number:

MILWAUKEE BOARD OF SCHOOL DIRECTORS PROFESSIONAL SERVICES CONTRACT MODIFICATION AND EXTENSION AGREEMENT

THIS MODIFICATION AND EXTENSION AGREEMENT ("Extension Agreement") is made as of the 12th day of April , 2019, by and between THE MILWAUKEE BOARD OF SCHOOL DIRECTORS (hereinafter "MPS") and CONSTELLATION NEWENERGY – GAS DIVISION, LLC (as successor by merger with Constellation Energy Services Natural Gas, LLC (hereinafter "Supplier").

WHEREAS, on May 4, 2016, MPS and Supplier entered into Professional Services Contract with a term of July 1, 2016 through June 30, 2019 ("Contract"); and

WHEREAS, the Contract consisted of (1) MPS' Professional Services Contract ("PSC"), (2) the Supplier's NAESB Base Contract for Sale and Purchase of Natural Gas ("NAESB Agreement"), and (3) Supplier's Retail Special Provisions to the NAESB Base Contract for Sale and Purchase of Natural Gas ("Special Provisions"); and

WHEREAS, Section 2 of the Contract provides for up to three additional one-year extensions subject to funding and an annual performance review; and

WHEREAS, the parties wish to modify and extend the Contract for a single additional three-year term, from July 1, 2019 through June 30, 2022, under the same terms and conditions as set forth in the original Contract, except for those specifically modified by this Extension Agreement;

NOW, THEREFORE, the parties hereto agree as follows (all capitalized terms used but not defined herein shall have the meaning set forth in the Contract):

1. Section 2 of the Contract is deleted in its entirety and replaced with the following:

"Subject to Section 12 of the NAESB Agreement, the Term of this Contract shall be July 1, 2016 through June 30, 2019, with one possible three-year extension subject to funding and a performance review.

No work shall commence before a Supplier receives a fully executed Contract and has been given approval to proceed. Any work performed by the Supplier prior to obtaining a fully-executed contract with approval to proceed shall not be compensated pursuant to this contract. Any continuation of the contract beyond this term must be set forth in writing and signed by authorized signatories on behalf of the Party for which it is executed."

2. Section 7 of the Contract is deleted in its entirety and replaced with the following:

"Supplier understands and agrees that financial responsibility for claims or damages to any person, or to Supplier's employees and agents, shall rest with the Supplier. Supplier and its subSuppliers

Purchase Requisition Number:
Contract Number:
Vendor Number:

shall effect and maintain any insurance coverage, including, but not limited to, Workers' Compensation, Employers' Liability, General Liability, Contractual Liability, Professional Liability, Automobile Liability and Umbrella Liability to support such financial obligations. The indemnification obligation, however, shall not be reduced in any way by existence or non-existence, limitation, amount or type of damages, compensation, or benefits payable under Workers' Compensation laws or other insurance provisions.

The minimum limits of insurance required of the Supplier by MPS shall be:

Workers' Compensation
Employers' Liability
General Liability
Professional Liability
Statutory Limits
\$100,000 per occurrence
\$1,000,000 per occurrence
\$1,000,000 per occurrence
\$1,000,000 per occurrence
\$1,000,000 per occurrence

Umbrella (exess) Liability \$1,000,000 per occurrence

The Milwaukee Board of School Directors shall be named as an additional insured under Supplier's and subSuppliers' general liability insurance and umbrella liability insurance. Evidence of all required insurances of Supplier shall be given to MPS. The certificate of insurance or policies of insurance evidencing all coverages shall include a statement that MPS shall be afforded a thirty (30) day written notice of cancellation, non-renewal or material change by any of Supplier's insurers providing the coverages required by MPS for the duration of this contract."

- 3. The parties hereby extend the Term of the Contract, as modified by this Extension Agreement, for one three-year term, from July 1, 2019 through June 30, 2022 ("Extension Term").
- 4. The NAESB Agreement to be in effect during the Extension Term shall be the NAESB Agreement attached hereto as Appendix A and incorporated into the Contract by reference ("NAESB Agreement in Extension Term").
- 5. The Special Provisions to be in effect during the Extension Term shall be the Special Provisions attached hereto as Appendix B and incorporated into the Contract by reference ("Special Provisions in Extension Term").
- 6. The Contract Quantity per year is 857,700 dekatherms during the Extension Term and is estimated not to exceed \$6,000,000 per year without Board approval and is based on historical usage and consideration of worst case scenarios. MPS shall be responsible for payment of actual volumes which may be higher or lower based on actual use by MPS and will be purchased or sold at prevailing market rates. All invoicing and payment obligations during the Extension Term shall be governed by the NAESB Agreement in Extension Term and Special Provisions in Extension Term. The total cost of transportation, daily balancing and basis during the Extension Term is NYMEX plus \$0.89 per dekatherm (Dth).

Purchase Requisition Number:
Contract Number:
Vendor Number:

Milwaukee Board of School Directors

IN WITNESS WHEREOF, the parties hereto have executed this Extension Agreement on the date first written above.

cellation NewEnergy – Gas Division, LLC, LLC	Milwaukee Board of School Directors
Dated: 04-24-19	Dated: 03-29-19
By: Mary Mars	By: My Manual (Principal or Administrator Signature)
	,
Address: <u>9400 Bunsen Parkway, Suite 100</u>	50 m
Louisville, KY 40220	By: 55 Frank B. CON
Phone Number: <u>(502) 426-4500</u>	Keith P. Posley, Ed.D.
	Superintendent of Schools
Tax ID o	By: Jan Mll 6/13/19
	Larry Miller, President

TRANSACTION CONFIRMATION

Page 1 of 2

This Transaction Confirmation dated May 4, 2019 is made a part of, and pursuant to terms of, the Base Contract for Sale and Purchase of Gas dated May 4, 2016 between Buyer and Seller identified herein (the "NAESB').

SELLER:

Constellation NewEnergy – Gas Division, LLC (as successor by merger to Constellation Energy Services Natural Gas, LLC)

1716 Lawrence Drive PO Box6025

De Pere, WI 54115-6025 Attn: Account Management E-mail: amwi@constellation.com

Phone: 888-367-4403 **(Option** 2)

Fax: 443-213-6116

BUYER:

Milwaukee Board of School Directors on behalf of Milwaukee Public Schools

1124 N. 11th St. Milwaukee WI 53233

Attn: Yuriy Indman, PE, CEM

E-mail: indmany@milwaukee.k12.wi.us

Phone: 414-416-5112 Agent / Assoc: Julie Cowling

Account Executive:

Delivery Period: July 1, 2019 to June 30, 2022; provided however, the Delivery Period shall begin on the later of, (i) the day after the first meter read during the period noted above, or (ii) upon acceptance by the utility of Seller as Buyer's gas supplier.

Contract Quantity (in dekatherms): see table

١	Month	Quantity	Month	Quantity	Month	Quantity	Month	Quantity
١	Jul	5,600	Oct	46,000	Jan	165,000	Apr	66,000
١	Aug	5,100	Nov	95,800	Feb	149,000	May	28,000
١	Sep	8,500	Dec	162,000	Mar	117,000	Jun	9,000
- 1								

Price: NYMEX Last Day Settle ("NYM LDS") plus Market Basis per dekatherm. Either or both of the price components may be fixed for specific quantities and delivery months with mutual agreement of the Parties. The Basis is fixed at \$0.89 per dekatherm.

Performance Obligation: Up to 7,300 dkthm/day shall be Firm. The remainder shall be Interruptible

Delivery Point: Buyer's Transporter Citygate

Buyer's Transporter: WE Energies

- SPECIAL PROVISIONS -

Balancing Services

1. A Balancing Service charge of \$0.03 per dekatherm applies in addition to the Price.

Buyer and Seller shall use commercially reasonable efforts to provide each other with sufficient notice to allow timely and accurate nominations in accordance with the nomination requirements of that Transporter, as they may change from time to time due to Constraint Day requirements or otherwise. Buyer accepts the following responsibilities: Buyer hereby appoints Seller as Its agent for the purposes of balancing, Including the receipt of current and historic Usage, storage, billing and transportation data from Transporter. Buyer agrees to assist Seller in the timely collection of data directly and through Transporter, notify Seller of usage expectations and any material changes to those usage expectations, and comply with any Daily Limitations concerning use. If Usage data is unavailable electronically, Buyer shall obtain usage data manually and deliver that information weekly via facsimile or electronic mail, or as requested by Seller.

- 2. Seller agrees to monitor and review all information provided to the Seller by Buyer and Transporter regarding Buyer's gas Use. Based on such information, Seller shall use commercially reasonable effort to keep Buyer within the balancing parameters established by Transporter. Seller shall pass through Io Buyer and Buyer shall pay for any Imbalance Charges and/or Cashout resulting from Buyer's failure to (I) fulfill its obligations set forth in paragraph 1 above, (ii) maintain telemetry, and/or (iii) meet Daily Limitations with respect to Use.
- 3. Mid-month variations to delivered quantity. In the event that the quantity of natural gas that can be delivered to Buyer (without the Transporter assessing Imbalance Charges or Cashout) is less than the quantity agreed to be purchased for the relevant period, Buyer shall remain liable for the entire quantity It agreed to purchase, however, Seller will provide a credit at prevailing market rates for that portion of the purchased quantity that Is not delivered. In the event that Buyer's expected Use will exceed the quantity of gas purchased by Buyer and/or available to Buyer without incurring Imbalance Charges for the relevant period, Seller will use commercially reasonable efforts to provide Incremental supply, plus Fuel (if applicable) at prevailing market rates. Credits for reductions in delivered quantity and/or charges for Incremental supply shall appear as separate line items on Seller's invoice to Buyer, including applicable price and quantity.

NAESB Transaction Confirmation 05-2015

ACCOUNTS:

[X] Check here if Attachment A Is attached and made a part of this Transaction Confirmation.

TRANSACTION CONFIRMATION

Page 2 of 2

Telemetry Obligations. Buyer acknowledges and agrees that it shall be responsible for installing and maintaining a meter telephone line. Buyer shall also be responsible for any charges imposed by a contractor for repairing the meter telephone line. As a convenience to Buyer, Seller may arrange for telemetry service or repairs to be provided and Buyer agrees it shall be responsible for any costs associated with such service and repairs.

"Constraint Day" means any period of time during which a Transporter issues balancing orders, operational now orders, operational matching orders (or the equivalent), or winter splits or deems the day a Critical Gas Day (or the equivalent) and thereby restricts deliveries by Seller or use by Buyer, requires a specific delivery by Seller or use by Buyer, or otherwise requires an affirmative action by either Party due to adverse operational conditions experienced by the Transporter.

"Daily Limitations' means Transporter enforced restrictions or mandates concerning quantities of gas delivered and/or used, including those imposed due to Constraint Day conditions and/or under ordinary operating conditions.

- Invoices. Invoices shall be prepared bas dupon quantities used.
- Consolidated Billing: Seller shall include the Transporter distribution fees on the monthly invoice. Past due and other charges due Transporter, that apply to service prior to the effective date of the Confirmation will not be Included in Seller's invoice to Buyer. Seller may discontinue this service at any time with notice to Buyer.
- Term: This Transaction Confirmation is effective for the Delivery Period as it may be extended or renewed as described below.
- Extending Deli very Period: Should the Parties agree to a fixed price for any pricing component for a delivery month (or billing cycle, as applicable) that occurs after the current Delivery Period, then the Delivery Period of this Transaction Confirmation shall be extended to the latest occurring delivery month (or billing cycle, as applicable) with a fixed price or fixed price component.
- Renewal: The Parties agree that this Transaction Confirmation shall automatically renew for successive twelve (12) month periods (each constituting a "Renewal Delivery Period") unless written notice pursuant to Section 9 of the Agreement terminating the Transaction Confirmation at the end of a Delivery Period Is given by either Party thirty (30) days prior to the end of the Delivery Perold. The Contract Quantity during any Renewal Delivery Period shall be the Contract Quantity stated in the table above for the respective month. The Price for the Contract Quantity during any Renewal Delivery Period shall be NYMEX LDS plus Market Basis.

All other terms and conditions of this Transaction Confirmation shall remain unchanged.

Special Condition: This Confirmation is not effective and binding upon the parties hereto unless signed concurrently with the Professional Services Contract Modification and Extension Agreement dated effective April 12, 2019:

Seller: Constellation NewEnergy - Gas Division, LLC

Signature:

By: Daniel Marks

Title: Executive Dir, Retail Gas Ops

Date: April 24, 2019

Buyer: Milwaukee Board of Schools Directors on behalf of Milwaukee Public Schools

100 B. 600h

Ву:

Superintendent of Schools Title:

5/29/19 Date:

By:

President, Milwaukee Board of School Directors

Date:

APPENDIX A

Base Contract for Sale and Purchase of Natural Gas

This Base Contract is entered into as of the following date: May 4, 2016 ("Effective Date")

The parties to this Base Contract are the following:

PARTY A Constellation Energy Services – Natural Gas, LLC ("CESNG" or "Seller")	PARTYNAME	PARTY B Milwaukee Board of School Directors on behalf of Milwaukee Public Schools ("Buyer")
9960 Corporate Campus Drive, Suite 2000 Louisville, KY 40223	ADDRESS	5225 W. Vilet Street Milwaukee, Wi 53208
	BUSINESS WEBSITE	
CESNG19189	CONTRACT NUMBER	
	D-U-N-SQ NUMBER	
D US FEDÉRAL: D OTHER:	TAX ID NUMBERS	o us federal: o other:
Wisconsin	JURISDICTION OF ORGANIZATION	•
□ Corporation ☑ LLC □ Limited Partnership □ Partnership □ LLP □ Other:	COMPANY TYPE GUARANTOR	Corporation
COL	(IF APPLICABLE) NTACT INFORMAT	ION
8960 Corporate Campus Drive, Suite 2000, Louisville, KY 40223 ATTN: Contracts Department TEL#: (502) 428-4500 FAX#: (502) 214-8381	COMMERCIAL (If ditierent from Contract/Legal notices)	ATTN: TEL#; FAX#; EMAIL:
9960 Corporate Campus Drive, Suite 2000, Louisville, KY 40223 ATTN: Contracts Department TEL#: (502) 426-4500 FAX#: (502) 214-6381	➤ SCHEDULING (if different from ContractA.egal nolices)	ATTN: YEL#: FAX#: EMAIL:
9960 Corporate Campus Drive, Sulte 2000, Louisville, KY 40223 ATTN: Contracts Department TEL#: (502) 426-4500 FAX#: (502) 214-5381	CONTRACT AND LEGAL NOTICES	1224 North 11" Street, Milwaukee, WI 53233 ATTN: John Linn TEL#: (414)-283-4600 FAX#: (414) 283-4682 EMAIL: IInnia@milwaukee.k12.wl.us
9960 Corporate Campus Drive, Sulte 2000, Louisville, KY 40223 ATTN: Contracte Department TEL#: (502) 426-4500 FAX#: (502) 214-8381	CREDIT (if different from Contract/Legal notices)	nia ATTN: TEL#: FAX#: EMAIL:
9980 Corporate Campus Drive, Suite 2000, Louisville, KY 40223 ATTN: Contracts Department TEL#: (502) 426-4500 FAX#: (502) 214-6381	TRANSACTION CONFIRMATIONS (if different from Contract/Legal notices)	ATTN: TEL#: FAX#: EMAIL:
ACCC	UNTING INFORM	ATION
ATTN: see invoice TEL#: FAX#:	INVOICES PAYMENTS SETTLEMENTS	ATTN: TELH: FAXH: EMAIL:
BANK: see invoice ABA: ACCT: OTHER DETAILS:	WIRE TRANSFER NUMBERS (IF APPLICABLE)	BANK: NA ABA: ACCT: OTHER DETAILS:
BANK: gee invoice ABA: ACCT:	ACH NUMBERS (IF APPLICABLE)	BANK: <u>N/A</u> ABA: ACCT: OTHER DETAILS:
ATTN: see invoice Address:	CHECKS (IF APPLICABLE)	ATTN:N/A ADDRESS:

Base Contract for Sale and Purchase of Natural Gas

(Continued)

This Base Contract incorporates by reference for all purposes the General Terms and Conditions for Sale and Purchase of Natural Gas published by the North American Energy Standards Board. The parties hereby agree to the following provisions offered in said General Terms and Conditions. In the event the parties fall to check a box, the specified default provision shall apply. Select the appropriate box(es) from each section:

Section 1.2 Transaction Procedure	⊠ OR □	Oral (defauli) Written	Section 10.2 Additional Events of Default	Ø O	No Additional Events of Default (default) Indebtedness Cross Default	
Section 2.7 Confirm Deadline	<u> </u>	2 Business Days after receipt (default) Business Days after receipt	Solution		Parly A: Parly B: Transactional Cross Default Specified Transactions:	
Section 2.8 Confirming Parly	OR O	Seller (defaull) Buyer				
Section 3.2 Performance Obligation	OR O	Cover Standard (default) Spot Price Standard	Section 10.3.1 Early Termination	⊠ OR	Early Termination Damages Apply (default)	
Obligation	J	opsi 24	Damages		Early Termination Damages Do Not Apply	
Note: The followin immediately prece	g Spo ding.	ot Price Publication applies to both of the	Section 10.3.2	X	Other Agreement Setoffs Apply (default)	
Section 2,31 Spot Price	⊠ OR	Gas Dally Midpoint (default)	Olher Agreement Setoffs		⊠ Bilateral (default) □ Triangular	
Publication			·			
Section 6 Taxes	×	Buyer Pays At and After Delivery Point (default)			Other Agreement Setoffs Do Not Apply	
	OF	Seller Pays Before and At Delivery Point				
Section 7.2 Payment Date	×	25 th Day of Month following Month of delivery (default)	Section 15.5 Choice Of Law		Wisconsin	
	а	10 days following date of invoice				
Section 7.2 Method of Payment	□ 1 □. ⊠		Section 15.10 Confidentially	図 OR ロ	Confidentiality applies (default) Confidentiality does not apply	
Section 7.7 Nelling	OF D	Netting does not apply				
☑ Special Provision ☐ Addendum(s):	ons Ņ	lumber of sheets attached <u>3</u>				

IN WITNESS WHEREOF, the parties hereto have executed this Base Contract as of the Effective Date.

VAUKEE PUBLIC SCHOOLS
Don
ID. (45)
ools
a.S.
nt
chool Directors

Copyright © 2006 North American Energy Standards Board, Inc. All Rights Reserved. Prepared: CSP Non-Standard: Legal Revisions

CESNG19189 CONFIDENTIAL

General Terms and Conditions Base Contract for Sale and Purchase of Natural Gas

SECTION 1. PURPOSE AND PROCEDURES

1.1. These General Terms and Conditions are intended to facilitate purchase and sale transactions of Gas on a Firm or interruptible basis. "Buyer" refers to the party receiving Gas and "Seller" refers to the party delivering Gas. The entire agreement between the parties shall be the Contract as defined in Section 2.9.

The parties have selected either the "Oral Transaction Procedure" or the "Written Transaction Procedure" as indicated on the Base Contract.

Oral Transaction Procedure:

1.2. The parties will use the following Transaction Confirmation procedure. Any Gas purchase and sale transaction may be effectuated in an EDI transmission or telephone conversation with the offer and acceptance constituting the agreement of the parties. The parties shall be legally bound from the time they so agree to transaction terms and may each rely thereon. Any such transaction shall be considered a "writing" and to have been "signed". Notwithstanding the foregoing sentence, the parties agree that Confirming Party shall, and the other party may, confirm a telephonic transaction by sending the other party a Transaction Confirmation by facsimile, EDI or mutually agreeable electronic means within three Business Days of a transaction covered by this Section 1.2 (Oral Transaction Procedure) provided that the failure to send a Transaction Confirmation shall not invalidate the oral agreement of the parties. Confirming Party adopts its confirming letterhead, or the like, as its signature on any Transaction Confirmation as the identification and authentication of Confirming Party. If the Transaction Confirmation contains any provisions other than those relating to the commercial terms of the transaction (i.e., price, quantity, performance obligation, delivery point, period of delivery and/or transportation conditions), which modify or supplement the Base Contract or General Terms and Conditions of this Contract (e.g., arbitration or additional representations and warranties), such provisions shall not be deemed to be accepted pursuant to Section 1.3 but must be expressly agreed to by both parties; provided that the foregoing shall not invalidate any transaction agreed to by the parties.

Written Transaction Procedure:

- 1.2. The parties will use the following Transaction Confirmation procedure. Should the parties come to an agreement regarding a Gas purchase and sale transaction for a particular Delivery Period, the Confirming Party shall, and the other party may, record that agreement on a Transaction Confirmation and communicate such Transaction Confirmation by facsimile, EDI or mutually agreeable electronic means, to the other party by the close of the Business Day following the date of agreement. The parties acknowledge that their agreement will not be binding until the exchange of nonconflicting Transaction Confirmations or the passage of the Confirm Deadline without objection from the receiving party, as provided in Section 1.3.
- 1.3. If a sending party's Transaction Confirmation is materially different from the receiving party's understanding of the agreement referred to in Section 1.2, such receiving party shall notify the sending party via facsimile, EDI or mutually agreeable electronic means by the Confirm Deadline, unless such receiving party has previously sent a Transaction Confirmation to the sending party. The failure of the receiving party to so notify the sending party in writing by the Confirm Deadline constitutes the receiving party's agreement to the terms of the transaction described in the sending party's Transaction Confirmation. If there are any material differences between timely sent Transaction Confirmations governing the same transaction, then neither Transaction Confirmation shall be binding until or unless such differences are resolved including the use of any evidence that clearly resolves the differences in the Transaction Confirmations. In the event of a conflict among the terms of (i) a binding Transaction Confirmation pursuant to Section 1.2, (ii) the oral agreement of the parties which may be evidenced by a recorded conversation, where the parties have selected the Oral Transaction Procedure of the Base Contract, (iii) the Base Contract, and (iv) these General Terms and Conditions, the terms of the documents shall govern in the priority listed in this sentence.
- 1.4. The parties agree that each party may electronically record all telephone conversations with respect to this Contract between their respective employees, without any special or further notice to the other party. Each party shall obtain any necessary consent of its agents and employees to such recording. Where the parties have selected the Oral Transaction Procedure in Section 1.2 of the Base Contract, the parties agree not to contest the validity or enforceability of telephonic recordings entered into in accordance with the requirements of this Base Contract.

SECTION 2. DEFINITIONS

The terms set forth below shall have the meaning ascribed to them below. Other terms are also defined elsewhere in the Contract and shall have the meanings ascribed to them herein.

- 2.1. "Additional Event of Default" shall mean Transactional Cross Default or Indebtedness Cross Default, each as and if selected by the parties pursuant to the Base Contract.
- 2.2. "Affiliate" shall mean, in relation to any person, any entity controlled, directly or indirectly, by the person, any entity that controls, directly or indirectly, the person or any entity directly or indirectly under common control with the person. For this purpose, "control" of any entity or person means ownership of at least 50 percent of the voting power of the entity or person.
- 2.3. "Alternative Damages" shall mean such damages, expressed in dollars or dollars per MMBtu, as the parties shall agree upon in the Transaction Confirmation, in the event either Seller or Buyer falls to perform a Firm obligation to deliver Gas in the case of Seller or to receive Gas in the case of Buyer.

- 2.4. "Base Contract" shall mean a contract executed by the parties that incorporates these General Terms and Conditions by reference; that specifies the agreed selections of provisions contained herein; and that sets forth other information required herein and any Special Provisions and addendum(s) as identified on page one.
- 2.5. "Brilish thermal unit" or "Blu" shall mean the International BTU, which is also called the Blu (IT).
- 2.6. "Business Day(s)" shall mean Monday through Friday, excluding Federal Banking Holidays for transactions in the U.S.
- 2.7. "Confirm Deadline" shall mean 5:00 p.m. In the receiving party's time zone on the second Business Day following the Day a Transaction Confirmation is received or, if applicable, on the Business Day agreed to by the parties in the Base Contract; provided, if the Transaction Confirmation is time stamped after 5:00 p.m. in the receiving party's time zone, it shall be deemed received at the opening of the next Business Day.
- 2.8. "Confirming Party" shall mean the party designated in the Base Contract to prepare and forward Transaction Confirmations to the other party.
- 2.9. "Contract" shall mean the legally-binding relationship established by (i) the Base Contract, (ii) any and all binding Transaction Confirmations and (iii) where the parties have selected the Oral Transaction Procedure in Section 1.2 of the Base Contract, any and all transactions that the parties have entered into through an EDI transmission or by telephone, but that have not been confirmed in a binding Transaction Confirmation, all of which shall form a single integrated agreement between the parties.
- 2.10. "Contract Price" shall mean the amount expressed in U.S. Dollars per MMBtu to be paid by Buyer to Seller for the purchase of Gas as agreed to by the parties in a transaction.
- 2.11. "Contract Quantity" shall mean the quantity of Gas to be delivered and taken as agreed to by the parties in a transaction.
- 2.12. "Cover Standard", as referred to in Section 3.2, shall mean that if there is an unexcused failure to take or deliver any quantity of Gas pursuant to this Contract, then the performing party shall use commercially reasonable efforts to (i) if Buyer is the performing party, obtain Gas, (or an alternate fuel if elected by Buyer and replacement Gas is not available), or (ii) if Seller is the performing party, sell Gas, in either case, at a price reasonable for the delivery or production area, as applicable, consistent with: the amount of notice provided by the nonperforming party; the immediacy of the Buyer's Gas consumption needs or Seller's Gas sales requirements, as applicable; the quantities involved; and the anticipated length of failure by the nonperforming party.
- 2.13. "Credit Support Obligation(s)" shall mean any obligation(s) to provide or establish credit support for, or on behalf of, a party to this Contract such as cash, an irrevocable standby letter of credit, a margin agreement, a prepayment, a security interest in an asset, guaranty, or other good and sufficient security of a continuing nature.
- 2.14. "Day" shall mean a period of 24 consecutive hours, coextensive with a "day" as defined by the Receiving Transporter in a particular transaction.
- 2.15. "Delivery Period" shall be the period during which deliveries are to be made as agreed to by the parties in a transaction.
- 2.16. "Delivery Point(s)" shall mean such point(s) as are agreed to by the parties in a transaction.
- 2.17. "EDI" shall mean an electronic data interchange pursuant to an agreement entered into by the parties, specifically relating to the communication of Transaction Confirmations under this Contract.
- 2.18. "EFP" shall mean the purchase, sale or exchange of natural Gas as the "physical" side of an exchange for physical transaction involving gas futures contracts. EFP shall incorporate the meaning and remedies of "Firm", provided that a party's excuse for nonperformance of its obligations to deliver or receive Gas will be governed by the rules of the relevant futures exchange regulated under the Commodity Exchange Act.
- 2.19. "Firm" shall mean that either party may interrupt its performance without liability only to the extent that such performance is prevented for reasons of Force Majeure; provided, however, that during Force Majeure interruptions, the party invoking Force Majeure may be responsible for any Imbalance Charges as set forth in Section 4.3 related to its interruption after the nomination is made to the Transporter and until the change in deliveries and/or receipts is confirmed by the Transporter.
- 2.20. "Gas" shall mean any mixture of hydrocarbons and noncombustible gases in a gaseous state consisting primarily of methane.
- "Guarantor" shall mean any entity that has provided a guaranty of the obligations of a party hereunder.
- 2.22. "Imbalance Charges" shall mean any fees, penalties, costs or charges (in cash or in kind) assessed by a Transporter for failure to satisfy the Transporter's balance and/or nomination requirements.
- 2.23. "Indebtedness Cross Default" shall mean if selected on the Base Contract by the parties with respect to a party, that it or its Guarantor, if any, experiences a default, or similar condition or event however therein defined, under one or more agreements or instruments, individually or collectively, relating to indebtedness (such indebtedness to include any obligation whether present or future, contingent or otherwise, as principal or surety or otherwise) for the payment or repayment of borrowed money in an aggregate amount greater than the threshold specified in the Base Contract with respect to such party or its Guarantor, if any, which results in such indebtedness becoming immediately due and payable.
- 2.24. "Interruptible" shall mean that either party may interrupt its performance at any time for any reason, whether or not caused by an event of Force Majeure, with no liability, except such interrupting party may be responsible for any imbalance Charges as set forth in Section 4.3 related to its interruption after the nomination is made to the Transporter and until the change in deliveries and/or receipts is confirmed by Transporter.

- 2.25. "MMBtu" shall mean one million British thermal units, which is equivalent to one dekatherm.
- 2.26. "Month" shall mean the period beginning on the first Day of the calendar month and ending immediately prior to the commencement of the first Day of the next calendar month.
- 2.27. "Payment Date" shall mean a date, as indicated on the Base Contract, on or before which payment is due Seller for Gas received by Buyer in the previous Month.
- 2.28. "Receiving Transporter" shall mean the Transporter receiving Gas at a Delivery Point, or absent such receiving Transporter, the Transporter delivering Gas at a Delivery Point.
- 2.29. "Scheduled Gas" shall mean the quantity of Gas confirmed by Transporter(s) for movement, transportation or management.
- 2.30. "Specified Transaction(s)" shall mean any other transaction or agreement between the parties for the purchase, sale or exchange of physical Gas, and any other transaction or agreement identified as a Specified Transaction under the Base Contract.
- 2.31. "Spot Price" as referred to in Section 3.2 shall mean the price listed in the publication indicated on the Base Contract, under the listing applicable to the geographic location closest in proximity to the Delivery Point(s) for the relevant Day; provided, if there is no single price published for such location for such Day, but there is published a range of prices, then the Spot Price shall be the average of such high and low prices. If no price or range of prices is published for such Day, then the Spot Price shall be the average of the following: (i) the price (determined as stated above) for the first Day for which a price or range of prices is published that next precedes the relevant Day; and (ii) the price (determined as stated above) for the first Day for which a price or range of prices is published that next follows the relevant Day.
- 2.32. "Transaction Confirmation" shall mean a document, similar to the form of Exhibit A, setting forth the terms of a transaction formed pursuant to Section 1 for a particular Delivery Period.
- 2.33. "Transactional Cross Default" shall mean if selected on the Base Contract by the parties with respect to a party, that it shall be in default, however therein defined, under any Specified Transaction.
- 2.34. "Termination Option" shall mean the option of either party to terminate a transaction in the event that the other party fails to perform a Firm obligation to deliver Gas in the case of Selier or to receive Gas in the case of Buyer for a designated number of days during a period as specified on the applicable Transaction Confirmation.
- 2.35. "Transporter(s)" shall mean all Gas gathering or pipeline companies, or local distribution companies, acting in the capacity of a transporter, transporting Gas for Seller or Buyer upstream or downstream, respectively, of the Delivery Point pursuant to a particular transaction.

SECTION 3. PERFORMANCE OBLIGATION

3.1. Seller agrees to sell and deliver, and Buyer agrees to receive and purchase, the Contract Quantity for a particular transaction in accordance with the terms of the Contract. Sales and purchases will be on a Firm or Interruptible basis, as agreed to by the parties in a transaction.

The parties have selected either the "Cover Standard" or the "Spot Price Standard" as indicated on the Base Contract.

Cover Standard:

3.2. The sole and exclusive remedy of the parties in the event of a breach of a Firm obligation to deliver or receive Gas shall be recovery of the following: (i) in the event of a breach by Seller on any Day(s), payment by Seller to Buyer in an amount equal to the positive difference, if any, between the purchase price paid by Buyer utilizing the Cover Standard and the Contract Price, adjusted for commercially reasonable differences in transportation costs to or from the Delivery Point(s), multiplied by the difference between the Contract Quantity and the quantity actually delivered by Seller for such Day(s) excluding any quantity for which no replacement is available; or (ii) In the event of a breach by Buyer on any Day(s), payment by Buyer to Seller in the amount equal to the positive difference, if any, between the Contract Price and the price received by Seller utilizing the Cover Standard for the resale of such Gas, adjusted for commercially reasonable differences in transportation costs to or from the Delivery Point(s), multiplied by the difference between the Contract Quantity and the quantity actually taken by Buyer for such Day(s) excluding any quantity for which no sale is available; and (iii) in the event that Buyer has used commercially reasonable efforts to replace the Gas or Seller has used commercially reasonable efforts to sell the Gas to a third party, and no such replacement or sale is available for all or any portion of the Contract Quantity of Gas, then in addition to (i) or (ii) above, as applicable, the sole and exclusive remedy of the performing party with respect to the Gas not replaced or sold shall be an amount equal to any unfavorable difference between the Contract Price and the Spot Price, adjusted for such transportation to the applicable Delivery Point, multiplied by the quantity of such Gas not replaced or sold. Imbalance Charges shall not be recovered under this Section 3.2, but Seller and/or Buyer shall be responsible for imbalance Charges, if any, as provided in Section 4.3. T

Spot Price Standard:

3.2. The sole and exclusive remedy of the parties in the event of a breach of a Firm obligation to deliver or receive Gas shall be recovery of the following: (i) In the event of a breach by Seller on any Day(s), payment by Seller to Buyer in an amount equal to the difference between the Contract Quantity and the actual quantity delivered by Seller and received by Buyer for such Day(s), multiplied by the positive difference, if any, obtained by subtracting the Contract Price from the Spot Price; or (ii) in the event of a breach by Buyer on any Day(s), payment by Buyer to Seller in an amount equal to the difference between the Contract Quantity and the actual quantity delivered by Seller and received by Buyer for such Day(s), multiplied by the positive difference, if any,

obtained by subtracting the applicable Spot Price from the Contract Price. Imbalance Charges shall not be recovered under this Section 3.2, but Seller and/or Buyer shall be responsible for Imbalance Charges, if any, as provided in Section 4.3. The amount of such unfavorable difference shall be payable five Business Days after presentation of the performing party's invoice, which shall set forth the basis upon which such amount was calculated.

- 3.3. Notwithstanding Section 3.2, the parties may agree to Alternative Damages in a Transaction Confirmation executed in writing by both parties.
- 3.4. In addition to Sections 3.2 and 3.3, the parties may provide for a Termination Option in a Transaction Confirmation executed in writing by both parties. The Transaction Confirmation containing the Termination Option will designate the length of nonperformance triggering the Termination Option and the procedures for exercise thereof, how damages for nonperformance will be compensated, and how figuidation costs will be calculated.

SECTION 4. TRANSPORTATION, NOMINATIONS, AND IMBALANCES

- 4.1. Seller shall have the sole responsibility for transporting the Gas to the Delivery Point(s). Buyer shall have the sole responsibility for transporting the Gas from the Delivery Point(s).
- 4.2. The parties shall coordinate their nomination activities, giving sufficient time to meet the deadlines of the affected Transporter(s). Each party shall give the other party timely prior Notice, sufficient to meet the requirements of all Transporter(s) involved in the transaction, of the quantities of Gas to be delivered and purchased each Day. Should either party become aware that actual deliveries at the Delivery Point(s) are greater or lesser than the Scheduled Gas, such party shall promptly notify the other party.
- 4.3. The parties shall use commercially reasonable efforts to avoid imposition of any imbalance Charges. If Buyer or Seller receives an invoice from a Transporter that includes imbalance Charges, the parties shall determine the validity as well as the cause of such imbalance Charges. If the imbalance Charges were incurred as a result of Buyer's receipt of quantities of Gas greater than or less than the Scheduled Gas, then Buyer shall pay for such imbalance Charges or reimburse Seller for such imbalance Charges paid by Seller. If the imbalance Charges were incurred as a result of Seller's delivery of quantities of Gas greater than or less than the Scheduled Gas, then Seller shall pay for such imbalance Charges or reimburse Buyer for such imbalance Charges paid by Buyer.

SECTION 5. QUALITY AND MEASUREMENT

All Gas delivered by Seller shall meet the pressure, quality and heat content requirements of the Receiving Transporter. The unit of quantitive measurement for purposes of this Contract shall be one MMBtu dry. Measurement of Gas quantities hereunder shall be in accordance with the established procedures of the Receiving Transporter.

SECTION 6. TAXES

The parties have selected either "Buyer Pays At and After Delivery Point" or "Seller Pays Before and At Delivery Point" as indicated on the Base Contract.

Buyer Pays At and After Delivery Point:

Seller shall pay or cause to be paid all taxes, fees, levies, penalties, licenses or charges imposed by any government authority ("Texes") on or with respect to the Gas prior to the Delivery Point(s). Buyer shall pay or cause to be paid all Taxes on or with respect to the Gas at the Delivery Point(s) and all Taxes after the Delivery Point(s). If a party is required to remit or pay Taxes that are the other party's responsibility hereunder, the party responsible for such Taxes shall promptly reimburse the other party for such Taxes. Any party entitled to an exemption from any such Taxes or charges shall furnish the other party any necessary documentation thereof.

Seller Pays Before and At Delivery Point:

Seller shall pay or cause to be paid all taxes, fees, levies, penalties, ilcenses or charges imposed by any government authority ("Taxes") on or with respect to the Gas prior to the Delivery Point(s) and all Taxes at the Delivery Point(s). Buyer shall pay or cause to be paid all Taxes on or with respect to the Gas after the Delivery Point(s). If a party is required to remit or pay Taxes that are the other party's responsibility hereunder, the party responsible for such Taxes shall promptly reimburse the other party for such Taxes. Any party entitled to an exemption from any such Taxes or charges shall furnish the other party any necessary documentation thereof.

SECTION 7. BILLING, PAYMENT, AND AUDIT

- 7.1. Selier shall invoice Buyer for Gas delivered and received in the preceding Month and for any other applicable charges, providing supporting documentation acceptable in industry practice to support the amount charged. If the actual quantity delivered is not known by the billing date, billing will be prepared based on the quantity of Scheduled Gas. The invoiced quantity will then be adjusted to the actual quantity on the following Month's billing or as soon thereafter as actual delivery information is available.
- 7.2. Buyer shall remit the amount due under Section 7.1 in the manner specified in the Base Contract, in immediately available funds, on or before the later of the Payment Date or 10 Days after receipt of the invoice by Buyer; provided that if the Payment Date is not a Business Day, payment is due on the next Business Day following that date. In the event any payments are due Buyer hereunder, payment to Buyer shall be made in accordance with this Section 7.2.
- 7.3. In the event payments become due pursuant to Sections 3.2 or 3.3, the performing party may submit an invoice to the nonperforming party for an accelerated payment setting forth the basis upon which the invoiced amount was calculated. Payment from the nonperforming party will be due five Business Days after receipt of invoice.
- 7.4. If the invoiced party, in good faith, disputes the amount of any such invoice or any part thereof, such invoiced party will pay such amount as it concedes to be correct; provided, however, if the invoiced party disputes the amount due, it must provide supporting

documentation acceptable in industry practice to support the amount paid or disputed without undue delay. In the event the parties are unable to resolve such dispute, either party may pursue any remedy available at law or in equity to enforce its rights pursuant to this Section.

- 7.5. If the involced party falls to remit the full amount payable when due, interest on the unpaid portion shall accrue from the date due until the date of payment at a rate equal to the lower of (i) the then-effective prime rate of interest published under "Money Rates" by The Wall Street Journal, plus two percent per annum; or (ii) the maximum applicable lawful interest rate.
- 7.6. A party shall have the right, at its own expense, upon reasonable Notice and at reasonable times, to examine and audit and to obtain copies of the relevant portion of the books, records, and telephone recordings of the other party only to the extent reasonably necessary to verify the accuracy of any statement, charge, payment, or computation made under the Contract. This right to examine, audit, and to obtain copies shall not be available with respect to proprietary information not directly relevant to transactions under this Contract. All invoices and billings shall be conclusively presumed final and accurate and all associated claims for under- or overpayments shall be deemed waived unless such invoices or billings are objected to in writing, with adequate explanation and/or documentation, within two years after the Month of Gas delivery. All retroactive adjustments under Section 7 shall be paid in full by the party owing payment within 30 Days of Notice and substantiation of such inaccuracy.
- 7.7. Unless the parties have elected on the Base Contract not to make this Section 7.7 applicable to this Contract, the parties shall not all undisputed amounts due and owing, and/or past due, arising under the Contract such that the party owing the greater amount shall make a single payment of the not amount to the other party in accordance with Section 7; provided that no payment required to be made pursuant to the terms of any Credit Support Obligation or pursuant to Section 7.3 shall be subject to netting under this Section. If the parties have executed a separate netting agreement, the terms and conditions therein shall prevail to the extent inconsistent herewith.

SECTION 8. TITLE, WARRANTY, AND INDEMNITY

- 8.1. Unless otherwise specifically agreed, title to the Gas shall pass from Seller to Buyer at the Delivery Point(s). Seller shall have responsibility for and assume any liability with respect to the Gas prior to its delivery to Buyer at the specified Delivery Point(s). Buyer shall have responsibility for and assume any liability with respect to said Gas after its delivery to Buyer at the Delivery Point(s).
- 8.2. Seller warrants that it will have the right to convey and will transfer good and merchantable title to all Gas sold hereunder and delivered by it to Buyer, free and clear of all liens, encumbrances, and claims. EXCEPT AS PROVIDED IN THIS SECTION 8.2 AND IN SECTION 15.8, ALL OTHER WARRANTIES, EXPRESS OR IMPLIED, INCLUDING ANY WARRANTY OF MERCHANTABILITY OR OF FITNESS FOR ANY PARTICULAR PURPOSE, ARE DISCLAIMED.
- 8.3. Seller agrees to Indemnify Buyer and save it harmless from all losses, liabilities or claims including reasonable attorneys' fees and costs of court ("Claims"), from any and all persons, arising from or out of claims of title, personal injury (including death) or property damage from sald Gas or other charges thereon which attach before title passes to Buyer. Buyer agrees to indemnify Seller and save it harmless from all Claims, from any and all persons, arising from or out of claims regarding payment, personal injury (including death) or property damage from sald Gas or other charges thereon which attach after title passes to Buyer.
- 8.4. The parties agree that the delivery of and the transfer of title to all Gas under this Contract shall take place within the Customs Territory of the United States (as defined in general note 2 of the Harmonized Tariff Schedule of the United States 19 U.S.C. §1202, General Notes, page 3); provided, however, that in the event Seller took title to the Gas outside the Customs Territory of the United States, Seller represents and warrants that it is the importer of record for all Gas entered and delivered into the United States, and shall be responsible for entry and entry summary fillings as well as the payment of duties, taxes and fees, if any, and all applicable record keeping requirements.
- 8.5. Notwithstanding the other provisions of this Section 8, as between Seller and Buyer, Seller will be liable for all Claims to the extent that such arise from the fallure of Gas delivered by Seller to meet the quality requirements of Section 5.

SECTION 9. NOTICES

- 9.1. All Transaction Confirmations, invoices, payment instructions, and other communications made pursuant to the Base Contract ("Notices") shall be made to the addresses specified in writing by the respective parties from time to time.
- 9.2. All Notices required hereunder shall be in writing and may be sent by facsimile or mutually acceptable electronic means, a nationally recognized overnight courier service, first class mall or hand delivered.
- 9.3. Notice shall be given when received on a Business Day by the addressee. In the absence of proof of the actual receipt date, the following presumptions will apply. Notices sent by facsimile shall be deemed to have been received upon the sending party's receipt of its facsimile machine's confirmation of successful transmission. If the day on which such facsimile is received is not a Business Day or is after five p.m. on a Business Day, then such facsimile shall be deemed to have been received on the next following Business Day. Notice by overnight mail or courier shall be deemed to have been received on the next Business Day after it was sent or such earlier time as is confirmed by the receiving party. Notice via first class mail shall be considered delivered five Business Days after mailing.
- 9.4. The party receiving a commercially acceptable Notice of change in payment instructions or other payment information shall not be obligated to implement such change until ten Business Days after receipt of such Notice.

SECTION 10. FINANCIAL RESPONSIBILITY

10.1. If either party ("X") has reasonable grounds for insecurity regarding the performance of any obligation under this Contract (whether or not then due) by the other party ("Y") (including, without limitation, the occurrence of a material change in the

creditworthiness of Y or its Guarantor, if applicable), X may demand Adequate Assurance of Performance. "Adequate Assurance of Performance" shall mean sufficient security in the form, amount, for a term, and from an issuer, all as reasonably acceptable to X, including, but not limited to cash, a standby irrevocable latter of credit, a prepayment, a security interest in an asset or guaranty. Y hereby grants to X a continuing first priority security interest in, lien on, and right of setoff against all Adequate Assurance of Performance in the form of cash transferred by Y to X pursuant to this Section 10.1. Upon the return by X to Y of such Adequate Assurance of Performance, the security interest and lien granted hereunder on that Adequate Assurance of Performance shall be released automatically and, to the extent possible, without any further action by either party.

- 10.2. In the event (each an "Event of Default") either party (the "Defaulting Party") or its Guarantor shall: (i) make an assignment or any general arrangement for the benefit of creditors; (ii) file a petition or otherwise commence, authorize, or acquiesce in the commencement of a proceeding or case under any bankruptcy or similar law for the protection of creditors or have such petition filed or proceeding commenced against it; (iii) otherwise become bankrupt or insolvent (however evidenced); (iv) be unable to pay its debts as they fall due; (v) have a receiver, provisional liquidator, conservator, custodian, trustee or other similar official appointed with respect to it or substantially all of its assets; (vi) fall to perform any obligation to the other party with respect to any Credit Support Obligations relating to the Contract; (vii) fall to give Adequate Assurance of Performance under Section 10.1 within 48 hours but at least one Business Day of a written request by the other party; (viii) not have paid any amount due the other party hereunder on or before the second Business Day following written Notice that such payment is due; or ix) be the affected party with respect to any Additional Event of Default; then the other party (the "Non-Defaulting Party") shall have the right, at its sole election, to immediately withhold and/or suspend deliveries or payments upon Notice and/or to terminate and liquidate the transactions under the Contract, in the manner provided in Section 10.3, in addition to any and all other remedies available hereunder.
- 10.3. If an Event of Default has occurred and is continuing, the Non-Defaulting Party shall have the right, by Notice to the Defaulting Party, to designate a Day, no earlier than the Day such Notice is given and no later than 20 Days after such Notice is given, as an early termination date (the "Early Termination Date") for the liquidation and termination pursuant to Section 10.3.1 of all transactions under the Contract, each a "Terminated Transaction". On the Early Termination Date, all transactions will terminate, other than those transactions, if any, that may not be liquidated and terminated under applicable law ("Excluded Transactions"), which Excluded Transactions must be liquidated and terminated as soon thereafter as is legally permissible, and upon termination shall be a Terminated Transaction and be valued consistent with Section 10.3.1 below. With respect to each Excluded Transaction, its actual termination date shall be the Early Termination Date for purposes of Section 10.3.1.

The parties have selected either "Early Termination Damages Apply" or "Early Termination Damages Do Not Apply" as indicated on the Base Contract.

Early Termination Damages Apply:

10.3.1. As of the Early Termination Date, the Non-Defaulting Party shall determine, in good faith and in a commercially reasonable manner, (i) the amount owed (whether or not then due) by each party with respect to all Gas delivered and received between the parties under Terminated Transactions and Excluded Transactions on and before the Early Termination Date and all other applicable charges relating to such deliverles and receipts (including without limitation any amounts owed under Section 3.2), for which payment has not yet been made by the party that owes such payment under this Contract and (ii) the Market Value, as defined below, of each Terminated Transaction. The Non-Defaulting Party shall (x) liquidate and accelerate each Terminated Transaction at its Market Value, so that each amount equal to the difference between such Market Value and the Contract Value, as defined below, of such Terminated Transaction(s) shall be due to the Buyer under the Terminated Transaction(s) if such Market Value exceeds the Contract Value and to the Seller if the opposite is the case; and (y) where appropriate, discount each amount then due under clause (x) above to present value in a commercially reasonable manner as of the Early Termination Date (to take account of the period between the date of liquidation and the date on which such amount would have otherwise been due pursuant to the relevant Terminated Transactions).

For purposes of this Section 10.3.1, "Contract Value" means the amount of Gas remaining to be delivered or purchased under a transaction multiplied by the Contract Price, and "Market Value" means the amount of Gas remaining to be delivered or purchased under a transaction multiplied by the market price for a similar transaction at the Delivery Point determined by the Non-Defaulting Party in a commercially reasonable manner. To ascertain the Market Value, the Non-Defaulting Party may consider, among other valuations, any or all of the settlement prices of NYMEX Gas futures contracts, quotations from leading dealers in energy swap contracts or physical gas trading markets, similar sales or purchases and any other bona fide third-party offers, all adjusted for the length of the term and differences in transportation costs. A party shall not be required to enter into a replacement transaction(s) in order to determine the Market Value. Any extension(s) of the term of a transaction to which parties are not bound as of the Early Termination Date (including but not limited to "evergreen provisions") shall not be considered in determining Contract Values and Market Values. For the avoidance of doubt, any option pursuant to which one party has the right to extend the term of a transaction shall be considered in determining Contract Values and Market Values. The rate of interest used in calculating net present value shall be determined by the Non-Defaulting Party in a commercially reasonable manner.

Early Termination Damages Do Not Apply:

10.3.1. As of the Early Termination Date, the Non-Defaulting Party shall determine, in good faith and in a commercially reasonable manner, the amount owed (whether or not then due) by each party with respect to all Gas delivered and received between the parties under Terminated Transactions and Excluded Transactions on and before the Early Termination Date and all other applicable charges relating to such deliveries and receipts (including without limitation any amounts owed under Section 3.2), for which payment has not yet been made by the party that owes such payment under this Contract.

The parties have selected either "Other Agreement Setoffs Apply" or "Other Agreement Setoffs Do Not Apply" as indicated on the Base Contract.

Other Agreement Setoffs Apply:

Bilateral Setoff Option:

10.3.2. The Non-Defaulting Party shall net or aggregate, as appropriate, any and all amounts owing between the parties under Section 10.3.1, so that all such amounts are netted or aggregated to a single liquidated amount payable by one party to the other (the "Net Settlement Amount"). At its sole option and without prior Notice to the Defaulting Party, the Non-Defaulting Party is hereby authorized to setoff any Net Settlement Amount against (i) any margin or other collateral held by a party in connection with any Credit Support Obligation relating to the Contract; and (ii) any amount(s) (including any excess cash margin or excess cash collateral) owed or held by the party that is entitled to the Net Settlement Amount under any other agreement or arrangement between the parties.

Triangular Setoff Option:

10.3.2. The Non-Defaulting Party shall net or aggregate, as appropriate, any and all amounts owing between the parties under Section 10.3.1, so that all such amounts are netted or aggregated to a single liquidated amount payable by one party to the other (the "Net Settlement Amount"). At its sole option, and without prior Notice to the Defaulting Party, the Non-Defaulting Party is hereby authorized to setoff (i) any Net Settlement Amount against any margin or other collateral held by a party in connection with any Credit Support Obligation relating to the Contract; (ii) any Net Settlement Amount against any amount(s) (including any excess cash margin or excess cash collateral) owed by or to a party under any other agreement or arrangement between the parties; (iii) any Net Settlement Amount owed to the Non-Defaulting Party against any amount(s) (including any excess cash margin or excess cash collateral) owed by the Non-Defaulting Party or its Affiliates to the Defaulting Party under any other agreement or arrangement; (Iv) any Net Settlement Amount owed to the Defaulting Party or its Affiliates under any other agreement or arrangement; and/or (v) any Net Settlement Amount owed to the Defaulting Party against any amount(s) (including any excess cash margin or excess cash collateral) owed by the Defaulting Party or its Affiliates to the Non-Defaulting Party under any other agreement or arrangement; and/or (v) any Net Settlement Amount owed to the Defaulting Party against any amount(s) (including any excess cash margin or excess cash collateral) owed by the Defaulting Party or its Affiliates to the Non-Defaulting Party under any other agreement or arrangement.

Other Agreement Setoffs Do Not Apply:

- 10.3.2. The Non-Defaulting Party shall net or aggregate, as appropriate, any and all amounts owing between the parties under Section 10.3.1, so that all such amounts are neited or aggregated to a single liquidated amount payable by one party to the other (the "Net Settlement Amount"). At its sole option and without prior Notice to the Defaulting Party, the Non-Defaulting Party may setoff any Net Settlement Amount against any margin or other collateral held by a party in connection with any Credit Support Obligation relating to the Contract.
- 10.3.3. If any obligation that is to be included in any netting, aggregation or setoff pursuant to Section 10.3.2 is unascertained, the Non-Defaulting Party may in good falth estimate that obligation and net, aggregate or setoff, as applicable, in respect of the estimate, subject to the Non-Defaulting Party accounting to the Defaulting Party when the obligation is ascertained. Any amount not then due which is included in any netting, aggregation or setoff pursuant to Section 10.3.2 shall be discounted to net present value in a commercially reasonable manner determined by the Non-Defaulting Party.
- 10.4. As soon as practicable after a liquidation, Notice shall be given by the Non-Defaulting Party to the Defaulting Party of the Net Settlement Amount, and whether the Net Settlement Amount is due to or due from the Non-Defaulting Party. The Notice shall include a written statement explaining in reasonable detail the calculation of the Net Settlement Amount, provided that failure to give such Notice shall not affect the validity or enforceability of the liquidation or give rise to any claim by the Defaulting Party against the Non-Defaulting Party. The Net Settlement Amount as well as any setoffs applied against such amount pursuant to Section 10.3.2, shall be paid by the close of business on the second Business Day following such Notice, which date shall not be earlier than the Early Termination Date. Interest on any unpaid portion of the Net Settlement Amount as adjusted by setoffs, shall accrue from the date due until the date of payment at a rate equal to the lower of (i) the then-effective prime rate of interest published under "Money Rates" by The Wall Street Journal, plus two percent per annum; or (ii) the maximum applicable lawful interest rate.
- 10.5. The parties agree that the transactions hereunder constitute a "forward contract" within the meaning of the United States Bankruptcy Code and that Buyer and Seller are each "forward contract merchants" within the meaning of the United States Bankruptcy Code.
- 10.6. The Non-Defaulting Party's remedies under this Section 10 are the sole and exclusive remedies of the Non-Defaulting Party with respect to the occurrence of any Early Termination Date. Each party reserves to itself all other rights, setoffs, counterclaims and other defenses that it is or may be entitled to arising from the Contract.
- 10.7. With respect to this Section 10, if the parties have executed a separate netting agreement with close-out netting provisions, the terms and conditions therein shall prevail to the extent inconsistent herewith.

SECTION 11. FORCE MAJEURE

- 11.1. Except with regard to a party's obligation to make payment(s) due under Section 7, Section 10.4, and Imbalance Charges under Section 4, neither party shall be liable to the other for fallure to perform a Firm obligation, to the extent such fallure was caused by Force Majeure. The term "Force Majeure" as employed herein means any cause not reasonably within the control of the party claiming suspension, as further defined in Section 11.2.
- 11.2. Force Majeure shall include, but not be limited to, the following: (i) physical events such as acts of God, landslides, lightning, earthquakes, fires, storms or storm warnings, such as hurricanes, which result in evacuation of the affected area, floods, washouts, explosions, breakage or accident or necessity of repairs to machinery or equipment or lines of pipe; (ii) weather related events affecting an entire geographic region, such as low temperatures which cause freezing or failure of wells or lines of pipe;

(iii) interruption and/or curtaliment of Firm transportation and/or storage by Transporters; (iv) acts of others such as strikes, lockouts or other industrial disturbances, riots, sabotage, insurrections or wars, or acts of terror; and (v) governmental actions such as necessity for compliance with any court order, law, statute, ordinance, regulation, or policy having the effect of law promulgated by a governmental authority having jurisdiction. Seller and Buyer shall make reasonable efforts to avoid the adverse impacts of a Force Majeure and to resolve the event or occurrence once it has occurred in order to resume performance.

- 11.3. Neither party shall be entitled to the benefit of the provisions of Force Majeure to the extent performance is affected by any or all of the following circumstances: (i) the curtailment of interruptible or secondary Firm transportation unless primary, in-path, Firm transportation is also curtailed; (ii) the party claiming excuse failed to remedy the condition and to resume the performance of such covenants or obligations with reasonable dispatch; or (iii) economic hardship, to include, without limitation, Seller's ability to sell Gas at a higher or more advantageous price than the Contract Price, Buyer's ability to purchase Gas at a lower or more advantageous price than the Contract Price, or a regulatory agency disallowing, in whole or in part, the pass through of costs resulting from this Contract; (iv) the loss of Buyer's market(s) or Buyer's inability to use or resell Gas purchased hereunder, except, in either case, as provided in Section 11.2; or (v) the loss or failure of Seller's gas supply or depletion of reserves, except, in either case, as provided in Section 11.2. The party claiming Force Majeure shall not be excused from its responsibility for imbalance Charges.
- 11.4. Notwithstanding anything to the contrary herein, the parties agree that the settlement of strikes, lockouts or other industrial disturbances shall be within the sole discretion of the party experiencing such disturbance.
- 11.5. The party whose performance is prevented by Force Majeure must provide Notice to the other party. Initial Notice may be given orally; however, written Notice with reasonably full particulars of the event or occurrence is required as soon as reasonably possible. Upon providing written Notice of Force Majeure to the other party, the affected party will be relieved of its obligation, from the onset of the Force Majeure event, to make or accept delivery of Gas, as applicable, to the extent and for the duration of Force Majeure, and neither party shall be deemed to have falled in such obligations to the other during such occurrence or event.
- 11.6. Notwithstanding Sections 11.2 and 11.3, the parties may agree to alternative Force Majeure provisions in a Transaction Confirmation executed in writing by both parties.

SECTION 12. TERM

This Contract may be terminated on 30 Day's written Notice, but shall remain in effect until the expiration of the latest Delivery Period of any transaction(s). The rights of either party pursuant to Section 7.6, Section 10, Section 13, the obligations to make payment hereunder, and the obligation of either party to indemnify the other, pursuant hereto shall survive the termination of the Base Contract or any transaction.

SECTION 13. LIMITATIONS

FOR BREACH OF ANY PROVISION FOR WHICH AN EXPRESS REMEDY OR MEASURE OF DAMAGES IS PROVIDED, SUCH EXPRESS REMEDY OR MEASURE OF DAMAGES SHALL BE THE SOLE AND EXCLUSIVE REMEDY. A PARTY'S LIABILITY HEREUNDER SHALL BE LIMITED AS SET FORTH IN SUCH PROVISION, AND ALL OTHER REMEDIES OR DAMAGES AT LAW OR IN EQUITY ARE WAIVED. IF NO REMEDY OR MEASURE OF DAMAGES IS EXPRESSLY PROVIDED HEREIN OR IN A TRANSACTION, A PARTY'S LIABILITY SHALL BE LIMITED TO DIRECT ACTUAL DAMAGES ONLY. SUCH DIRECT ACTUAL DAMAGES SHALL BE THE SOLE AND EXCLUSIVE REMEDY, AND ALL OTHER REMEDIES OR DAMAGES AT LAW OR IN EQUITY ARE WAIVED. UNLESS EXPRESSLY HEREIN PROVIDED, NEITHER PARTY SHALL BE LIABLE FOR CONSEQUENTIAL, INCIDENTAL, PUNITIVE, EXEMPLARY OR INDIRECT DAMAGES, LOST PROFITS OR OTHER BUSINESS INTERRUPTION DAMAGES, BY STATUTE, IN TORT OR CONTRACT, UNDER ANY INDEMNITY PROVISION OR OTHERWISE. IT IS THE INTENT OF THE PARTIES THAT THE LIMITATIONS HEREIN IMPOSED ON REMEDIES AND THE MEASURE OF DAMAGES BE WITHOUT REGARD TO THE CAUSE OR CAUSES RELATED THERETO, INCLUDING THE NEGLIGENCE OF ANY PARTY, WHETHER SUCH NEGLIGENCE BE SOLE, JOINT OR CONCURRENT, OR ACTIVE OR PASSIVE. TO THE EXTENT ANY DAMAGES REQUIRED TO BE PAID HEREUNDER ARE LIQUIDATED, THE PARTIES ACKNOWLEDGE THAT THE DAMAGES ARE DIFFICULT OR IMPOSSIBLE TO DETERMINE, OR OTHERWISE OBTAINING AN ADEQUATE REMEDY IS INCONVENIENT AND THE DAMAGES CALCULATED HEREUNDER CONSTITUTE A REASONABLE APPROXIMATION OF THE HARM OR LOSS.

SECTION 14. MARKET DISRUPTION

If a Market Disruption Event has occurred then the parties shall negotiate in good faith to agree on a replacement price for the Floating Price) for the affected Day, and if the parties have not so agreed on or before the second Business Day following the affected Day then the replacement price for the Floating Price shall be determined within the next two following Business Days with each party obtaining, in good faith and from non-affiliated market participants in the relevant market, two quotes for prices of Gas for the affected Day of a similar quality and quantity in the geographical location closest in proximity to the Delivery Point and averaging the four quotes. If either party fails to provide two quotes then the average of the other party's two quotes shall determine the replacement price for the Floating Price. "Floating Price" means the price or a factor of the price agreed to in the transaction as being based upon a specified index. "Market Disruption Event" means, with respect to an index specified for a transaction, any of the following events: (a) the failure of the index to announce or publish information necessary for determining the Floating Price; (b) the failure of trading to commence or permanent discontinuation or material suspension of trading on the exchange or market acting as the index; (c) the temporary or permanent discontinuance or unavailability of the index; (d) the temporary or permanent closing of any exchange acting as the index; or (e) both parties agree that a material change in the formula for or the method of determining the Floating Price has occurred. For the purposes of the calculation of a replacement price for the Floating Price, all numbers shall be increased by one and if the fourth decimal number is less than five, then the third decimal number shall remain unchanged.

SECTION 15. MISCELLANEOUS

- 15.1. This Contract shall be binding upon and inure to the benefit of the successors, assigns, personal representatives, and heirs of the respective parties hereto, and the covenants, conditions, rights and obligations of this Contract shall run for the full term of this Contract. No assignment of this Contract, in whole or in part, will be made without the prior written consent of the non-assigning party (and shall not relieve the assigning party from liability hereunder), which consent will not be unreasonably withheld or delayed; provided, either party may (i) transfer, sell, pledge, encumber, or assign this Contract or the accounts, revenues, or proceeds hereof in connection with any financing or other financial arrangements, or (ii) transfer its interest to any parent or Affiliate by assignment, merger or otherwise without the prior approval of the other party. Upon any such assignment, transfer and assumption, the transferor shall remain principally liable for and shall not be relieved of or discharged from any obligations hereunder.
- 15.2. If any provision in this Contract is determined to be invalid, void or unenforceable by any court having jurisdiction, such determination shall not invalidate, void, or make unenforceable any other provision, agreement or covenant of this Contract.
- 15.3. No walver of any breach of this Contract shall be held to be a walver of any other or subsequent breach.
- 15.4. This Contract sets forth all understandings between the parties respecting each transaction subject hereto, and any prior contracts, understandings and representations, whether oral or written, relating to such transactions are merged into and superseded by this Contract and any effective transaction(s). This Contract may be amended only by a writing executed by both parties.
- 15.5. The Interpretation and performance of this Contract shall be governed by the laws of the jurisdiction as indicated on the Base Contract, excluding, however, any conflict of laws rule which would apply the law of another jurisdiction.
- 15.6. This Contract and all provisions herein will be subject to all applicable and valid statutes, rules, orders and regulations of any governmental authority having jurisdiction over the parties, their facilities, or Gas supply, this Contract or transaction or any provisions thereof.
- 15.7. There is no third party beneficiary to this Contract.
- 15.8. Each party to this Contract represents and warrants that it has full and complete authority to enter into and perform this Contract. Each person who executes this Contract on behalf of either party represents and warrants that it has full and complete authority to do so and that such party will be bound thereby.
- 15.9. The headings and subheadings contained in this Contract are used solely for convenience and do not constitute a part of this Contract between the parties and shall not be used to construe or interpret the provisions of this Contract.
- 15.10. Unless the parties have elected on the Base Contract not to make this Section 15.10 applicable to this Contract, neither party shall disclose directly or indirectly without the prior written consent of the other party the terms of any transaction to a third party (other than the employees, lenders, royalty owners, counsel, accountants and other agents of the party, or prospective purchasers of all or substantially all of a party's assets or of any rights under this Contract, provided such persons shall have agreed to keep such terms confidential) except (i) in order to comply with any applicable law, order, regulation, or exchange rule, (ii) to the extent necessary for the enforcement of this Contract, (iii) to the extent necessary to implement any transaction, (iv) to the extent necessary to comply with a regulatory agency's reporting requirements including but not limited to gas cost recovery proceedings; or (v) to the extent such information is delivered to such third party for the sole purpose of calculating a published index. Each party shall notify the other party of any proceeding of which it is aware which may result in disclosure of the terms of any transaction (other than as permitted hereunder) and use reasonable efforts to prevent or limit the disclosure. The existence of this Contract is not subject to this confidentiality obligation. Subject to Section 13, the parties shall be entitled to all remedies available at law or in equity to enforce, or seek relief in connection with this confidentiality obligation. The terms of any transaction hereunder shall be kept confidential by the parties hereto for one year from the expiration of the transaction.

In the event that disclosure is required by a governmental body or applicable law, the party subject to such requirement may disclose the material terms of this Contract to the extent so required, but shall promptly notify the other party, prior to disclosure, and shall cooperate (consistent with the disclosing party's legal obligations) with the other party's efforts to obtain protective orders or similar restraints with respect to such disclosure at the expense of the other party.

- 15.11. The parties may agree to dispute resolution procedures in Special Provisions attached to the Base Contract or in a Transaction Confirmation executed in writing by both parties
- 15.12. Any original executed Base Contract, Transaction Confirmation or other related document may be digitally copled, photocopied, or stored on computer tapes and disks (the "Imaged Agreement"). The Imaged Agreement, if introduced as evidence on paper, the Transaction Confirmation, if introduced as evidence in automated facsimile form, the recording, if introduced as evidence in its original form, and all computer records of the foregoing, if introduced as evidence in printed format, in any judicial, arbitration, mediation or administrative proceedings will be admissible as between the parties to the same extent and under the same conditions as other business records originated and maintained in documentary form. Neither Party shall object to the admissibility of the recording, the Transaction Confirmation, or the Imaged Agreement on the basis that such were not originated or maintained in documentary form. However, nothing herein shall be construed as a waiver of any other objection to the admissibility of such evidence.

DISCLAIMER: The purposes of this Contract are to facilitate trade, avoid misunderstandings and make more definite the terms of contracts of purchase and sale of natural gas. Further, NAESB does not mandate the use of this Contract by any party. NAESB DISCLAIMS AND EXCLUDES, AND ANY USER OF THIS CONTRACT ACKNOWLEDGES AND AGREES TO NAESB'S DISCLAIMER OF, ANY AND ALL WARRANTIES, CONDITIONS OR REPRESENTATIONS, EXPRESS OR IMPLIED, ORAL OR WRITTEN, WITH RESPECT TO THIS CONTRACT OR ANY PART THEREOF, INCLUDING ANY AND ALL IMPLIED WARRANTIES OR CONDITIONS OF TITLE, NON-INFRINGEMENT, MERCHANTABILITY, OR FITNESS OR SUITABILITY FOR ANY PARTICULAR PURPOSE (WHETHER OR NOT NAESB KNOWS, HAS REASON OF CUSTOM OR USAGE IN THE TRADE, OR BY COURSE OF DEALING, EACH USER OF THIS CONTRACT ALSO AGREES THAT UNDER NO CIRCUMSTANCES WILL NAESB BE LIABLE FOR ANY DIRECT, SPECIAL, INCIDENTAL, EXEMPLARY, PUNITIVE OR CONSEQUENTIAL DAMAGES ARISING OUT OF ANY USE OF THIS CONTRACT.

EXHIBIT A - FORM OF TRANSACTION CONFIRMATION*

*The actual Transaction Confirmation en and other local requirements and terms a only and CESNG reserves the right to alt	opplicable to the Facility. This	Form of Transact	er from this form to reflect state regulatory ion Confirmation is for illustrative purposes ne.
· On /the Ulfact	ter Agreement"), by and be	tween Constellation in the contract of the con	natural gas supply agreement effective on Energy Services - Natural Gas, LLC ne terms and conditions of such Master ed to them in the Master Agreement.
Trade Date:	•		
Facility Name:			
Delivery Period:	<u>.</u>		·
Deal Type:			
Nature of Parties' Obligation:			
Contract Quantity/Price:	•		
Month/Yr	Contract Quantity (i	n MMBtu)	Contract Price
Utility: This Transaction Confirmation docum it is binding and shall be deemed acc in writing within two (2) business days CONSTELLATION ENERGY SERVICES	epted as the definitive exp of CESNG's execution dat	e, e,	uthorized representatives of the Parties. ansaction unless disputed by Customer
	•	: SAMPLE	
By:	N:	ame: NOT FOR I	
Title	n.	ate:	
	EXHIBIT A	·	
	CONFIDENTIAL		

TRANSACTION CONFIRMATION

Page 1 of 2

This Transaction Confirmation dated 4/4/2019 is made a part of, and pursuant to terms of, the Base Contract for Sale and Purchase of Gas dated May 4, 2016 between Buyer and Seller identified herein (the "NAESB').

SELLER:

Constellation NewEnergy - Gas Division, LLC (as successor by merger to Constellation Energy Services Natural Gas, LLC)

1716 Lawrence Drive PO Box6025 De Pere, WI 54115-6025 Attn: Account Management E-mail: amwi@constellation.com

Phone: 888-367-4403 (Option 2)

Fax: 443-213-6116

BUYER:

Milwaukee Board of School Directors on behalf of Milwaukee Public Schools

1124 N. 11th St. Milwaukee WI 53233

Attn: Yuriy Indman, PE, CEM

E-mail: indmany@milwaukee.k12.wi.us

Phone: 414-416-5112

Agent /Assoc: Julie Cowling

Account Executive:

Delivery Period: July 1, 2019 to June 30, 2022; provided however, the Delivery Period shall begin on the later of, (i) the day after the first meter read during the period noted above, or (ii) upon acceptance by the utility of Seller as Buyer's gas supplier.

Contract Quantity (in dekatherms): see table

Month	Quantity	Month	Quantity	Month	Quantity	Month	Quantity
Jul	5,600	Oct	46,000	Jan	165,000	Apr	66,000
Aug	5,100	Nov	95,800	Feb	149,000	May	28,000
Sep	8,500	Dec	162,000	Mar	117,000	Jun	9,000

Price: NYMEX Last Day Settle ("NYM LDS") plus Market Basis per dekatherm. Either or both of the price components may be fixed for specific quantities and delivery months with mutual agreement of the Parties. The Basis is fixed at \$0.89 per dekatherm.

Performance Obligation: Up to 7,300 dkthm/day shall be Firm. The remainder shall be Interruptible

Delivery Point: Buyer's Transporter Citygate Buyer's Transporter: WE Energies

- SPECIAL PROVISIONS -

Balancing Services

1. A Balancing Service charge of \$0.03 per dekatherm applies in addition to the Price.

Buyer and Seller shall use commercially reasonable efforts to provide each other with sufficient notice to allow timely and accurate nominations in accordance with the nomination requirements of that Transporter, as they may change from time to time due to Constraint Day requirements or otherwise. Buyer accepts the following responsibilities: Buyer hereby appoints Seller as Its agent for the purposes of balancing, Including the receipt of current and historic Usage, storage, billing and transportation data from Transporter. Buyer agrees to assist Seller in the timely collection of data directly and through Transporter, notify Seller of usage expectations and any material changes to those usage expectations, and comply with any Daily Limitations concerning use. If Usage data is unavailable electronically, Buyer shall obtain usage data manually and deliver that information weekly via facsimile or electronic mail, or as requested by Seller.

- Seller agrees to monitor and review all information provided to the Seller by Buyer and Transporter regarding Buyer's gas Use. Based on such information, Seller shall use commercially reasonable effort to keep Buyer within the balancing parameters established by Transporter. Seller shall pass through Io Buyer and Buyer shall pay for any Imbalance Charges and/or Cashout resulting from Buyer's failure to (I) fulfill its obligations set forth in paragraph 1 above, (ii) maintain telemetry, and/or (iii) meet Daily Limitations with respect to Use.
- Mid-month variations to delivered quantity. In the event that the quantity of natural gas that can be delivered to Buyer (without the Transporter assessing Imbalance Charges or Cashout) is less than the quantity agreed to be purchased for the relevant period, Buyer shall remain liable for the entire quantity It agreed to purchase, however, Seller will provide a credit at prevailing market rates for that portion of the purchased quantity that Is not delivered. In the event that Buyer's expected Use will exceed the quantity of gas purchased by Buyer and/or available to Buyer without incurring Imbalance Charges for the relevant period, Seller will use commercially reasonable efforts to provide Incremental supply, plus Fuel (if applicable) at prevailing market rates. Credits for reductions in delivered quantity and/or charges for Incremental supply shall appear as separate line items on Seller's invoice to Buyer, including applicable price and quantity.

NAESB Transaction Confirmation 05-2015

ACCOUNTS:

[X] Check here if Attachment A Is attached and made a part of this Transaction Confirmation.

Appendix B

TRANSACTION CONFIRMATION

Page 2 of 2

4. **Telemetry Obligations.** Buyer acknowledges and agrees that it shall be responsible for installing and maintaining a meter telephone line. Buyer shall also be responsible for any charges imposed by a contractor for repairing the meter telephone line. As a convenience to Buyer, Seller may arrange for telemetry service or repairs to be provided and Buyer agrees it shall be responsible for any costs associated with such service and repairs.

"Constraint Day" means any period of time during which a Transporter issues balancing orders, operational now orders, operational matching orders (or the equivalent), or winter splits or deems the day a Critical Gas Day (or the equivalent) and thereby restricts deliveries by Seller or use by Buyer, requires a specific delivery by Seller or use by Buyer, or otherwise requires an affirmative action by either Party due to adverse operational conditions experienced by the Transporter.

"Daily Limitations' means Transporter enforced restrictions or mandates concerning quantities of gas delivered and/or used, including those imposed due to Constraint Day conditions and/or under ordinary operating conditions.

- 5. Invoices. Invoices shall be prepared bas dupon quantities used.
- 6. Consolidated Billing: Seller shall include the Transporter distribution fees on the monthly invoice. Past due and other charges due Transporter, that apply to service prior to the effective date of the Confirmation will not be Included in Seller's invoice to Buyer. Seller may discontinue this service at any time with notice to Buyer.
- 7. Term: This Transaction Confirmation is effective for the Delivery Period as it may be extended or renewed as described below.
- 8. Extending Deli very Period: Should the Parties agree to a fixed price for any pricing component for a delivery month (or billing cycle, as applicable) that occurs after the current Delivery Period, then the Delivery Period of this Transaction Confirmation shall be extended to the latest occurring delivery month (or billing cycle, as applicable) with a fixed price or fixed price component.
- 9. Renewal: The Parties agree that this Transaction Confirmation shall automatically renew for successive twelve (12) month periods (each constituting a "Renewal Delivery Period") unless written notice pursuant to Section 9 of the Agreement terminating the Transaction Confirmation at the end of a Delivery Period Is given by either Party thirty (30) days prior to the end of the Delivery Period. The Contract Quantity during any Renewal Delivery Period shall be the Contract Quantity stated in the table above for the respective month. The Price for the Contract Quantity during any Renewal Delivery Period shall be NYMEX LDS plus Market Basis.

All other terms and conditions of this Transaction Confirmation shall remain unchanged.

Special Condition: This Confirmation is not effective and binding upon the parties hereto unless signed concurrently with the Professional Services Contract Modification and Extension Agreement dated effective April 12, 2019:

Signature: Signature: By: Daniel Marks

By: Title: Executive Director Retail Gas Operations

Signature: By: Signature: By: By:

Seller: Constellation NewEnergy - Gas Division, LLC

Date: April 12, 2019 Date:

Signature: By:

Buyer: Milwaukee Board of Schools Directors on behalf of

Milwaukee Public Schools

Title: Date:

Attachment A to Master Retail Gas Sales Agreement	<u>Utility Account</u> <u>Number</u>	<u>LDC</u>
Academy of Accelerated Learning Site 316	1422059622	WG
Administration Building Site 501	3023456227	WG
Allen-Field School Site 73	4036305614	WG
Arthur Fletcher School Site 197	4037518053	WG
Auer Avenue School Site 77	3279560425	WG
Bayview High School Site 12	6012173238	WG
Beulah Brinton Site 722	1626350526	WG
Boys & Girls Club Site 823	5042259688	WG
Brown Street School Site 89	0045882513	WG
Browning School Site 123	1883231151	WG
Bruce School Site 93	0017839039	WG
Building Operations Service Site 459	8494684667	WG
Burbank School Site 95	6641478566	WG
Byron Kilbourn Elem School Site 237	7201071166	WG
Carleton School Site 101	7676252124	WG
Cass Street School Site 104	8843713868	WG
Clara Barton Elementary 081	5679816655	WG
Clarke Street School Site 110	3283379352	WG
Clemens School Site 114	9418112981	WG
Clement Avenue School Site 113	6456733856	WG
Congress School Site 116	2491414658	WG
Cooper School Site 117	3655001981	WG
Craig Site 119	3267844363	WG
Curtin School Site 122	6813728760	WG
Doerfler School Site 125	5404475668	WG
Douglas Community Academy Site 055	5696643549	WG
Dover School Site 131	8002114158	WG
Dr Ben Carson Academy Site 309	8024279050	WG
Edison Middle School Site 047	680146981	WG
Eighty-First Street School Site 143	9272192283	WG
Elm Creative Arts School Site 148	2865421970	WG
Engleburg School Site 152	9079476386	WG
Facilities & Maintenance Services Site 451	1473762114	WG
Fairview School Site 155	6409464924	WG
Fernwood Montessori Site 158	3814469445	WG
Fifty-Third Street School Site 170	2603553431	WG
Forest Home School Site 173	4838859755	WG
Franklin School Site 179	3090098279	WG
Fratney School Site 182	842227409	WG
French Immersion School Site 062	4035543923	WG
Gaenslen School Site 185	7030603669	WG
Garden Homes School Site 188	6437054867	WG
George Washington Carver Acad Site 298	8457438263	WG
Gilbert Stuart Elem School Site 344	2050239508	WG
Goodrich School Site 192	9031242448	WG
Grand Ave Middle site 051	2255987370	WG

Grant School Site 193	0419252083	WG
Grantosa School Site 196	2887872471	WG
Green Bay Avenue School Site 199	6479350926	WG
Greenfield School Site 202	5463838436	WG
Groppi High School Site 383	6285652553	WG
Gwen T Jackson Site 377	0844240964	WG
Hamilton High School Site 018	4457386671	WG
Hamlin Garland Elem School Site 399	0244416617	WG
Hampton School Site 205	7033239278	WG
HAMS at Tippecanoe Site 362	3625560600	WG
Hartford Avenue School Site 208	5449989785	WG
Hawley Elementary School Site 211	5814976876	WG
Hayes School Site 214	853826927	WG
Hi-Mount School Site 217	4687719468	WG
Holmes School Site 218	4427624131	WG
Hopkins Lloyd Street School Site 220	0018179292	WG
Humboldt Park Site 223	4444612091	WG
Ideal School Site 191	2693256667	WG
John Audubon Multiplex Site 041	3280349236	WG
John Madison Academic Camput Site 022	6491688983	WG
Kagel Elementary School Site 232	4887713808	WG
Keefe Avenue School Site 235	4070086201	WG
King High School Site 020	6630039544	WG
King Jr Elementary School Site 083	2051031784	WG
Kluge School Site 337	3256764630	WG
Kosciuszko Middle School Site 050	6232048392	WG
Lafollette School Site 238	8418063106	WG
Lancaster School Site 241	0230753590	WG
Lee School Site 244	8229155317	WG
Lincoln Avenue School Site 250	0860949971	WG
Lincoln Middle of the Arts site 006	7427614174	WG
Longfellow School Site 256	8470682787	WG
Louisa May Alcott Elem Site 075	2200891356	WG
Lowell Elementary Site 257	4805296202	WG
MacDowell Montessori at Solomon Juneua Site	1694375731	WG
Malcolm X Academy Site 048	6463377073	WG
Manitoba School Site 265	6815818990	WG
Maple Tree Elementary School Site 267	1050274006	WG
Maryland Avenue School Site 268	2607704309	WG
Meir Elementary School Site 176	2643629076	WG
Metcalfe School Site 270	3833479443	WG
Milwaukee Campus Site 016	0834863893	WG
Milwaukee Ed Center Site 067	3475025278	WG
Milwaukee German Immersion Site 146	3447010427	WG
Milwaukee High School of the Arts Site 038	6608051249	WG
Milwaukee School of Languages Site 071	4611469735	WG
Milwaukee Sign Language School Site 226	5655016448	WG
J J J		

Milwaukee Spanish Immersion Site 167	8045337066	WG
Mitchell Elementary Site 274	1260641942	WG
Morgandale School Site 277	6631734217	WG
Morse Marshall School Site 008	6088198150	WG
Nathaniel Hawthorne Elem School Site 212	8457585865	WG
Neeskara Site 283	8855841788	WG
Ninety-Fifth Street School Site 289	6499338122	WG
•	6671619413	WG
North Division High School Site 024	6246501615	WG
Northwest Academy Site 053		WG
Parkside School Site 049	0046762243 4051595249	WG
Parkview School Site 301		WG
Pierce School Site 307	6013623399	
Pulaski High School Meter 1196742 Site 26	0081537199	WG
Pulaski High School Meter 1196757 Site 26	7267683275	WG
Ralph Waldo Emerson Elem School Site 150	9237920249	WG
Riley Site 313	1298709315	WG
River Trail School Site 194	3857163924	WG
Riverside University High School Site 029	1820504560	WG
Roger Street Academy Site 318	6474102242	WG
Roosevelt Middle School Site 059	2083687498	WG
Rufus King International Site 380	0019765210	WG
Seventy Sixth Street School Site 092	4831292654	WG
Sherman Site 319	5652553168	WG
Sholes Middle School Site 060	3275244493	WG
Siefert School Site 322	2238213451	WG
Silver Spring School Site 325	8279853457	WG
Sixty-Eighth Street Elementary Site 328	4639290587	WG
Sixty-Fifth Street School Site 331	7469847593	WG
South Division High School Site 032	1865545358	WG
Special Services Center Site 412	2692766324	WG
Starms Discovery Learning Center Site 312	5865709589	WG
Starms Early Education Center Site 149	662854426	WG
Story School Site 343	5870187291	WG
Thirty-Fifth Street School Site 350	7848259439	WG
Thirty-Seventh Street Elem Site 356	8634011937	WG
Thoreau School Site 360	2611573666	WG
Thurston Woods Campus Elem Site 154	0230314468	WG
Tippecanoe Elementary Site 362	3625560600	WG
Townsend School Site 365	6644807292	WG
Trowbridge School Site 368	4089190512	WG
Victory School Site 387	5426784096	WG
Vincent High School Site 033	4004598630	WG
Walker Middle School Site 065	6222535735	WG
Washington High - Mtr 1085084 - Site 035	0275485991	WG
Washington High Meter 1196768 Site 035	0275485991	WG
Webster Middle School Site 066	8602575500	WG
Wedgewood Middle School Site 044	2236182960	WG
Westside 2 Academy Site 370 acct 4249149688	4249149688	WG

Westside 2 Academy Site 370 acct 5889372066	5889372066	WG
Westside One Academy 31st Street Site 353	7080577427	WG
Wheatley School Site 374	0694915814	WG
Whitman School Site 397	5401267708	WG
William Cullen Bryant Elem School Site 094	3480511683	WG
Wisconsin Ave Site 401	6405565322	WG
Wisconsin Conserv of Lifelong Learn Site 061	5297268551	WG
Zablocki Site 295	8416137916	WG

APPENDIX B

RETAIL SPECIAL PROVISIONS TO THE BASE CONTRACT FOR SALE AND PURCHASE OF NATURAL GAS (2006) BETWEEN CONSTELLATION ENERGY SERVICES - NATURAL GAS, LLC ("CESNG" or "Seller") AND MILWAUKEE BOARD OF SCHOOL DIRECTORS ON BEHALF OF MILWAUKEE PUBLIC SCHOOLS ("Buyer") **DATED MAY 4, 2016**

The following special provisions to the Base Contract and General Terms and Conditions shall supplement and form part of the Base Contract and General Terms and Conditions between the parties. In the event of any conflict or inconsistency between such special provisions and the Base Contract or General Terms and Conditions, the special provisions shall govern.

SECTION 1. PURPOSE AND PROCEDURES

Section 1.1 is deleted in its entirety and replaced with the following:

- These General Terms and Conditions are intended to facilitate purchase and sale transactions of Gas (each a "Transaction") for one or more of Buyer's accounts or facilities (each a "Facility") on a Firm, Secondary Firm or interruptible basis. The terms of any Transaction between CESNG and Customer shall be as set forth on the applicable TC, but CESNG will procure gas from its suppliers on terms CESNG deems appropriate. The parties may further agree to implement certain strategies or related services, the specifics of which will be set forth in a rider executed by the parties ("Rider"). The entire agreement between the parties shall be the Contract as defined in Section 2.9.
- Section 1.2 "Oral Transaction Procedure" is amended by adding the phrase "or other electronic means of communication" immediately after the phrase "EDI transmission" in the second sentence.
- The last senience of Section 1.3 is deleted in its entirety and replaced with the following:

"In the event of a conflict among the terms of (i) a binding Transaction Confirmation pursuant to Section 1.2, (ii) a Rider, (iii) the oral agreement of the parties which may be evidenced by a recorded conversation, where the parties have selected the Oral Transaction Procedure of the Base Contract, (iv) the Base Contract, and (v) these General Terms and Conditions, the terms of the documents shall govern in the priority listed in this sentence.*

The following shall be added to Section 1:

1.5 Price Locks. Unless otherwise set forth in a Transaction Confirmation or a Rider, Buyer can request to lock in the commodity price for any month(s) at any time during the Delivery Period, prior to 12:00 p.m. EST/EDT on the final day of NYMEX last day settlement for each applicable delivery month. Commodity purchase/sale prices exclude pipeline and utility distribution charges. Buyer also has the right to lock basis at a fixed price and at predetermined volumes. Basis includes interstate and intrastate pipeline transportation but does not include the commodity cost or the utility distribution

SECTION 2. DEFINITIONS

Section 2.9 is deteted in its entirety and replaced with the following;

- 2.9 "Contract" shall mean the legally-binding relationship established by (i) the Milwaukee Board of School Directors Professional Services Contract, (ii) the Base Contract, (iii) any and all binding Transaction Confirmations, (iv) any and all Riders, and (v) where the parties have selected the Oral Transaction Procedure in Section 1.2 of the Base Contract, any and all transactions that the parties have entered into through an EDI transmission or by telephone, but that have not been confirmed in a binding Transaction Confirmation.
- Section 2.10 'Contract Price' shall be amended by adding the following at the end of the section: 'The Contract Price includes reimbursement to Seller for any production, severance, ad valorem or other Taxes owed with respect to Gas prior to delivery to Buyer, all of which Taxes shall be borne and paid exclusively by Seller." 2.12 Section 2.12 "Cover Standard" is amended by deleting the words "(or an alternate fuel if elected by Buyer and replacement Gas is not available)" in the third line of Section 2.12 ("Cover Standard").

Section 2.16 is deleted in its entirely and replaced with the following:

- "Delivery Point" shall mean city gate interconnection between the Buyer's utility and the upstream Transporter or such other delivery point(s) as are agreed in a Transaction. However, to the extent that Buyerowned transportation capacity is used to deliver Gas to the Buyer, the Delivery Point shall be the receipt point of such capacity.
- Section 2.21 "Guarantor" shall be deleted in its entirety and replaced with the following: "Guarantor shall mean any entity or individual that has provided a guaranty or surety of the obligations of a party hereunder."
- Section 2.32 "Transaction Confirmation" shall be amended by adding the following at the end of the section: "The term "Transaction Confirmation" shall have the same meaning as "TC","
- 2.36 Section 2.35 "Transporter" is amended by adding the word ", storage," after the word "gathering" on the first line of Section 2.35 and adding the words "or storing" after the word "transporting" on the second line of Section 2.35.

The following sections shall be added to Section 2:

- 2.36 "Costs" means, with respect to the Non-Defaulting Party, brokerage fees, commissions and other similar transaction costs and expenses reasonably incurred by such party as a result of the Event of Default except for such amounts already included in the Net Seltlement Amount.
- "Market Price" is a price comprised of (i) the spot commodily cost of Gas as determined by CESNG in its reasonable discretion, plus (ii) all related interstate and intrastate pipeline charges required to deliver Gas to the Delivery Point, plus (III) a reasonable market based margin. Market Price does not include the cost of utility imposed charges, including but not limited to utility or pipeline balancing charges unless otherwise agreed upon.
- 2.38 "Secondary Firm" means deliveries and receipts of Gas under this Contract will be on a best-efforts basis up to Buyer's maximum daily quantily as set forth in a Transaction Confirmation, if any. Seller may interrupt performance without liability to the extent that one or more of the following conditions are present: (i) Force Majeure, (ii) curtaliment by Buyer's local distribution company, (iii) curtaliment of supply by a Gas supplier, (iv) curtaliment of storage by a storage provider, (v) curtaliment of iransportation by a Transporter, including, but not limited to, transportation between secondary firm points, (vi) recall of transportation capacity release by its releaser, (vii) curtaliment of Gas production behind a specific meter. If Seller interrupts for any other reason, Buyer's exclusive remedy shall be that Buyer may recover damages as provided in Section 3.2 of the Base Contract.

SECTION 3. PERFORMANCE OBLIGATION

- Section 3.1 shall be amended by inserting ", Secondary Firm," before "Firm" in the second sentence.
- Section 3.2 shall be amended by inserting " or Secondary Firm" after "Firm" in the first sentence.

SECTION 6. TAXES

Section 6 is deleted in its entirety and replaced with the following:

6. Buyer shall pay all federal, state, municipal and local taxes, dulles, fees, levies, premiums or other charges imposed by any governmental authority, directly or indirectly, on or with respect to the Gas and related products and services provided under the Contract, including without limitation any production, severance or ad valorem taxes, and including any taxes enacted after the Effective Date (collectively, "Taxes"). The term "Taxes" shall include any amounts imposed on Buyer discretive or on CESNG in its function as Buyer's supplier, and that are associated with the supply of Gas to Buyer (in which case the Buyer shall be responsible to relimburse CESNG for such amounts). If Buyer is exempt from any Taxes, Buyer shall provide CESNG with any takes realized level accompliance order to Buyer and the Buyer under this Contract will be with any state and/or local exemption certificate prior to the Issue date of Buyer's first invoice. All Taxes invoiced to Buyer under this Contract will be included on the invoice or in the applicable fixed price as allowed by law.

SECTION 7. BILLING, PAYMENT AND AUDIT

Section 7.1 is deleted in its entirety and replaced with the following:

- 7.1 Buyer will be billed for natural gas usage and related products and services supplied under the Contract in one of the following ways based on availability and eligibility of each Facility, which may change from time to time: (i) Dual Billing: Buyer will receive two invoices, one from CESNG for the natural gas supply and one from the utility for the amounts payable by Buyer for services provided by the utility ("Delivery Charges"); (ii) CESNG Consolidated Billing: Buyer will receive one invoice from CESNG that includes both the natural gas supply charges and the Delivery Charges, in the Consolidated Billing: Buyer will receive one invoice from CESNG that includes both the natural gas supply charges and the Delivery Charges. In the case of CESNG Consolidated Billing, Buyer agrees that (i) Buyer remains exclusively liable to the utility for all Delivery Charges, (ii) CESNG has no obligation to review Delivery Charges for accuracy, (iii) should Buyer dispute Delivery Charges, that is a matter for Buyer and the utility to resolve without any involvement or obligation on the part of CESNG, and (iv) CESNG may withhold any payments due to the utility if Buyer falls to pay CESNG invoices in accordance with this Contract. If the actual quantity of Gas delivered is not known by the invoice date, invoices will be prepared based on estimated meter data or other estimated information. The invoiced quantity will then be adjusted to the actual quantity on the following Month's invoice or as soon thereafter as actual delivery information is available Section 7.2 is deleted to the entirety and replaced with the following. thereafter as actual delivery information is available. Section 7.2 is deleted in its entirety and replaced with the following:
- 7.2 All amounts set forth in an invoice are payable to the party issuing the invoice and will include, in addition to the natural gas supply charges, Delivery Charges, and Taxes, all other amounts related to the purchase and delivery of natural gas. CESNG's invoices will be sent to Buyer in accordance with CESNG's normal billing cycle, as adjusted from time to time. CESNG's invoices are due and payable on the Payment Date set forth on the payment page of the Page Control. the cover page of the Base Contract, or such other date as set forth in a Rider, without offset or reduction of any kind, to the address set forth on the invoice; provided that if the Payment Date is not a Business Day, payment is due on the next Business Day following that date.
- Section 7.4 shall be deleted in its entirety.

SECTION 8. TITLE, WARRANTY, AND INDEMNITY

The last sentence of Section 8.1 is deleted in its entirely and replaced with the following:

"Buyer shall have responsibility for and assume any liability with respect to said Gas at and after its delivery to Buyer at the Delivery Point(s)."

SECTION 10. FINANCIAL RESPONSIBILITY

Section 10.1 will be deleted in its entirety and replaced with the following:

If GESNG has reasonable grounds: (i) to believe that Buyer's creditworthiness has become unsatisfactory; or (ii) for insecurity with respect to Buyer's performance under the Agreement, CESNG may demand, in willing, adequate assurance of future performance from Buyer in a form, in an amount, from an issuer, and for a term, all as reasonably satisfactory to CESNG ("Adequate Assurance of Performance"). To satisfy a demand, Buyer shall provide Adequate Assurance of Performance to CESNG within three (3) Business Days of the date of the written demand. If at any time CESNG requires Buyer to prepay for gas, then (i) CESNG shall be under no obligation to deliver gas if Buyer falls to pay any prepayment by its due date, and (ii) in the event the aggregate cost of any quantities of gas required by Buyer in a month in excess of the scheduled nominated quantities exceeds \$15,000.00, Buyer shall pay CESNG the aggregate cost of such quantities within three (3) Business Days of CESNG's request for

The following sections shall be amended as follows:

Section 10.2 shall be amended by deleting the word "or" before "(ix)" in such Section; and adding the following immediately after the ";" in 10.2 subclause (ix):

"(x) make any representation or warranty herein which is false or inisteading in any material respect when made or when deemed made or repeated; (xi) fall to perform any material covenant or obligation set forth in this Contract (except to the extent such failure constitutes a separate Event of Default, and except for such party's obligations to deliver or receive Gas (the exclusive remedy for which is provided in Section 3)) if such failure is not remedied except for such party's obligations to deliver or receive Gas (the exclusive remedy for which is provided in Section 3)) if such failure is not remedied within five (5) Business Days after receipt of written notice thereof; (xil) or consolidate or amalgamate with, or merge with or into, or transfer all or substantially all of its assets to, another entity and, at the time of such consolidation, amalgamation, merger or transfer, the resulting, surviving or transferee entity falls to assume all of the obligations of such party under the Contract; then

Section 10.2 shall further be amended by adding the following immediately before the "." at the end of such Section:

"provided, that no suspension shall continue for more than ten (10) Business Days unless an Early Termination Date has been declared and the Defaulling Party has been given notice thereof in accordance with Section 10.3."

10.3.1 The following sentence shall be added at the end of Section 10.3.1: "In calculating its early termination damages hereunder, the Non-Defaulting Parly may include its Costs."

SECTION 11. FORCE MAJEURE

Section 11.6 shall be deleted in its entirety.

SECTION 13. LIMITATIONS

Section 13 shall be deleted in its entirety and replaced with the following:

THIS SECTION DOES NOT WAIVE ANY LIMITATION ON INDEMNITY OR DAMAGES IMPOSED ON BUYER OR SELLER BY WISCONSIN STATE STATUTES. NOTWITHSTANDING ANY OTHER PROVISION OF THE CONTRACT TO THE CONTRARY, THE ENTIRE LIABILITY OF EACH PARTY FOR ANY AND ALL CLAIMS WILL BE LIMITED TO DIRECT ACTUAL DAMAGES ONLY, SUBJECT IN ALL CASES TO AN AFFIRMATIVE OBLIGATION ON THE PART OF EACH PARTY TO MITIGATE ITS DAMAGES, AND NEITHER PARTY WILL BE LIABLE FOR ANY CONSEQUENTIAL, EXEMPLARY, SPECIAL, INCIDENTAL OR PUNITIVE DAMAGES, INCLUDING, WITHOUT LIMITATION, LOST OPPORTUNITIES OR LOST PROFITS NOT CONTEMPLATED BY SECTION 10 ABOVE. The rights, powers, remedies and privileges provided in this Contract are cumulative and not exclusive of any rights, powers, remedies and privileges provided by law. Buyer acknowledges and agrees that the utility is exclusively responsible for the gas distribution and delivery system, that CESNG has no independent control over their systems and will have no liability for any of their acts or omissions.

SECTION 15, MISCELLANEOUS

The following sections shall be amended as follows:

Section 15.4 shall be deteted in its entirely and replaced with the following:

Supersedes Prior Agreements. The Contract embodies the parties' entire agreement and understanding and may not be contradicted by any prior or contemporaneous oral or written agreement. Notwithstanding any provision to the contrary in a prior agreement, the parties agree that this Base Contract shall supersede and replace all prior written and oral agreements or arrangements between the parties with respect to the purchase and sale of Gas and related services and that all transactions under any such prior agreement(s) are, as of the Effective Date, now governed solely by the terms of this Base Contract and shall be Transactions hereunder and a part of the single integrated agreement between the parties. This Contract may be amended only by a writing executed by both parties.

The following shall be added to the end of Section 15.8:

"On the effective date and the date of entering into each transaction, each party represents and warrants that: (i) it is duly organized, validly operating and in good standing under the laws of the jurisdiction of its formation; (ii) execution, delivery and performance of the Contract are duly authorized and do not violate any governing documents or any of its contracts or any applicable Law; (iii) it has the knowledge and experience to evaluate the merits and risks associated with the Contract; (iv) it is an "eligible contract participant" as that term is defined in the Commodity Exchange Act, as amended; (v) this Contract, each transaction hereunder, and each other document executed and delivered in accordance with this Contract constitutes a legally valid and binding obligation enforceable against it in accordance with the terms of said document, subject to any equitable defenses; (vi) it, or its credit support provider, if applicable, is not bankrupt and there are no proceedings pending or being contemplated by it, its credit support provider, if any, or, to its knowledge, threatened against it which would result in it being or becoming bankrupt and there is not pending or, to its knowledge, threatened against it, or its credit support provider, if any, any legal proceedings that could materially adversely affect its ability to perform its obligations under this Contract and each transaction hereunder; and (vii) no Event of Default with respect to it has occurred and is continuing and no such event or circumstance would occur as a result of its entering into or performing its obligations under this Contract and each transaction hereunder. The parties agree that this occur as a result of its entering into or performing its obligations under this Contract and each transaction hereunder. The parties agree that this Contract constitutes a "qualified financial contract" as that term is defined in N.Y.G.O.L. §5-701(b) and a "master netting agreement" as defined in 11 U.S.C. 101(38A) or any successor provisions.

U.S.C. 101(38A) or any successor provisions.

To the extent, if any, that a transaction does not qualify as a "first sale" as defined by the Natural Gas Act and §§ 2 and 601 of the Natural Gas Policy Act, each Party Irrevocably walves its rights, including lis rights under §§ 4-5 of the Natural Gas Act, unilaterally to seek or support a change to any terms and conditions of the Contract, including but not limited to the rate(s), charges, or classifications set forth therein. By this provision, each Party expressly waives its right to seek or support, either directly or indirectly, and by whatever means: (i) an order from the U.S. Federal Energy Regulatory Commission ("FERC") seeking to change any of the terms and conditions of the Contract agreed to by the Parties; and (ii) any refund from the other Party with respect to the Contract. Each Party further agrees that this walver and covenant shall be binding upon it notwithstanding any regulatory or market changes that may occur after the date of the Contract or any transaction entered into between the Parties. Absent the agreement of both Parties to the proposed change, the standard of review for changes to any terms and conditions of the Contract proposed by (a) a Party, to the extent that the walver set forth in this Section 15.8 is unenforceable or ineffective as to such Party due to a final determination being made under applicable law that precludes the Party from walving its rights to seek or support changes from the FERC to the terms and conditions of this Contract, (b) a non-party, or (c) the FERC acting sua sponte, shall solely be the "public interest" application of the "just and reasonable" standard of review set forth in United Gas Pipe United Co. v. Mobile Gas Service Corp., 350 U.S. 332 (1966) and Federal Power Commission v. Sierra Pacific Power Co., 350 U.S. 348 (1956) (the "Mobile-Sierra Doctrine"), as the MobileSierra Doctrine has been clarified by Morgan Stanley Capital Group, Inc. v. Public Util. Dist. No. 1 of Snohomish 128 S.Ct. 2733 (2008). 128 S.Cl. 2733 (2008).

Each party represents to the other party (with such representations deemed to be repeated by each party on each date on which a relevant transaction is entered into) that, with respect to any commodity option transaction, it is: (1) a producer, processor, or commercial user of, or a merchant handling the commodity that is the subject of the option transaction; (2) offering or has been offered or is entering into the transaction solely for purposes related to its business as such; and (3) intends to make or take, as applicable, physical delivery of the commodity if the option is exercised.

Further, Buyer warrants, represents and covenants that: (a) the data given and representations made concerning its Facility(les) are true and correct; (b) It is entering into this Contract to purchase its Gas requirements only and not for speculative or resale purposes, and that the natural gas purchased under this Contract will be consumed at the Facility(les); (c) If any transaction hereunder gives Buyer the right to adjust the Contract Quantity, or to require CESNG to provide some other quantity of Gas, Buyer's election to exercise such right is based predominantly on supply and demand factors related to Buyer's business; (d) it is the party of record of the Facility(les), or if it is not the party of record, it has the authority to enter into and bind its principal to the Contract; and (e) if Buyer is a Governmental Entity, it will not claim immunity on the grounds of sovereignty or similar grounds from enforcement of the Contract. For purposes of this Section, defenses identified in Wis. Stats. §§ 893.80 and 895.46 are not considered similar defenses to sovereign immunity. "Governmental Entity" means a municipality, country, governmental board, governmental department, commission, agency, bureau, administrative body, joint action agency, court or other similar political subdivision (including a public school district or special purpose district or authority), or public entity or instrumentability of the United States or one or more states."

15.10 Section 15.10 is amended by adding the words "and such party's affiliates" after the word "party" in the third line of such Section and by adding the parenthetical "(excluding the name of the other party)" after the word "information" in the sub-section (v).

Section 15.11 shall be deleted in its entirety and replaced with the following:

This Contract, and the rights and duties of the parties arising therefrom, shall be governed by, and interpreted and construed in accordance with, the law of the jurisdiction identified on the cover page of the Base Contract (without reference to choice of law doctrine). EACH PARTY HEREBY IRREVOCABLY WAIVES ANY AND ALL RIGHT TO TRIAL BY JURY

IN ANY PROCEEDING ARISING OUT OF OR RELATING TO THIS CONTRACT.

The following sections shall be added to Section 15:

15.13 Information and Authorization. Buyer shall designate CESNG to the utility and any other applicable Transporter(s) as an authorized recipient of Buyer's current and historical Gas billing and usage data. Buyer authorizes CESNG to take such actions it deems necessary to enroil the Buyer's accounts with the Utility as accounts to be served by CESNG solely for fulfilling CESNG's obligations under this Contract, including executing on Buyer's behalf any documents necessary to effectuate any account enrollment or election, undertaking the management of any storage or transportation capacity allocated to Buyer by the utility or other Transporter(s), disposing of storage balances, adding or deleting accounts as necessary, receiving usage nominations from the utility relating to Buyer's Gas requirements and other similar documents. Buyer shall take any actions and execute any documents as CESNG shall reasonably request in this regard.

15.14 Change in Law. CESNG may pass through or allocate, as the case may be, to Buyer any increase or decrease in CESNG's costs related to the natural gas and related products and services sold to Buyer that results from the implementation of new, or changes (including changes to transportation rates) to existing, Laws, or other requirements or changes in administration or interpretation of Laws or other requirements. "Law" means any law, rule, regulation, ordinance, statute, judicial decision, administrative order, Transporter business practices or protocol, Transporter tariff, or rule of any commission or agency with jurisdiction in the state in which the Facilities are located. Such adjusted amounts will be included in subsequent involces to Buyer. 15.15 Transporter Restrictions. If either Party receives an operational flow order or any other usage or operating instructions, restrictions or similar notice (each an "OFO") from a Transporter requiring action to be taken in connection with the flow and/or consumption of as covered by this Agraement, such party will use commercially reasonable efforts to notify the other party of such event in a timely consumption of gas covered by this Agreement, such party will use commercially reasonable efforts to notify the other party of such event in a timely

manner. Such notice may be given by electronic means. Each Party will take all commercially reasonable actions required by the OFO within the time prescribed. If OFO penalties, charges, fees, costs or expenses result from the actions or inactions of one Party, then such Party shall be solely responsible for any such penalties, charges, fees, costs or expenses. Both parties agree that OFOs may require one or both parties to buy or sell Gas quantities in the then-current market conditions, which may be appreciably higher or lower than the original pricing. If CEGS is responsible for nominations and balancing on a Transporter imposing an OFO or similar restriction, CEGS may (but is not required to) increase or decrease nominations, as appropriate, to avoid penalties.

IN WITNESS WHEREOF, the parties hereto have executed these Special Provisions as of the Effective Date.

ALL MOBY:	Dound	J	Ornat	-
Name:	David	T	Donat	

CONSTELLATION ENERGY SERVICES - NATURAL GAS, LLC

MILWAUKEE BOARD OF SCHOOL DIRECTORS ON BEHALF OF MILWAUKEE PUBLIC STREET

WILWAUKEE PUBLIC SCHOOLS

Name: Darlenne B. Driver, Ed D

Title: SuperIntendent of Schools

Name: Mark A. Sain, President

Tille: Mllwaukee Board of School Directors

(ATTACHMENT 11) ACTION ON MONTHLY FACILITIES MATTERS: FMS AWARD OF CONSTRUCTION CONTRACTS AND PROFESSIONAL SERVICE CONTRACTS RECOMMENDATION

PR#:	
Contract Number:	
RFP Number (if applicable):	

MILWAUKEE BOARD OF SCHOOL DIRECTORS PROFESSIONAL SERVICES CONTRACT

This Professional Services Contract is being entered into this 4th day of May, 2016, by and between Constellation Energy Services - Natural Gas, LLC ("Supplier") and Milwaukee Board of School Directors ("MPS"). The "Contract" between the parties consists of this Professional Services Contract ("PSC"), Supplier's NAESB Base Contract for Sale and Purchase of Natural Gas dated May 4, 2016 (hereinafter "NAESB Agreement") and Supplier's Retail Special Provisions to the NAESB Base Contract for Sale and Purchase of Natural Gas (hereinafter "Special Provisions").

1. SCOPE OF SERVICES

Supplier shall sell, and MPS shall purchase, natural gas.

MPS solicited proposals via Request for Proposal ("RFP"). The RFP and Supplier's response to the RFP dated March 30, 2016 are incorporated herein by reference.

Supplier shall provide, at its own expense, all personnel required to perform the services under this Contract. All gas purchases and sales shall be governed by the NAESB Agreement,

Subject to Section 12 of the NAESB Agreement, the Term of this Contract shall be July 1, 2016 through June 30, 2019, with three possible, one-year extensions subject to funding and an annual performance review.

No work shall commence before a Supplier receives a fully executed Contract and has been given approval to proceed. Any work performed by the Supplier prior to obtaining a fully-executed contract with approval to proceed shall not be compensated pursuant to this contract. Any continuation of the contract beyond this term must be set forth in writing and signed by the original signatories to the contract.

3. COMPENSATION

All invoicing and payment obligations shall be governed by the NAESB Agreement and Special Provisions, and consistent with Supplier's response to the RFP dated March 30, 2016. The total cost of transportation, daily balancing and basis is NYMEX plus \$0.785 per dekatherm (Dth). The Contract Quantity per year is 857,700 dekatherms for the Term of this Contract and is estimated not to exceed \$6,000,000 per year without board approval and is based on historical usage and consideration of worst case scenarios. MPS shall be responsible for payment of actual volumes which may be higher or lower based on actual use by MPS and will be purchased or sold at prevailing market rates.

4. NON APPROPRIATION OF FUNDS

This Contract is contingent upon the appropriation of sufficient funds by appropriate MPS officials. In the event that MPS fails to appropriate monies with respect to natural gas service, the Contract shall be terminated with notice from MPS to Supplier and all terminated Riders or Transactions shall be treated as Terminated Transactions arising under an Event of Default and Supplier shall be deemed to be the Non-Defaulting Party for purposes of the termination payment calculated in accordance with Section 10.3.1 of the NAESB Agreement.

5. NON-DISCRIMINATION

In the performance of work under this Contract, Supplier shall not discriminate in any way against any employee or applicant for employment because of race, religion, color, national origin, ancestry, physical handicap, mental disability, medical condition, marital status, age (over 40) or sex. This prohibition includes but is not be limited to employment; promotions, demotions and transfers; recruitment; advertising; layoff or termination; rates of pay or other forms of compensation; and selection for training, including apprenticeships. Supplier is required to include a similar provision in all subcontracts to this contract. For purposes of this Contract, Supplier, as a subsidiary of Exelon Corporation, is subject to Exelon Corporation's Equal Opportunity/Affirmative Action program and written policy.

If MPS determines Supplier has violated this non-discrimination policy, MPS may terminate this Contract without liability for undelivered services or materials. MPS may also deem the Supplier ineligible to participate in future contracts with MPS.

6. BACKGROUND CHECKS

A criminal information background check is required for all persons providing services under this Contract, including volunteers, that: (1) provide services in MPS facility(ies) on a regular and ongoing basis or more than 5 hours per week; and (2) come into contact with or have access to MPS students with or without the presence of an MPS teacher or MPS supervisor.

The purpose of this check is to ensure there is nothing that would render the person(s) unfit to perform services under this Contract where there is contact and or access to MPS students. MPS will, in its sole discretion, determine whether there is anything in a background check that would render a person unfit to work in an MPS facility with contact or access to MPS students. MPS shall perform background checks in the state(s) in which the individual resided for at least 6 months in the last 5 years, and was 18 years old or older at the time.

Supplier may perform its own criminal background checks through the Wisconsin Department of Justice Crime Information Bureau ("CIB"). Supplier shall provide the completed criminal background checks at least 10 days prior to any services being performed pursuant to this contract.

MPS will perform the necessary background investigation at the rate of \$10.00 per person. In the event Supplier chooses this option, Supplier may contact the Office of Employment Relations at 475-8280 to obtain the necessary forms. Please note that all forms must be filled out and submitted at least 30 days prior to the commencement of the services.

All background checks must be completed prior to the commencement of services under this contract. MPS will NOT be responsible for the payment of any services rendered by Supplier before the completion of these criminal information background checks.

7. INSURANCE AND PROOF OF FINANCIAL RESPONSIBILITY

Supplier understands and agrees that financial responsibility for claims or damages to any person, or to Supplier's employees and agents, shall rest with the Supplier. Supplier and its subSuppliers shall effect and maintain any insurance coverage, including, but not limited to, Workers' Compensation, Employers' Liability, General Liability, Contractual Liability, Professional Liability, Automobile Liability and Umbrella Liability to support such financial obligations. The indemnification obligation, however, shall not be reduced in any way by existence or non-existence, limitation, amount or type of damages, compensation, or benefits payable under Workers' Compensation laws or other insurance provisions.

The minimum limits of insurance required of the Supplier by MPS shall be:

Workers' Compensation Employers' Liability General Liability Professional Liability Auto Llability Umbrella (excess) Liability

Statutory Limits \$100,000 per occurrence

\$1,000,000 per occurrence/\$2,000,000 aggregate

\$1,000,000 per occurrence \$1,000,000 per occurrence \$1,000,000 per occurrence

MPS shall be named as an additional insured under Supplier's and subSuppliers' general liability insurance and umbrella liability insurance. Evidence of all required insurances of Supplier shall be given to MPS. The certificate of insurance or policies of insurance evidencing all coverages shall include a statement that MPS shall be afforded a thirty (30) day written notice of cancellation, non-renewal or material change by any of Supplier's insurers providing the coverages required by MPS for the duration of this contract.

8. IRREPARABLE HARM

It is mutually agreed the breach of this Contract on Supplier's part shall result in irreparable and continuing damage to MPS for which money damages may not provide adequate relief. Therefore, the breach of this Contract on Supplier's part shall entitle MPS to both preliminary and permanent injunctive relief and money damages insofar as they can be determined under the circumstances.

9. BREACH BY A PARTY

A failure of either party to fulfill its material obligations under this Contract shall constitute an Event of Default under the NAESB Agreement. If either party experiences an Event of Default under the NAESB Agreement, such Event of Default shall be a failure to fulfill its material obligations under this Contract. Parties agree that the process for termination for cause shall be governed by the Event of Default provision of the NAESB Agreement.

10. TERMINATION BY MPS

MPS further reserves the right to terminate this Contract at any time for any reason by giving Supplier written notice by Registered or Certified Mail of such termination. MPS will attempt to give Supplier 30 days' notice, but reserves the right to give immediate notice. In the event of said termination, Supplier shall reduce its activities hereunder, as mutually agreed to, upon receipt of said notice. Upon said termination, Supplier shall be paid for all services rendered through the date of termination, including any early termination penalties pursuant to Section 10.3.1 of the NAESB Agreement if less than 30 days' notice is provided.

11. INDEPENDENT SUPPLIER

Supplier agrees and stipulates that in performing this Contract, it is acting as an Independent Supplier, and that no relationship of employer and employee, partnership or joint venture is created by this Contract. Supplier has exclusive control over work hours, location, and other details of such services, and MPS' sole interest is to ensure that said service shall be performed and rendered in a competent, safe, efficient, timely and satisfactory manner in accordance with the terms of this Contract.

Supplier has the sole obligation to provide for and pay any contribution or taxes required by federal, state or local authorities imposed on or measured by income. Supplier specifically covenant not to file any complaint, charge, or claim with any local, state or federal agency or court in which Supplier claims to be or to have been an employee of MPS during the period of time covered by this Contract and that if any such agency or court assumes jurisdiction of any complaint, charge or claim against MPS on Supplier's behalf, Supplier will request such agency or court to dismiss such matter. MPS shall not be charged any obligation or responsibility whatsoever of extending any fringe benefits which may be extended to MPS employees, including any insurance, or pension plans.

Supplier further agrees that MPS is not to be charged with the obligation or responsibility of extending any fringe benefits such as hospital, medical and life insurance, or pension plans which may be extended to employees of MPS from time-to-time,

12. ASSIGNMENT LIMITATION

This Contract shall be binding upon and inure to the benefit of the parties and their successors and assigns; provided, however, that neither party shall assign its obligations hereunder without the prior written consent of the other, such consent not to be unreasonably withheld, conditioned or delayed.

13. PROHIBITED PRACTICES

- A. Supplier during the period of this contract shall not hire, retain or use for compensation any member, officer, or employee of MPS to perform services under this Contract, or any other person who, to the knowledge of Supplier, has a conflict of interest.
- B. Supplier hereby attests it is familiar with MPS's Code of Ethics, providing in pertinent part, "[a]n employee of Milwaukee Public Schools may not accept any gift or gratuity in excess of \$25.00 annually from any person, persons, group or any firm which does business with or is attempting to do business with MPS."
- C. Supplier shall adhere to the MPS' Livable Wage Policy that requires all Suppliers to pay their employees a minimum of \$7,70 per hour.
- D. No current or former MPS employee may perform services on a professional services contract without the prior written consent of the Director of Human Relations of her designee.
 - 14. NOTICES

Notices are provided for in Section 9 of the NAESB Agreement.

The waiver or failure of either Party to exercise in any respect any rights provided for in this Contract shall not be deemed a waiver of any further right under this Contract.

16. INTEGRATION/SEVERABILITY

This Contract, the NAESB Agreement and Special Provisions constitute the entire Contract among the Parties with respect to the subject matter hereof and supersede all prior proposals, negotiations, conversations, discussions and Contracts among the Parties concerning the subject matter hereof. No amendment or modification of any provision of this Contract shall be effective unless the same shall be in writing and signed by both Parties.

If any term or provision of this Contract should be declared invalid by a court of competent jurisdiction or by operation of law, the remaining terms and provisions of this Contract shall be interpreted as if such invalid Contracts or covenants were not contained herein.

17. ORDER OF PRIORITY

In the event of a conflict between the documents which make up this Contract, the terms of the NAESB Agreement or Special Provisions shall control. Specific exceptions to this paragraph are noted.

18. CHOICE OF LAW & FORUM

With respect to any conflict in the terms regarding choice of law, venue or construction of this Contract, the terms of the PSC shall control. The state and federal courts of Wisconsin shall be the sole forum for all disputes arising of this contract. Venue shall be the Milwaukee County Circuit Court or the Eastern District of Wisconsin. The validity, construction, enforcement and effect of this Contract shall be governed solely by the laws of the State of Wisconsin.

19. TIMING

Time is of the essence in this Contract,

20. CERTIFICATION REGARDING DEBARMENT OR SUSPENSION

Supplier certifies that neither Supplier or its principals; its subcontractors or their principals; the sub-recipients (if applicable) or their principals are suspended, debarred, proposed for debarment, voluntarily excluded from covered transactions, or otherwise disqualified by any federal department or agency from doing business with the Federal Government pursuant to Executive Orders 12549 and 12689. Supplier specifically covenants that neither the Supplier or its principals, its sub-Suppliers or their principals, or the sub-recipients (if applicable) or their principals are included on the Excluded Parties List System ("EPLS") maintained by the General Services Administration ("GSA").

21. STUDENT DATA

Supplier acknowledges that student data is protected by both federal and state law. See Wis. Stat. § 118.125; 20 U.S.C. § 1232g(b); 34 C.F.R. § 99.1 et seq. If MPS determines that Supplier has disclosed any student record information in violation of either federal or state law, without prejudice to any other rights or remedies the MPS may have, MPS shall be entitled to immediately terminate this and every other existing Contract without further liability. Moreover, MPS may bar Supplier from future MPS contracts for varying periods up to and including permanent debarment.

22. MPS LOGO/PUBLICITY

No Supplier shall use the MPS Logo in its literature or issue a press release about the subject of this Contract without prior written notice to and written approval of MPS' Director of Communication and Public Affairs.

23. PUBLIC RECORDS

Both parties understand that the Board is bound by the Wisconsin Public Records Law, and as such, all of the terms of this Contract are subject to and conditioned on the provisions of Wis. Stat. § 19.21, et seq. Supplier acknowledges that it is obligated to assist the Board in retaining and producing records that are subject to Wisconsin Public Records Law, and that the failure to do so shall constitute a material breach of this Contract, and that the Supplier must defend and hold the Board harmless from liability under the law. Except as otherwise authorized, those records shall be maintained for a period of seven years after receipt of final payment under this Contract.

24. HUB REQUIREMENT

The HUB requirement on this contract is 0%.

25. COUNTERPARTS

This Contract may be executed and delivered in counterparts (including facsimile transmission), each of which will be deemed an original and all of which together will constitute one and the same instrument. By way of clarification however, this Contract will not be effective unless both Parties have executed a counterpart.

26. AUTHORITY

By signing below, each individual additionally warrants that he or she is authorized to sign this Contract on behalf of the Party for which it was executed.

IN WITNESS WHEREOF, the parties here to have executed this Contract on the day, month and year first above written.

Constellation Energy Services - Natural Gas, LLC

Milwaukee Board Of School Directors

Dated: 6/6/6

Dated: 05-27-16

all 188

Donat

(Principal or Administrator Signature)

Updated 7/10

Address: 9960 Corporate Campus Drive, Suite 2000	By: A Marian
<u>Louisville, KY 40223</u> Phone Number: <u>(502) 426-4500</u>	Darienne B. Driver, Ed.D.
Thone runner. (1904) 120 120	Superintendent of Schools
Tax Id or St	By: The San
Budget Code:	Mark Sain, President
Page coasi	Milwaukee Board of School Directors

Base Contract for Sale and Purchase of Natural Gas

This Base Contract is entered into as of the following date: May 4, 2016 ("Effective Date")

The parties to this Base Contract are the following:

PARTY A Constellation Energy Services – Natural Gas, LLC	PARTY NAME	PARTY B Milwaukee Board of School Directors on behalf of Milwaukee Public Schools ("Buyer")		
("CESNG" or "Seller") 9960 Corporate Campus Drive, Suite 2000 Louisville, KY 40223	ADDRESS	5225 W. Vilet Street Milwaukee, WI 53208		
	BUSINESS WEBSITE			
CESNG19189	CONTRACT NUMBER			
	D-U-N-S® NUMBER			
US FEDERAL: O OTHER:	TAX ID NUMBERS	□ US FEDERAL: □ OTHER:		
Wisconsin	JURISDICTION OF ORGANIZATION			
□ Corporation ⊠ LLC □ Limited Partnership □ Partnership □ LLP □ Other:	COMPANY TYPĖ	□ Corporation □ LLC □ Limited Partnership □ Partnership □ LLP □ Other:		
	GUARANTOR (IF APPLICABLE)			
CON	TACT INFORMAT	ION		
9860 Corporate Campus Drive, Suite 2000, Louisville, KY 40223 ATTN: Contracts Department TEL#: (502) 426-4500	COMMERCIAL (If different from Contract/Legal notices)	ATTN: TEL#: FAX#: EMAJL:		
9960 Corporate Campus Drive, Suite 2000, Louisville, KY 40223 ATTN: Contracts Department TEL#: (502) 426-4500 FAX#: (502) 214-6381	SCHEDULING (If different from Contract/Legal notices)	ATTN: TELH: FAX#: EMAIL:		
9960 Corporate Campus Drive, Suite 2000, Louisville, KY 40223 ATTN: Contracts Department TEL#: (502) 426-4500	= CONTRACT AND LEGAL NOTICES	1224 North 11 th Street, Milwaukee, WI 53233 ATTN: John Linn TEL#: (414)-283-4600 FAX#: (414) 283-4682 EMAIL: Iinnja@milwaukee.k12.wi.us		
9960 Corporate Campus Drive, Suite 2000, Louisville, KY 40223 ATTN: Contracts Department TEL#: (502) 426-4500	CREDIT (if different from Contract/Legal notices)	n/a ATTN: TEL#: FAX#: EMAIL:		
9960 Corporate Campus Drive, Suite 2000, Louisville, KY 40223 ATTN: Contracts Department TEL#: (502) 426-4500 FAX#: (502) 214-6381	TRANSACTION CONFIRMATIONS (if different from Contract/Legal notices)	ATTN: TEL#: FAX#: EMAIL:		
ACCOUNTING INFORMATION				
ATTN: see invoice TEL#: FAX#: EMAIL:	• INVOICES • PAYMENTS • SETTLEMENTS	ATTN: TEL#: FAX#: EMAIL:		
BANK: see invoice ABA: ACCT: OTHER DETAILS:	WIRE TRANSFER NUMBERS (IF APPLICABLE)	BANK: N/A ABA: ACCT: OTHER DETAILS:		
BANK: see Invoice ABA:	ACH NUMBERS (IF APPLICABLE)	BANK: N/A ABA: ACCY: OTHER DETAILS:		
ATTN: gee invoice Address:	CHECKS (IF APPLICABLE)	ATTN:N/A ADDRESS:		

Base Contract for Sale and Purchase of Natural Gas

(Continued)

This Base Contract incorporates by reference for all purposes the General Terms and Conditions for Sale and Purchase of Natural Gas published by the North American Energy Standards Board. The parties hereby agree to the following provisions offered in said General Terms and Conditions. In the event the parties fail to check a box, the specified default provision shall apply. Select the appropriate box(es) from each section:

Section 3.2	Transaction Procedure Section 2.7 Confirm Deadline Section 2.8		Oral (default) Written 2 Business Days after receipt (default) Business Days after receipt Seller (default)	Section 10.2 Additional Events of Default	⊠ □	No Additional Events of Default (default) Indebtedness Cross Default Party A: Party B: Transactional Cross Default Specified Transactions:
Performance OR Obligation			Buyer			
Note: The following Spot Price Publication applies to both of the immediately preceding. Section 2.31	Performance	OR		Early Termination		Early Termination Damages Apply (default)
Section 2.31				Damages .		Early Termination Damages Do Not Apply
Section 6 Buyer Pays At and After Delivery Point Taxes OR OR	Immediately preced Section 2.31 Spot Price	ding. OR	Gas Dally Midpoint (default)	Other Agreement		図 Bilateral (default)
		OF	(default)			Olher Agreement Setoffs Do Not Apply
Section 7.2			(default)			Wisconsin
Section 7.2 Urice transfer (default) Method of Payment Automated Clearinghouse Credit (ACH) Check Section 15.10 Confidentiality applies (default) Confidentiality Confidenti		0	Wire transfer (default) Automated Clearinghouse Credit (ACH)		OR	
Section 7.7 Netting applies (default) Netting OR Netting does not apply	Netting	OF	R Netting does not apply			
⊠ Special Provisions Number of sheets attached <u>3</u> □ Addendum(s):		ons N	lumber of sheets attached 3			

IN WITNESS WHEREOF, the parties hereto have executed this Base Contract as of the Effective Date.

CONSTELLATION ENERGY SERVICES - NATURAL GAS, LLC

By:

WHEREOF, the parties hereto have executed this Base Contract as of the Effective Date.

MILWAUKEE BOARD OF SCHOOL DIRECTORS ON BEHALF-OF MILWAUKEE PUBLIC SCHOOLS

SIGNATURE

By:

Wark A. Sain, President

Milwaukee Board of School Directors

General Terms and Conditions Base Contract for Sale and Purchase of Natural Gas

PURPOSE AND PROCEDURES SECTION 1.

These General Terms and Conditions are intended to facilitate purchase and sale transactions of Gas on a Firm or Interruptible basis. "Buyer" refers to the party receiving Gas and "Seller" refers to the party delivering Gas. The entire agreement between the parties shall be the Contract as defined in Section 2.9.

The parties have selected either the "Oral Transaction Procedure" or the "Written Transaction Procedure" as indicated on the Base Contract.

Oral Transaction Procedure:

The parties will use the following Transaction Confirmation procedure. Any Gas purchase and sale transaction may be 1.2. effectuated in an EDI transmission or telephone conversation with the offer and acceptance constituting the agreement of the parties. The parties shall be legally bound from the time they so agree to transaction terms and may each rely thereon. Any such transaction shall be considered a "writing" and to have been "signed". Notwithstanding the foregoing sentence, the parties agree that Confirming Party shall, and the other party may, confirm a telephonic transaction by sending the other party a Transaction Confirmation by facsimile, EDI or mutually agreeable electronic means within three Business Days of a transaction covered by this Section 1.2 (Oral Transaction Procedure) provided that the failure to send a Transaction Confirmation shall not invalidate the oral agreement of the parties. Confirming Party adopts its confirming letterhead, or the like, as its signature on any Transaction Confirmation as the identification and authentication of Confirming Party. If the Transaction Confirmation contains any provisions other than those relating to the commercial terms of the transaction (i.e., price, quantity, performance obligation, delivery point, period of delivery and/or transportation conditions), which modify or supplement the Base Contract or General Terms and Conditions of this Contract (e.g., arbitration or additional representations and warranties), such provisions shall not be deemed to be accepted pursuant to Section 1.3 but must be expressly agreed to by both parties; provided that the foregoing shall not invalidate any transaction agreed to by the parties.

Written Transaction Procedure:

- The parties will use the following Transaction Confirmation procedure. Should the parties come to an agreement regarding 1.2. a Gas purchase and sale transaction for a particular Delivery Period, the Confirming Party shall, and the other party may, record that agreement on a Transaction Confirmation and communicate such Transaction Confirmation by facsimile, EDI or mutually agreeable electronic means, to the other party by the close of the Business Day following the date of agreement. The parties acknowledge that their agreement will not be binding until the exchange of nonconflicting Transaction Confirmations or the passage of the Confirm Deadline without objection from the receiving party, as provided in Section 1.3.
- If a sending party's Transaction Confirmation is materially different from the receiving party's understanding of the agreement referred to in Section 1.2, such receiving party shall notify the sending party via facsimile, EDI or mutually agreeable electronic means by the Confirm Deadline, unless such receiving party has previously sent a Transaction Confirmation to the sending party. The failure of the receiving party to so notify the sending party in writing by the Confirm Deadline constitutes the receiving party's agreement to the terms of the transaction described in the sending party's Transaction Confirmation. If there are any material differences between timely sent Transaction Confirmations governing the same transaction, then neither Transaction Confirmation shall be binding until or unless such differences are resolved including the use of any evidence that clearly resolves the differences in the Transaction Confirmations. In the event of a conflict among the terms of (i) a binding Transaction Confirmation pursuant to Section 1.2, (ii) the oral agreement of the parties which may be evidenced by a recorded conversation, where the parties have selected the Oral Transaction Procedure of the Base Contract, (iii) the Base Contract, and (iv) these General Terms and Conditions, the terms of the documents shall govern in the priority listed in this sentence.
- The parties agree that each party may electronically record all telephone conversations with respect to this Contract between their respective employees, without any special or further notice to the other party. Each party shall obtain any necessary consent of its agents and employees to such recording. Where the parties have selected the Oral Transaction Procedure in Section 1.2 of the Base Contract, the parties agree not to contest the validity or enforceability of telephonic recordings entered into in accordance with the requirements of this Base Contract.

SECTION 2. **DEFINITIONS**

The terms set forth below shall have the meaning ascribed to them below. Other terms are also defined elsewhere in the Contract and shall have the meanings ascribed to them herein.

- "Additional Event of Default" shall mean Transactional Cross Default or Indebtedness Cross Default, each as and if selected by the parties pursuant to the Base Contract.
- "Affiliate" shall mean, in relation to any person, any entity controlled, directly or indirectly, by the person, any entity that controls, directly or indirectly, the person or any entity directly or indirectly under common control with the person. For this purpose, "control" of any entity or person means ownership of at least 50 percent of the voting power of the entity or person.
- "Alternative Damages" shall mean such damages, expressed in dollars or dollars per MMBtu, as the parties shall agree upon in the Transaction Confirmation, in the event either Seller or Buyer fails to perform a Firm obligation to deliver Gas in the case of Seller or to receive Gas in the case of Buyer.

- 2.4. "Base Contract" shall mean a contract executed by the parties that incorporates these General Terms and Conditions by reference; that specifies the agreed selections of provisions contained herein; and that sets forth other information required herein and any Special Provisions and addendum(s) as identified on page one.
- 2.5. "British thermal unit" or "Btu" shall mean the International BTU, which is also called the Btu (IT).
- 2.6. "Business Day(s)" shall mean Monday through Friday, excluding Federal Banking Holidays for transactions in the U.S.
- 2.7. "Confirm Deadline" shall mean 5:00 p.m. in the receiving party's time zone on the second Business Day following the Day a Transaction Confirmation is received or, if applicable, on the Business Day agreed to by the parties in the Base Contract; provided, if the Transaction Confirmation is time stamped after 5:00 p.m. in the receiving party's time zone, it shall be deemed received at the opening of the next Business Day.
- 2.8. "Confirming Party" shall mean the party designated in the Base Contract to prepare and forward Transaction Confirmations to the other party.
- 2.9. "Contract" shall mean the legally-binding relationship established by (i) the Base Contract, (ii) any and all binding Transaction Confirmations and (iii) where the parties have selected the Oral Transaction Procedure in Section 1.2 of the Base Contract, any and all transactions that the parties have entered into through an EDI transmission or by telephone, but that have not been confirmed in a binding Transaction Confirmation, all of which shall form a single integrated agreement between the parties.
- 2.10. "Contract Price" shall mean the amount expressed in U.S. Dollars per MMBtu to be paid by Buyer to Seller for the purchase of Gas as agreed to by the parties in a transaction.
- 2.11. "Contract Quantity" shall mean the quantity of Gas to be delivered and taken as agreed to by the parties in a transaction.
- 2.12. "Cover Standard", as referred to in Section 3.2, shall mean that if there is an unexcused failure to take or deliver any quantity of Gas pursuant to this Contract, then the performing party shall use commercially reasonable efforts to (i) if Buyer is the performing party, obtain Gas, (or an alternate fuel if elected by Buyer and replacement Gas is not available), or (ii) if Seller is the performing party, sell Gas, in either case, at a price reasonable for the delivery or production area, as applicable, consistent with: the amount of notice provided by the nonperforming party; the immediacy of the Buyer's Gas consumption needs or Seller's Gas sales requirements, as applicable; the quantities involved; and the anticipated length of failure by the nonperforming party.
- 2.13. "Credit Support Obligation(s)" shall mean any obligation(s) to provide or establish credit support for, or on behalf of, a party to this Contract such as cash, an irrevocable standby letter of credit, a margin agreement, a prepayment, a security interest in an asset, guaranty, or other good and sufficient security of a continuing nature.
- 2.14. "Day" shall mean a period of 24 consecutive hours, coextensive with a "day" as defined by the Receiving Transporter in a particular transaction.
- 2.15. "Delivery Period" shall be the period during which deliveries are to be made as agreed to by the parties in a transaction.
- 2.16. "Delivery Point(s)" shall mean such point(s) as are agreed to by the parties in a transaction.
- 2.17. "EDI" shall mean an electronic data interchange pursuant to an agreement entered into by the parties, specifically relating to the communication of Transaction Confirmations under this Contract.
- 2.18. "EFP" shall mean the purchase, sale or exchange of natural Gas as the "physical" side of an exchange for physical transaction involving gas futures contracts. EFP shall incorporate the meaning and remedies of "Firm", provided that a party's excuse for nonperformance of its obligations to deliver or receive Gas will be governed by the rules of the relevant futures exchange regulated under the Commodity Exchange Act.
- 2.19. "Firm" shall mean that either party may interrupt its performance without liability only to the extent that such performance is prevented for reasons of Force Majeure; provided, however, that during Force Majeure interruptions, the party invoking Force Majeure may be responsible for any Imbalance Charges as set forth in Section 4.3 related to its interruption after the nomination is made to the Transporter and until the change in deliveries and/or receipts is confirmed by the Transporter.
- 2.20. "Gas" shall mean any mixture of hydrocarbons and noncombustible gases in a gaseous state consisting primarily of methane.
- 2.21. "Guarantor" shall mean any entity that has provided a guaranty of the obligations of a party hereunder.
- 2.22. "Imbalance Charges" shall mean any fees, penalties, costs or charges (in cash or in kind) assessed by a Transporter for failure to satisfy the Transporter's balance and/or nomination requirements.
- 2.23. "Indebtedness Cross Default" shall mean if selected on the Base Contract by the parties with respect to a party, that it or its Guarantor, if any, experiences a default, or similar condition or event however therein defined, under one or more agreements or instruments, individually or collectively, relating to indebtedness (such indebtedness to include any obligation whether present or future, contingent or otherwise, as principal or surety or otherwise) for the payment or repayment of borrowed money in an aggregate amount greater than the threshold specified in the Base Contract with respect to such party or its Guarantor, if any, which results in such indebtedness becoming immediately due and payable.
- 2.24. "Interruptible" shall mean that either party may interrupt its performance at any time for any reason, whether or not caused by an event of Force Majeure, with no liability, except such interrupting party may be responsible for any imbalance Charges as set forth in Section 4.3 related to its interruption after the nomination is made to the Transporter and until the change in deliveries and/or receipts is confirmed by Transporter.

- 2.25. "MMBtu" shall mean one million British thermal units, which is equivalent to one dekatherm.
- 2.26. "Month" shall mean the period beginning on the first Day of the calendar month and ending immediately prior to the commencement of the first Day of the next calendar month.
- 2.27. "Payment Date" shall mean a date, as indicated on the Base Contract, on or before which payment is due Seller for Gas received by Buyer in the previous Month.
- 2.28. "Receiving Transporter" shall mean the Transporter receiving Gas at a Delivery Point, or absent such receiving Transporter, the Transporter delivering Gas at a Delivery Point.
- 2.29. "Scheduled Gas" shall mean the quantity of Gas confirmed by Transporter(s) for movement, transportation or management.
- 2.30. "Specified Transaction(s)" shall mean any other transaction or agreement between the parties for the purchase, sale or exchange of physical Gas, and any other transaction or agreement identified as a Specified Transaction under the Bese Contract.
- 2.31. "Spot Price" as referred to in Section 3.2 shall mean the price listed in the publication Indicated on the Base Contract, under the listing applicable to the geographic location closest in proximity to the Delivery Point(s) for the relevant Day; provided, if there is no single price published for such location for such Day, but there is published a range of prices, then the Spot Price shall be the average of such high and low prices. If no price or range of prices is published for such Day, then the Spot Price shall be the average of the following: (i) the price (determined as stated above) for the first Day for which a price or range of prices is published that next precedes the relevant Day; and (ii) the price (determined as stated above) for the first Day for which a price or range of prices is published that next follows the relevant Day.
- 2.32. "Transaction Confirmation" shall mean a document, similar to the form of Exhibit A, setting forth the terms of a transaction formed pursuant to Section 1 for a particular Delivery Period.
- 2.33. "Transactional Cross Default" shall mean if selected on the Base Contract by the parties with respect to a party, that it shall be in default, however therein defined, under any Specified Transaction.
- 2.34. "Termination Option" shall mean the option of either party to terminate a transaction in the event that the other party fails to perform a Firm obligation to deliver Gas in the case of Seller or to receive Gas in the case of Buyer for a designated number of days during a period as specified on the applicable Transaction Confirmation.
- 2.35. "Transporter(s)" shall mean all Gas gathering or pipeline companies, or local distribution companies, acting in the capacity of a transporter, transporting Gas for Seller or Buyer upstream or downstream, respectively, of the Delivery Point pursuant to a particular transaction.

SECTION 3. PERFORMANCE OBLIGATION

3.1. Seller agrees to sell and deliver, and Buyer agrees to receive and purchase, the Contract Quantity for a particular transaction in accordance with the terms of the Contract. Sales and purchases will be on a Firm or Interruptible basis, as agreed to by the parties in a transaction.

The parties have selected either the "Cover Standard" or the "Spot Price Standard" as indicated on the Base Contract.

Cover Standard:

The sole and exclusive remedy of the parties in the event of a breach of a Firm obligation to deliver or receive Gas shall be recovery of the following: (i) in the event of a breach by Seller on any Day(s), payment by Seller to Buyer in an amount equal to the positive difference, if any, between the purchase price paid by Buyer utilizing the Cover Standard and the Contract Price, adjusted for commercially reasonable differences in transportation costs to or from the Delivery Point(s), multiplied by the difference between the Contract Quantity and the quantity actually delivered by Seller for such Day(s) excluding any quantity for which no replacement is available; or (ii) in the event of a breach by Buyer on any Day(s), payment by Buyer to Seller in the amount equal to the positive difference, if any, between the Contract Price and the price received by Seller utilizing the Cover Standard for the resale of such Gas, adjusted for commercially reasonable differences in transportation costs to or from the Delivery Point(s), multiplied by the difference between the Contract Quantity and the quantity actually taken by Buyer for such Day(s) excluding any quantity for which no sale is available; and (iii) in the event that Buyer has used commercially reasonable efforts to replace the Gas or Seller has used commercially reasonable efforts to sell the Gas to a third party, and no such replacement or sale is available for all or any portion of the Contract Quantity of Gas, then in addition to (i) or (ii) above, as applicable, the sole and exclusive remedy of the performing party with respect to the Gas not replaced or sold shall be an amount equal to any unfavorable difference between the Contract Price and the Spot Price, adjusted for such transportation to the applicable Delivery Point, multiplied by the quantity of such Gas not replaced or sold. Imbalance Charges shall not be recovered under this Section 3.2, but Seller and/or Buyer shall be responsible for imbalance Charges, If any, as provided in Section 4.3. The amount of such unfavorable difference shall be payable five Business Days after presentation of the performing party's invoice, which shall set forth the basis upon which such amount was calculated.

Spot Price Standard:

3.2. The sole and exclusive remedy of the parties in the event of a breach of a Firm obligation to deliver or receive Gas shall be recovery of the following: (I) in the event of a breach by Seller on any Day(s), payment by Seller to Buyer in an amount equal to the difference between the Contract Quantity and the actual quantity delivered by Seller and received by Buyer for such Day(s), multiplied by the positive difference, if any, obtained by subtracting the Contract Price from the Spot Price; or (ii) in the event of a breach by Buyer on any Day(s), payment by Buyer to Seller in an amount equal to the difference between the Contract Quantity and the actual quantity delivered by Seller and received by Buyer for such Day(s), multiplied by the positive difference, if any,

obtained by subtracting the applicable Spot Price from the Contract Price. Imbalance Charges shall not be recovered under this Section 3.2, but Seller and/or Buyer shall be responsible for imbalance Charges, if any, as provided in Section 4.3. The amount of such unfavorable difference shall be payable five Business Days after presentation of the performing party's invoice, which shall set forth the basis upon which such amount was calculated.

- Notwithstanding Section 3.2, the parties may agree to Alternative Damages in a Transaction Confirmation executed in writing by both parties.
- In addition to Sections 3.2 and 3.3, the parties may provide for a Termination Option in a Transaction Confirmation executed in writing by both parties. The Transaction Confirmation containing the Termination Option will designate the length of nonperformance triggering the Termination Option and the procedures for exercise thereof, how damages for nonperformance will be compensated, and how liquidation costs will be calculated.

SECTION 4. TRANSPORTATION, NOMINATIONS, AND IMBALANCES

- Seller shall have the sole responsibility for transporting the Gas to the Delivery Point(s). Buyer shall have the sole responsibility for transporting the Gas from the Delivery Point(s).
- The parties shall coordinate their nomination activities, giving sufficient time to meet the deadlines of the affected Transporter(s). Each party shall give the other party timely prior Notice, sufficient to meet the requirements of all Transporter(s) involved in the transaction, of the quantities of Gas to be delivered and purchased each Day. Should either party become aware that actual deliveries at the Delivery Point(s) are greater or lesser than the Scheduled Gas, such party shall promptly notify the other party.
- The parties shall use commercially reasonable efforts to avoid imposition of any Imbalance Charges. If Buyer or Seller receives an involce from a Transporter that includes imbalance Charges, the parties shall determine the validity as well as the cause of such Imbalance Charges. If the Imbalance Charges were incurred as a result of Buyer's receipt of quantities of Gas greater than or less than the Scheduled Gas, then Buyer shall pay for such Imbalance Charges or reimburse Seller for such Imbalance Charges paid by Seller. If the Imbalance Charges were incurred as a result of Seller's delivery of quantities of Gas greater than or less than the Scheduled Gas, then Seller shall pay for such Imbalance Charges or reimburse Buyer for such Imbalance Charges paid by Buyer.

SECTION 5. QUALITY AND MEASUREMENT

All Gas delivered by Seller shall meet the pressure, quality and heat content requirements of the Receiving Transporter. The unit of quantity measurement for purposes of this Contract shall be one MMBtu dry. Measurement of Gas quantities hereunder shall be in accordance with the established procedures of the Receiving Transporter.

SECTION 6. **TAXES**

The partles have selected either "Buyer Pays At and After Delivery Point" or "Seller Pays Before and At Delivery Point" as Indicated on the Base Contract.

Buyer Pays At and After Delivery Point:

Seller shall pay or cause to be paid all taxes, fees, levies, penalties, licenses or charges imposed by any government authority ("Taxes") on or with respect to the Gas prior to the Delivery Point(s). Buyer shall pay or cause to be paid all Taxes on or with respect to the Gas at the Delivery Point(s) and all Taxes after the Delivery Point(s). If a party is required to remit or pay Taxes that are the other party's responsibility hereunder, the party responsible for such Taxes shall promptly reimburse the other party for such Taxes. Any party entitled to an exemption from any such Taxes or charges shall furnish the other party any necessary documentation thereof.

Seller Pays Before and At Delivery Point:

Seller shall pay or cause to be paid all taxes, fees, levies, penalties, licenses or charges imposed by any government authority ("Taxes") on or with respect to the Gas prior to the Delivery Point(s) and all Taxes at the Delivery Point(s). Buyer shall pay or cause to be paid all Taxes on or with respect to the Gas after the Delivery Point(s). If a party is required to remit or pay Taxes that are the other party's responsibility hereunder, the party responsible for such Taxes shall promptly reimburse the other party for such Taxes. Any party entitled to an exemption from any such Taxes or charges shall furnish the other party any necessary documentation thereof.

SECTION 7. BILLING, PAYMENT, AND AUDIT

- Seller shall invoice Buyer for Gas delivered and received in the preceding Month and for any other applicable charges, providing supporting documentation acceptable in industry practice to support the amount charged. If the actual quantity delivered is not known by the billing date, billing will be prepared based on the quantity of Scheduled Gas. The invoiced quantity will then be adjusted to the actual quantity on the following Month's billing or as soon thereafter as actual delivery Information is available.
- Buyer shall remit the amount due under Section 7.1 in the manner specified in the Base Contract, in immediately available funds, 7.2. on or before the later of the Payment Date or 10 Days after receipt of the Invoice by Buyer; provided that if the Payment Date is not a Business Day, payment is due on the next Business Day following that date. In the event any payments are due Buyer hereunder, payment to Buyer shall be made in accordance with this Section 7.2.
- In the event payments become due pursuant to Sections 3.2 or 3.3, the performing party may submit an invoice to the nonperforming party for an accelerated payment setting forth the basis upon which the invoiced amount was calculated. Payment from the nonperforming party will be due five Business Days after receipt of invoice.
- If the invoiced party, in good faith, disputes the amount of any such invoice or any part thereof, such invoiced party will pay such amount as it concedes to be correct; provided, however, if the invoiced party disputes the amount due, it must provide supporting

documentation acceptable in industry practice to support the amount paid or disputed without undue delay. In the event the parties are unable to resolve such dispute, either party may pursue any remedy available at law or in equity to enforce its rights pursuant to this Section.

- 7.5. If the invoiced party fails to remit the full amount payable when due, interest on the unpaid portion shall accrue from the date due until the date of payment at a rate equal to the lower of (i) the then-effective prime rate of interest published under "Money Rates" by The Wall Street Journal, plus two percent per annum; or (ii) the maximum applicable lawful interest rate.
- 7.6. A party shall have the right, at its own expense, upon reasonable Notice and at reasonable times, to examine and audit and to obtain copies of the relevant portion of the books, records, and telephone recordings of the other party only to the extent reasonably necessary to verify the accuracy of any statement, charge, payment, or computation made under the Contract. This right to examine, audit, and to obtain copies shall not be available with respect to proprietary information not directly relevant to transactions under this Contract. All invoices and billings shall be conclusively presumed final and accurate and all associated claims for under- or overpayments shall be deemed waived unless such invoices or billings are objected to in writing, with adequate explanation and/or documentation, within two years after the Month of Gas delivery. All retroactive adjustments under Section 7 shall be paid in full by the party owing payment within 30 Days of Notice and substantiation of such inaccuracy.
- 7.7. Unless the parties have elected on the Base Contract not to make this Section 7.7 applicable to this Contract, the parties shall net all undisputed amounts due and owing, and/or past due, arising under the Contract such that the party owing the greater amount shall make a single payment of the net amount to the other party in accordance with Section 7; provided that no payment required to be made pursuant to the terms of any Credit Support Obligation or pursuant to Section 7.3 shall be subject to netting under this Section. If the parties have executed a separate netting agreement, the terms and conditions therein shall prevail to the extent inconsistent herewith.

SECTION 8. TITLE, WARRANTY, AND INDEMNITY

- 8.1. Unless otherwise specifically agreed, title to the Gas shall pass from Seller to Buyer at the Delivery Point(s). Seller shall have responsibility for and assume any liability with respect to the Gas prior to its delivery to Buyer at the specified Delivery Point(s). Buyer shall have responsibility for and assume any liability with respect to said Gas after its delivery to Buyer at the Delivery Point(s).
- 8.2. Seller warrants that it will have the right to convey and will transfer good and merchantable title to all Gas sold hereunder and delivered by it to Buyer, free and clear of all liens, encumbrances, and claims. EXCEPT AS PROVIDED IN THIS SECTION 8.2 AND IN SECTION 15.8, ALL OTHER WARRANTIES, EXPRESS OR IMPLIED, INCLUDING ANY WARRANTY OF MERCHANTABILITY OR OF FITNESS FOR ANY PARTICULAR PURPOSE, ARE DISCLAIMED.
- 8.3. Seller agrees to indemnify Buyer and save it harmless from all losses, liabilities or claims including reasonable attorneys' fees and costs of court ("Claims"), from any and all persons, arising from or out of claims of title, personal injury (including death) or property damage from sald Gas or other charges thereon which attach before title passes to Buyer. Buyer agrees to indemnify Seller and save it harmless from all Claims, from any and all persons, arising from or out of claims regarding payment, personal injury (including death) or property damage from said Gas or other charges thereon which attach after title passes to Buyer.
- 8.4. The parties agree that the delivery of and the transfer of title to all Gas under this Contract shall take place within the Customs Territory of the United States (as defined in general note 2 of the Harmonized Tariff Schedule of the United States 19 U.S.C. §1202, General Notes, page 3); provided, however, that in the event Seller took title to the Gas outside the Customs Territory of the United States, Seller represents and warrants that it is the importer of record for all Gas entered and delivered into the United States, and shall be responsible for entry and entry summary filings as well as the payment of duties, taxes and fees, if any, and all applicable record keeping requirements.
- 8.5. Notwithstanding the other provisions of this Section 8, as between Seller and Buyer, Seller will be liable for all Claims to the extent that such arise from the failure of Gas delivered by Seller to meet the quality requirements of Section 5.

SECTION 9. NOTICES

- 9.1. All Transaction Confirmations, Involces, payment instructions, and other communications made pursuant to the Base Contract ("Notices") shall be made to the addresses specified in writing by the respective parties from time to time.
- 9.2. All Notices required hereunder shall be in writing and may be sent by facsimile or mutually acceptable electronic means, a nationally recognized overnight courier service, first class mail or hand delivered.
- 9.3. Notice shall be given when received on a Business Day by the addressee. In the absence of proof of the actual receipt date, the following presumptions will apply. Notices sent by facsimile shall be deemed to have been received upon the sending party's receipt of its facsimile machine's confirmation of successful transmission. If the day on which such facsimile is received is not a Business Day or is after five p.m. on a Business Day, then such facsimile shall be deemed to have been received on the next following Business Day. Notice by overnight mall or courier shall be deemed to have been received on the next Business Day after it was sent or such earlier time as is confirmed by the receiving party. Notice via first class mail shall be considered delivered five Business Days after mailing.
- 9.4. The party receiving a commercially acceptable Notice of change in payment instructions or other payment information shall not be obligated to implement such change until ten Business Days after receipt of such Notice.

SECTION 10. FINANCIAL RESPONSIBILITY

10.1. If either party ("X") has reasonable grounds for insecurity regarding the performance of any obligation under this Contract (whether or not then due) by the other party ("Y") (including, without limitation, the occurrence of a material change in the

creditworthiness of Y or its Guarantor, if applicable), X may demand Adequate Assurance of Performance. "Adequate Assurance of Performance" shall mean sufficient security in the form, amount, for a term, and from an issuer, all as reasonably acceptable to X, including, but not limited to cash, a standby irrevocable letter of credit, a prepayment, a security interest in an asset or guaranty. Y hereby grants to X a continuing first priority security interest in, lien on, and right of setoff against all Adequate Assurance of Performance in the form of cash transferred by Y to X pursuant to this Section 10.1. Upon the return by X to Y of such Adequate Assurance of Performance, the security interest and lien granted hereunder on that Adequate Assurance of Performance shall be released automatically and, to the extent possible, without any further action by either party.

- 10.2. In the event (each an "Event of Default") either party (the "Defaulting Party") or its Guarantor shall: (i) make an assignment or any general arrangement for the benefit of creditors; (ii) file a petition or otherwise commence, authorize, or acquiesce in the commencement of a proceeding or case under any bankruptcy or similar law for the protection of creditors or have such petition filed or proceeding commenced against it; (iii) otherwise become bankrupt or insolvent (however evidenced); (iv) be unable to pay its debts as they fall due; (v) have a receiver, provisional liquidator, conservator, custodian, trustee or other similar official appointed with respect to it or substantially all of its assets; (vi) fall to perform any obligation to the other party with respect to any Credit Support Obligations relating to the Contract; (vii) fail to give Adequate Assurance of Performance under Section 10.1 within 48 hours but at least one Business Day of a written request by the other party; (viii) not have paid any amount due the other party hereunder on or before the second Business Day following written Notice that such payment is due; or ix) be the affected party with respect to any Additional Event of Default; then the other party (the "Non-Defaulting Party") shall have the right, at its sole election, to immediately withhold and/or suspend deliveries or payments upon Notice and/or to terminate and liquidate the transactions under the Contract, in the manner provided in Section 10.3, in addition to any and all other remedies available hereunder.
- 10.3. If an Event of Default has occurred and is continuing, the Non-Defaulting Party shall have the right, by Notice to the Defaulting Party, to designate a Day, no earlier than the Day such Notice is given and no later than 20 Days after such Notice is given, as an early termination date (the "Early Termination Date") for the liquidation and termination pursuant to Section 10.3.1 of all transactions under the Contract, each a "Terminated Transaction". On the Early Termination Date, all transactions will terminate, other than those transactions, if any, that may not be liquidated and terminated under applicable law ("Excluded Transactions"), which Excluded Transactions must be liquidated and terminated as soon thereafter as is legally permissible, and upon termination shall be a Terminated Transaction and be valued consistent with Section 10.3.1 below. With respect to each Excluded Transaction, its actual termination date shall be the Early Termination Date for purposes of Section 10.3.1.

The parties have selected either "Early Termination Damages Apply" or "Early Termination Damages Do Not Apply" as indicated on the Base Contract.

Early Termination Damages Apply:

10.3.1. As of the Early Termination Date, the Non-Defaulting Party shall determine, in good faith and in a commercially reasonable manner, (i) the amount owed (whether or not then due) by each party with respect to all Gas delivered and received between the parties under Terminated Transactions and Excluded Transactions on and before the Early Termination Date and all other applicable charges relating to such deliveries and receipts (including without limitation any amounts owed under Section 3.2), for which payment has not yet been made by the party that owes such payment under this Contract and (ii) the Market Value, as defined below, of each Terminated Transaction. The Non-Defaulting Party shall (x) liquidate and accelerate each Terminated Transaction at its Market Value, so that each amount equal to the difference between such Market Value and the Contract Value, as defined below, of such Terminated Transaction(s) shall be due to the Buyer under the Terminated Transaction(s) if such Market Value exceeds the Contract Value and to the Seller if the opposite is the case; and (y) where appropriate, discount each amount then due under clause (x) above to present value in a commercially reasonable manner as of the Early Termination Date (to take account of the period between the date of liquidation and the date on which such amount would have otherwise been due pursuant to the relevant Terminated Transactions).

For purposes of this Section 10.3.1, "Contract Value" means the amount of Gas remaining to be delivered or purchased under a transaction multiplied by the Contract Price, and "Market Value" means the amount of Gas remaining to be delivered or purchased under a transaction multiplied by the Mon-Defaulting Party in a commercially reasonable manner. To ascertain the Market Value, the Non-Defaulting Party may consider, among other valuations, any or all of the settlement prices of NYMEX Gas futures contracts, quotations from leading dealers in energy swap contracts or physical gas trading markets, similar sales or purchases and any other bona fide third-party offers, all adjusted for the length of the term and differences in transportation costs. A party shall not be required to enter into a replacement transaction(s) in order to determine the Market Value. Any extension(s) of the term of a transaction to which parties are not bound as of the Early Termination Date (including but not limited to "evergreen provisions") shall not be considered in determining Contract Values and Market Values. For the avoidance of doubt, any option pursuant to which one party has the right to extend the term of a transaction shall be considered in determining Contract Values and Market Values. The rate of interest used in calculating net present value shall be determined by the Non-Defaulting Party in a commercially reasonable manner.

Early Termination Damages Do Not Apply:

10.3.1. As of the Early Termination Date, the Non-Defaulting Party shall determine, in good faith and in a commercially reasonable manner, the amount owed (whether or not then due) by each party with respect to all Gas delivered and received between the parties under Terminated Transactions and Excluded Transactions on and before the Early Termination Date and all other applicable charges relating to such deliveries and receipts (including without limitation any amounts owed under Section 3.2), for which payment has not yet been made by the party that owes such payment under this Contract.

The parties have selected either "Other Agreement Setoffs Apply" or "Other Agreement Setoffs Do Not Apply" as indicated on the Base Contract.

Other Agreement Setoffs Apply:

Bliateral Setoff Option:

10.3.2. The Non-Defaulting Party shall net or aggregate, as appropriate, any and all amounts owing between the parties under Section 10.3.1, so that all such amounts are netted or aggregated to a single liquidated amount payable by one party to the other (the "Net Settlement Amount"). At its sole option and without prior Notice to the Defaulting Party, the Non-Defaulting Party is hereby authorized to setoff any Net Settlement Amount against (i) any margin or other collateral held by a party in connection with any Credit Support Obligation relating to the Contract; and (ii) any amount(s) (including any excess cash margin or excess cash collateral) owed or held by the party that is entitled to the Net Settlement Amount under any other agreement or arrangement between the parties.

Triangular Setoff Option:

10.3.2. The Non-Defaulting Party shall net or aggregate, as appropriate, any and all amounts owing between the parties under Section 10.3.1, so that all such amounts are netted or aggregated to a single liquidated amount payable by one party to the other (the "Net Settlement Amount"). At its sole option, and without prior Notice to the Defaulting Party, the Non-Defaulting Party is hereby authorized to setoff (i) any Net Settlement Amount against any margin or other collateral held by a party in connection with any Credit Support Obligation relating to the Contract; (ii) any Net Settlement Amount against any amount(s) (including any excess cash margin or excess cash collateral) owed by or to a party under any other agreement or arrangement between the parties; (iii) any Net Settlement Amount owed to the Non-Defaulting Party against any amount(s) (including any excess cash margin or excess cash collateral) owed by the Non-Defaulting Party or its Affiliates to the Defaulting Party under any other agreement or arrangement; (iv) any Net Settlement Amount owed to the Defaulting Party against any amount(s) (including any excess cash margin or excess cash collateral) owed by the Defaulting Party to the Non-Defaulting Party against any amount(s) (including any excess cash margin or excess cash collateral) owed by the Defaulting Party or its Affiliates under any other agreement or arrangement; and/or (v) any Net Settlement Amount owed to the Defaulting Party or its Affiliates to the Non-Defaulting Party under any other agreement or arrangement.

Other Agreement Setoffs Do Not Apply:

- 10.3.2. The Non-Defaulting Party shall net or aggregate, as appropriate, any and all amounts owing between the parties under Section 10.3.1, so that all such amounts are netted or aggregated to a single liquidated amount payable by one party to the other (the "Net Settlement Amount"). At its sole option and without prior Notice to the Defaulting Party, the Non-Defaulting Party may setoff any Net Settlement Amount against any margin or other collateral held by a party in connection with any Credit Support Obligation relating to the Contract.
- 10.3.3. If any obligation that is to be included in any netting, aggregation or setoff pursuant to Section 10.3.2 is unascertained, the Non-Defaulting Party may in good faith estimate that obligation and net, aggregate or setoff, as applicable, in respect of the estimate, subject to the Non-Defaulting Party accounting to the Defaulting Party when the obligation is ascertained. Any amount not then due which is included in any netting, aggregation or setoff pursuant to Section 10.3.2 shall be discounted to net present value in a commercially reasonable manner determined by the Non-Defaulting Party.
- 10.4. As soon as practicable after a liquidation, Notice shall be given by the Non-Defaulting Party to the Defaulting Party of the Net Settlement Amount, and whether the Net Settlement Amount is due to or due from the Non-Defaulting Party. The Notice shall include a written statement explaining in reasonable detail the calculation of the Net Settlement Amount, provided that failure to give such Notice shall not affect the validity or enforceability of the liquidation or give rise to any claim by the Defaulting Party against the Non-Defaulting Party. The Net Settlement Amount as well as any setoffs applied against such amount pursuant to Section 10.3.2, shall be paid by the close of business on the second Business Day following such Notice, which date shall not be earlier than the Early Termination Date. Interest on any unpaid portion of the Net Settlement Amount as adjusted by setoffs, shall accrue from the date due until the date of payment at a rate equal to the lower of (i) the then-effective prime rate of interest published under "Money Rates" by The Wall Street Journal, plus two percent per annum; or (ii) the maximum applicable lawful interest rate.
- 10.5. The parties agree that the transactions hereunder constitute a "forward contract" within the meaning of the United States Bankruptcy Code and that Buyer and Seller are each "forward contract merchants" within the meaning of the United States Bankruptcy Code.
- 10.6. The Non-Defaulting Party's remedies under this Section 10 are the sole and exclusive remedies of the Non-Defaulting Party with respect to the occurrence of any Early Termination Date. Each party reserves to itself all other rights, setoffs, counterclaims and other defenses that it is or may be entitled to arising from the Contract.
- 10.7. With respect to this Section 10, if the parties have executed a separate netting agreement with close-out netting provisions, the terms and conditions therein shall prevail to the extent inconsistent herewith.

SECTION 11. FORCE MAJEURE

- 11.1. Except with regard to a party's obligation to make payment(s) due under Section 7, Section 10.4, and Imbalance Charges under Section 4, neither party shall be liable to the other for fallure to perform a Firm obligation, to the extent such failure was caused by Force Majeure. The term "Force Majeure" as employed herein means any cause not reasonably within the control of the party claiming suspension, as further defined in Section 11.2.
- 11,2. Force Majeure shall include, but not be limited to, the following: (i) physical events such as acts of God, landslides, lightning, earthquakes, fires, storms or storm warnings, such as hurricanes, which result in evacuation of the affected area, floods, washouts, explosions, breakage or accident or necessity of repairs to machinery or equipment or lines of pipe; (ii) weather related events affecting an entire geographic region, such as low temperatures which cause freezing or fallure of wells or lines of pipe;

- (iii) Interruption and/or curtailment of Firm transportation and/or storage by Transporters; (iv) acts of others such as strikes, lockouts or other industrial disturbances, riots, sabotage, insurrections or wars, or acts of terror; and (v) governmental actions such as necessity for compliance with any court order, law, statute, ordinance, regulation, or policy having the effect of law promulgated by a governmental authority having jurisdiction. Seller and Buyer shall make reasonable efforts to avoid the adverse impacts of a Force Majeure and to resolve the event or occurrence once it has occurred in order to resume performance.
- 11.3. Neither party shall be entitled to the benefit of the provisions of Force Majeure to the extent performance is affected by any or all of the following circumstances: (i) the curtailment of interruptible or secondary Firm transportation unless primary, in-path, Firm transportation is also curtailed; (ii) the party claiming excuse failed to remedy the condition and to resume the performance of such covenants or obligations with reasonable dispatch; or (iii) economic hardship, to include, without limitation, Seller's ability to sell Gas at a higher or more advantageous price than the Contract Price, Buyer's ability to purchase Gas at a lower or more advantageous price than the Contract Price, or a regulatory agency disallowing, in whole or in part, the pass through of costs resulting from this Contract; (iv) the loss of Buyer's market(s) or Buyer's inability to use or resell Gas purchased hereunder, except, in either case, as provided in Section 11.2; or (v) the loss or failure of Seller's gas supply or depletion of reserves, except, in either case, as provided in Section 11.2. The party claiming Force Majeure shall not be excused from its responsibility for Imbalance Charges.
- 11.4. Notwithstanding anything to the contrary herein, the parties agree that the settlement of strikes, lockouts or other industrial disturbances shall be within the sole discretion of the party experiencing such disturbance.
- 11.5. The party whose performance is prevented by Force Majeure must provide Notice to the other party. Initial Notice may be given orally; however, written Notice with reasonably full particulars of the event or occurrence is required as soon as reasonably possible. Upon providing written Notice of Force Majeure to the other party, the affected party will be relieved of its obligation, from the onset of the Force Majeure event, to make or accept delivery of Gas, as applicable, to the extent and for the duration of Force Majeure, and neither party shall be deemed to have failed in such obligations to the other during such occurrence or event.
- 11.6. Notwithstanding Sections 11.2 and 11.3, the parties may agree to alternative Force Majeure provisions in a Transaction Confirmation executed in writing by both parties.

SECTION 12. TERM

This Contract may be terminated on 30 Day's written Notice, but shall remain in effect until the expiration of the latest Delivery Period of any transaction(s). The rights of either party pursuant to Section 7.6, Section 10, Section 13, the obligations to make payment hereunder, and the obligation of either party to indemnify the other, pursuant hereto shall survive the termination of the Base Contract or any transaction.

SECTION 13. LIMITATIONS

FOR BREACH OF ANY PROVISION FOR WHICH AN EXPRESS REMEDY OR MEASURE OF DAMAGES IS PROVIDED, SUCH EXPRESS REMEDY OR MEASURE OF DAMAGES SHALL BE THE SOLE AND EXCLUSIVE REMEDY. A PARTY'S LIABILITY HEREUNDER SHALL BE LIMITED AS SET FORTH IN SUCH PROVISION, AND ALL OTHER REMEDIES OR DAMAGES AT LAW OR IN EQUITY ARE WAIVED. IF NO REMEDY OR MEASURE OF DAMAGES IS EXPRESSLY PROVIDED HEREIN OR IN A TRANSACTION, A PARTY'S LIABILITY SHALL BE LIMITED TO DIRECT ACTUAL DAMAGES ONLY. SUCH DIRECT ACTUAL DAMAGES SHALL BE THE SOLE AND EXCLUSIVE REMEDY, AND ALL OTHER REMEDIES OR DAMAGES AT LAW OR IN EQUITY ARE WAIVED. UNLESS EXPRESSLY HEREIN PROVIDED, NEITHER PARTY SHALL BE LIABLE FOR CONSEQUENTIAL, INCIDENTAL, PUNITIVE, EXEMPLARY OR INDIRECT DAMAGES, LOST PROFITS OR OTHER BUSINESS INTERRUPTION DAMAGES, BY STATUTE, IN TORT OR CONTRACT, UNDER ANY INDEMNITY PROVISION OR OTHERWISE. IT IS THE INTENT OF THE PARTIES THAT THE LIMITATIONS HEREIN IMPOSED ON REMEDIES AND THE MEASURE OF DAMAGES BE WITHOUT REGARD TO THE CAUSE OR CAUSES RELATED THERETO, INCLUDING THE NEGLIGENCE OF ANY PARTY, WHETHER SUCH NEGLIGENCE BE SOLE, JOINT OR CONCURRENT, OR ACTIVE OR PASSIVE. TO THE EXTENT ANY DAMAGES REQUIRED TO BE PAID HEREUNDER ARE LIQUIDATED, THE PARTIES ACKNOWLEDGE THAT THE DAMAGES ARE DIFFICULT OR IMPOSSIBLE TO DETERMINE, OR OTHERWISE OBTAINING AN ADEQUATE REMEDY IS INCONVENIENT AND THE DAMAGES CALCULATED HEREUNDER CONSTITUTE A REASONABLE APPROXIMATION OF THE HARM OR LOSS.

SECTION 14. MARKET DISRUPTION

If a Market Disruption Event has occurred then the parties shall negotiate in good faith to agree on a replacement price for the Floating Price (or on a method for determining a replacement price for the Floating Price) for the affected Day, and if the parties have not so agreed on or before the second Business Day following the affected Day then the replacement price for the Floating Price shall be determined within the next two following Business Days with each party obtaining, in good faith and from non-affiliated market participants in the relevant market, two quotes for prices of Gas for the affected Day of a similar quality and quantity in the geographical location closest in proximity to the Delivery Point and averaging the four quotes. If either party fails to provide two quotes then the average of the other party's two quotes shall determine the replacement price for the Floating Price. "Floating Price" means the price or a factor of the price agreed to in the transaction as being based upon a specified index. "Market Disruption Event" means, with respect to an index specified for a transaction, any of the following events: (a) the failure of the index of announce or publish information necessary for determining the Floating Price; (b) the failure of trading to commence or the permanent discontinuation or material suspension of trading on the exchange or market acting as the index; (c) the temporary or permanent discontinuance or unavailability of the index; (d) the temporary or permanent closing of any exchange acting as the index; or (e) both parties agree that a material change in the formula for or the method of determining the Floating Price has occurred. For the purposes of the calculation of a replacement price for the Floating Price, all numbers shall be increased by one and if the fourth decimal number is less than five, then the third decimal number shall remain unchanged.

SECTION 15. MISCELLANEOUS

- 15.1. This Contract shall be binding upon and inure to the benefit of the successors, assigns, personal representatives, and heirs of the respective parties hereto, and the covenants, conditions, rights and obligations of this Contract shall run for the full term of this Contract. No assignment of this Contract, in whole or in part, will be made without the prior written consent of the non-assigning party (and shall not relieve the assigning party from liability hereunder), which consent will not be unreasonably withheld or delayed; provided, either party may (i) transfer, sell, pledge, encumber, or assign this Contract or the accounts, revenues, or proceeds hereof in connection with any financing or other financial arrangements, or (ii) transfer its interest to any parent or Affiliate by assignment, merger or otherwise without the prior approval of the other party. Upon any such assignment, transfer and assumption, the transferor shall remain principally liable for and shall not be relieved of or discharged from any obligations hereunder.
- 15.2. If any provision in this Contract is determined to be invalid, void or unenforceable by any court having jurisdiction, such determination shall not invalidate, void, or make unenforceable any other provision, agreement or covenant of this Contract.
- 15.3. No waiver of any breach of this Contract shall be held to be a waiver of any other or subsequent breach.
- 15.4. This Contract sets forth all understandings between the parties respecting each transaction subject hereto, and any prior contracts, understandings and representations, whether oral or written, relating to such transactions are merged into and superseded by this Contract and any effective transaction(s). This Contract may be amended only by a writing executed by both parties.
- 15.5. The interpretation and performance of this Contract shall be governed by the laws of the jurisdiction as indicated on the Base Contract, excluding, however, any conflict of laws rule which would apply the law of another jurisdiction.
- 15.6. This Contract and all provisions herein will be subject to all applicable and valid statutes, rules, orders and regulations of any governmental authority having jurisdiction over the parties, their facilities, or Gas supply, this Contract or transaction or any provisions thereof.
- 15.7. There is no third party beneficiary to this Contract.
- 15.8. Each party to this Contract represents and warrants that it has full and complete authority to enter into and perform this Contract. Each person who executes this Contract on behalf of either party represents and warrants that it has full and complete authority to do so and that such party will be bound thereby.
- 15.9. The headings and subheadings contained in this Contract are used solely for convenience and do not constitute a part of this Contract between the parties and shall not be used to construe or interpret the provisions of this Contract.
- 15.10. Unless the parties have elected on the Base Contract not to make this Section 15.10 applicable to this Contract, neither party shall disclose directly or indirectly without the prior written consent of the other party the terms of any transaction to a third party (other than the employees, lenders, royalty owners, counsel, accountants and other agents of the party, or prospective purchasers of all or substantially all of a party's assets or of any rights under this Contract, provided such persons shall have agreed to keep such terms confidential) except (i) in order to comply with any applicable law, order, regulation, or exchange rule, (ii) to the extent necessary for the enforcement of this Contract, (iii) to the extent necessary to implement any transaction, (iv) to the extent necessary to comply with a regulatory agency's reporting requirements including but not limited to gas cost recovery proceedings; or (v) to the extent such information is delivered to such third party for the sole purpose of calculating a published index. Each party shall notify the other party of any proceeding of which it is aware which may result in disclosure of the transaction (other than as permitted hereunder) and use reasonable efforts to prevent or limit the disclosure. The existence of this Contract is not subject to this confidentially obligation. Subject to Section 13, the parties shall be entitled to all remedies available at law or in equity to enforce, or seek relief in connection with this confidentially obligation. The terms of any transaction hereunder shall be kept confidential by the parties hereto for one year from the expiration of the transaction.

In the event that disclosure is required by a governmental body or applicable law, the party subject to such requirement may disclose the material terms of this Contract to the extent so required, but shall promptly notify the other party, prior to disclosure, and shall cooperate (consistent with the disclosing party's legal obligations) with the other party's efforts to obtain protective orders or similar restraints with respect to such disclosure at the expense of the other party.

- 15.11. The parties may agree to dispute resolution procedures in Special Provisions attached to the Base Contract or in a Transaction Confirmation executed in writing by both parties
- 15.12. Any original executed Base Contract, Transaction Confirmation or other related document may be digitally copied, photocopied, or stored on computer tapes and disks (the "Imaged Agreement"). The Imaged Agreement, if introduced as evidence on paper, the Transaction Confirmation, if introduced as evidence in automated facsimile form, the recording, if introduced as evidence in its original form, and all computer records of the foregoing, if introduced as evidence in printed format, in any judicial, arbitration, mediation or administrative proceedings will be admissible as between the parties to the same extent and under the same conditions as other business records originated and maintained in documentary form. Neither Party shall object to the admissibility of the recording, the Transaction Confirmation, or the Imaged Agreement on the basis that such were not originated or maintained in documentary form. However, nothing herein shall be construed as a waiver of any other objection to the admissibility of such evidence.

DISCLAIMER: The purposes of this Contract are to facilitate trade, avoid misunderstandings and make more definite the terms of contracts of purchase and sale of natural gas. Further, NAESB does not mandate the use of this Contract by any party. NAESB DISCLAIMS AND EXCLUDES, AND ANY USER OF THIS CONTRACT ACKNOWLEDGES AND AGREES TO NAESB'S DISCLAIMER OF, ANY AND ALL WARRANTIES, CONDITIONS OR REPRESENTATIONS, EXPRESS OR IMPLIED, ORAL OR WRITTEN, WITH RESPECT TO THIS CONTRACT OR ANY PART THEREOF, INCLUDING ANY AND ALL IMPLIED WARRANTIES OR CONDITIONS OF TITLE, NON-INFRINGEMENT, MERCHANTABILITY, OR FITNESS OR SUITABILITY FOR ANY PARTICULAR PURPOSE (WHETHER OR NOT NAESB KNOWS, HAS REASON TO KNOW, HAS BEEN ADVISED, OR IS OTHERWISE IN FACT AWARE OF ANY SUCH PURPOSE), WHETHER ALLEGED TO ARISE BY LAW, BY REASON OF CUSTOM OR USAGE IN THE TRADE, OR BY COURSE OF DEALING. EACH USER OF THIS CONTRACT ALSO AGREES THAT UNDER NO CIRCUMSTANCES WILL NAESB BE LIABLE FOR ANY DIRECT, SPECIAL, INCIDENTAL, EXEMPLARY, PUNITIVE OR CONSEQUENTIAL DAMAGES ARISING OUT OF ANY USE OF THIS CONTRACT.

EXHIBIT A - FORM OF TRANSACTION CONFIRMATION*

				•	
*The actual Transaction Confirmation entered into between Buyer and CESNG may differ from this form to reflect state regulatory and other local requirements and terms applicable to the Facility. This Form of Transaction Confirmation is for illustrative purposes only and CESNG reserves the right to alter its form of Transaction Confirmation at any time.					
This Transaction Confirmation is delive	ered nursuant to and in	accord	ance with a	natural gas supply agreement effective	
20 (the "Mas	ter Agreement"), by and	between	n Constellati	on Energy Services - Natural Gas, LLC	
("CESNG") and ("Bu	ver"), and is subject to	and mad	de part of the	he terms and conditions of such Master	
Agreement, Capitalized terms used here	in but not defined will have	e the me	anings ascrib	ped to them in the Master Agreement.	
,					
Trade Date:					
	•				
Facility Name:					
Dollyany Darlad	•			•	
Delivery Period:					
Deal Type:					
Nature of Parties' Obligation:				•	
Contract Quantity/Price:					
		44		C	
Month/Yr	Contract Quantity	y (in MM	iBtu)	Contract Price	
		••••			
Delivery Point(s):					
DONALL TOWNS					
<u>Utility</u> :					
-					
				the second of the Bookley	
This Transaction Confirmation docum	ents a Transaction previ	ously re	eached by a	uthorized representatives of the Parties. ansaction unless disputed by Customer	
in writing within two (2) business days	epted as the definitive e. of CESNG's execution o	xpressi date.	on or me m	ansaction unless disputed by Odstoline	
is willing within two (2) additions and	, -10110 t ontonion ,				
CONSTELLATION ENERGY SERVICES	- NATURAL GAS, LLC	BUYER			
Ву:		Ву:			
Name:				EXECUTION:	
Title					
Date:		Date:			
•	•				
				•	

EXHIBIT A

TRANSACTION CONFIRMATION

Page 1 of 2

This Transaction Confirmation dated May 4, 2016 is made a part of, and pursuant to terms of, the Base Contract for Sale and Purchase of Gas dated May 4, 2016 between Buyer and Seller identified herein (the "NAESB").

SELLER:

Constellation Energy Services - Natural Gas, LLC

BUYER:

Milwaukee Board of School Directors on behalf of Milwaukee Public Schools

ACCOUNTS:

□ Check here if Attachment A is attached and made a part of this Transaction Confirmation.

1716 Lawrence Drive PO Box 6025 De Pere, WI 54115-6025

De Pere, WI 54115-6025
Attn: Account Management
E-mail: amwi@constellation.com
Phone: 888-367-4403 (Option 2)

Phone: 888-367-440 Fax: 443-213-6116 Attn: John Linn

1124 N. 11th St.

Mliwaukee WI 53233

E-mail: linnja@mllwaukee.k12.wi.us

Phone: 414-283-4703 Fax: 414-283-4682

Agent / Assoc: Julie Cowling

Account Executive:

Delivery Period: July 1, 2016 to June 30, 2019; provided however, the Delivery Period shall begin on the later of, (i) the day after the first meter read during the period noted above, or (ii) upon acceptance by the utility of Seller as Buyer's gas supplier.

Contract Quantity (in dekatherms): see table

Г		0	Month	Quantity	Month	Quantity	Month	Quantity
L	<u>Month</u>	Quantity	MOULT	MINGINITA	WALL CO.	 		00 700
ſ	Jul	5,600	Oct	46,000	Jan	165,000	Apr	66,700
ŀ	Aug	5,100	Nov	95,800	Feb	149,000	May	28,000
ł	709	9.500	Dec	162,000	· Mar	117,000	Jun	9,000

Price: NYMEX Last Day Settle ("NYM LDS") plus Market Basis per dekatherm. Either or both of the price components may be fixed for specific quantities and delivery months with mutual agreement of the Parties. The Basis is fixed at \$0.755 per dekatherm.

Performance Obligation: Up to 7,300 dkthm/day shall be Firm. The remainder shall be interruptible

Delivery Point: Buyer's Transporter Citygate

Buyer's Transporter: WE Energles

- SPECIAL PROVISIONS -

Balancing Services.

1. A Balancing Service charge of \$0.03 per dekatherm applies in addition to the Price.

Buyer and Seller shall use commercially reasonable efforts to provide each other with sufficient notice to allow timely and accurate nominations in accordance with the nomination requirements of that Transporter, as they may change from time to time due to Constraint Day requirements or otherwise. Buyer accepts the following responsibilities: Buyer hereby appoints Seller as its agent for the purposes of balancing, including the receipt of current and historic Usage, storage, billing and transportation data from Transporter. Buyer agrees to assist Seller in the timely collection of data directly and through Transporter, notify Seller of usage expectations and any material changes to those usage expectations, and comply with any Daily Limitations concerning use. If Usage data is unavailable electronically, Buyer shall obtain usage data manually and deliver that information weekly via facsimile or electronic mail, or as requested by Seller.

- 2. Seller agrees to monitor and review all information provided to the Seller by Buyer and Transporter regarding Buyer's gas Use. Based on such information, Seller shall use commercially reasonable effort to keep Buyer within the belancing parameters established by Transporter. Seller shall pass through to Buyer and Buyer shall pay for any Imbalance Charges and/or Cashout resulting from Buyer's failure to (i) fulfill its obligations set forth in paragraph 1 above, (ii) maintain telemetry, and/or (iii) meet Daily Limitations with respect to Use.
- 3. Mid-month variations to delivered quantity. In the event that the quantity of natural gas that can be delivered to Buyer (without the Transporter assessing Imbalance Charges or Cashout) is less than the quantity agreed to be purchased for the relevant period, Buyer shall remain liable for the entire quantity it agreed to purchase, however, Seller will provide a credit at prevailing market rates for that portion of the purchased quantity that is not delivered. In the event that Buyer's expected Use will exceed the quantity of gas purchased by Buyer and/or available to Buyer without incurring imbalance Charges for the relevant period, Seller will use commercially reasonable efforts to provide incremental supply, plus Fuel (if applicable) at prevailing market rates. Credits for reductions in delivered quantity and/or charges for incremental supply shall appear as separate line items on Seller's invoice to Buyer, including applicable price and quantity.

EXHIBIT A

TRANSACTION CONFIRMATION

Page 2 of 2

4. Telemetry Obligations. Buyer acknowledges and agrees that it shall be responsible for installing and maintaining a meter telephone line. Buyer shall also be responsible for any charges imposed by a contractor for repairing the meter telephone line. As a convenience to Buyer, Seller may arrange for telemetry service or repairs to be provided and Buyer agrees it shall be responsible for any costs associated with such service and repairs.

"Constraint Day" means any period of time during which a Transporter issues balancing orders, operational flow orders, operational matching orders (or the equivalent), or winter splits or deems the day a Critical Gas Day (or the equivalent) and thereby restricts deliveries by Seller or use by Buyer, requires a specific delivery by Seller or use by Buyer, or otherwise requires an affirmative action by either Parly due to adverse operational conditions experienced by the Transporter.

"Daily Limitations" means Transporter enforced restrictions or mandates concerning quantities of gas delivered and/or used, including those imposed due to Constraint Day conditions and/or under ordinary operating conditions.

- 5. Invoices. Invoices shall be prepared based upon quantities used.
- 6. Gonsolidated Billing: Seller shall include the Transporter distribution fees on the monthly invoice. Past due and other charges due Transporter, that apply to service prior to the effective date of the Confirmation will not be included in Seller's invoice to Buyer. Seller may discontinue this service at any time with notice to Buyer.
- 7. Term: This Transaction Confirmation is effective for the Delivery Period as it may be extended or renewed as described below.
- 8. Extending Delivery Period: Should the Parties agree to a fixed price for any pricing component for a delivery month (or billing cycle, as applicable) that occurs after the current Delivery Period, then the Delivery Period of this Transaction Confirmation shall be extended to the latest occurring delivery month (or billing cycle, as applicable) with a fixed price or fixed price component.
- 9. Renewal: The Parties agree that this Transaction Confirmation shall automatically renew for successive twelve (12) month periods (each constituting a "Renewal Delivery Period") unless written notice pursuant to Section 9 of the Agreement terminating the Transaction Confirmation at the end of a Delivery Period Is given by either Party thirty (30) days prior to the end of the Delivery Period. The Contract Quantity during any Renewal Delivery Period shall be the Contract Quantity stated in the table above for the respective month. The Price for the Contract Quantity during any Renewal Delivery Period shall be NYM LDS plus Market Basis.

All other terms and conditions of this Transaction Confirmation shall remain unchanged.

Special Condition: This Confirmation is not effective and binding upon the parties hereto unless signed concurrently with the Base Contract for Sale and Purchase of Gas (NAESB) dated effective May 4, 2016

Seller: Constellation Energy Services - Natural Gas, LLC

Buyer: Milwaukee Board of School Directors on behalf of Milwaukee Public Schools

Signature:

By: David T Dona+

Title: Vice fresident Retail operations

Date: With 116

Signature:

Buyer: Milwaukee Board of School Directors on behalf of Milwaukee Public Schools

Signature:

By: Darienne B. Driver, Ed.D.

Title: Superintendent of Schools

Date: B/24/20/6

Signature:

By: Mark A. Sain

Title: President

Date: C-25-16

RETAIL SPECIAL PROVISIONS

TO THE BASE CONTRACT FOR SALE AND PURCHASE OF NATURAL GAS (2008) BETWEEN CONSTELLATION ENERGY SERVICES - NATURAL GAS, LLC ("CESNG" or "Seller") AND MILWAUKEE BOARD OF SCHOOL DIRECTORS ON BEHALF OF MILWAUKEE PUBLIC SCHOOLS ("Buyer") **DATED MAY 4, 2016**

The following special provisions to the Base Contract and General Terms and Conditions shall supplement and form part of the Base Contract and General Terms and Conditions between the parties. In the event of any conflict or inconsistency between such special provisions and the Base Contract or General Terms and Conditions, the special provisions shall govern.

SECTION 1. PURPOSE AND PROCEDURES

Section 1.1 is deleted in its entirety and replaced with the following:

- These General Terms and Conditions are intended to facilitate purchase and sale transactions of Gas (each a "Transaction") for one or more of Buyer's accounts or facilities (each a "Facility") on a Firm, Secondary Firm or Interruptible basis. The terms of any Transaction between CESNG and Customer shall be as set forth on the applicable TC, but CESNG will procure gas from its suppliers on terms CESNG deems appropriate. The parties may further agree to implement certain strategies or related services, the specifics of which will be set forth in a rider executed by the parties ("Rider"). The entire agreement between the parties shall be the Contract as defined in Section 2.9.
- Section 1.2 "Oral Transaction Procedure" is amended by adding the phrase "or other electronic means of communication" immediately after the phrase "EDI transmission" in the second sentence.
- The last sentence of Section 1.3 is deleted in its entirety and replaced with the following:

"In the event of a conflict among the terms of (i) a binding Transaction Confirmation pursuant to Section 1.2, (ii) a Rider, (iii) the oral agreement of the parties which may be evidenced by a recorded conversation, where the parties have selected the Oral Transaction Procedure of the Base Contract, (iv) the Base Contract, and (v) these General Terms and Conditions, the terms of the documents shall govern in the priority listed in this sentence."

The following shall be added to Section 1:

1.5 Price Locks. Unless otherwise set forth in a Transaction Confirmation or a Rider, Buyer can request to lock in the commodity price for any month(s) at any time during the Delivery Period, prior to 12:00 p.m. EST/EDT on the final day of NYMEX last day settlement for each applicable delivery month. Commodity purchase/sale prices exclude pipeline and utility distribution charges. Buyer also has the right to lock basis at a fixed price and at predetermined volumes. Basis includes interstate and intrastate pipeline transportation but does not include the commodity cost or the utility distribution

SECTION 2. DEFINITIONS

Section 2.9 is deleted in its entirety and replaced with the following:

- 2.9 "Contract" shall mean the legally-binding relationship established by (i) the Milwaukee Board of School Directors Professional Services Contract, (ii) the Base Contract, (iii) any and all binding Transaction Confirmations, (iv) any and all Riders, and (v) where the parties have selected the Oral Transaction Procedure in Section 1.2 of the Base Contract, any and all transactions that the parties have entered into through an EDI transmission or by telephone, but that have not been confirmed in a binding Transaction Confirmation.
- 2.10 Section 2.10 "Contract Price" shall be amended by adding the following at the end of the section: "The Contract Price includes reimbursement to Seller for any production, severance, ad valorem or other Taxes owed with respect to Gas prior to delivery to Buyer, all of which Taxes shall be borne and paid exclusively by Seller." 2.12 Section 2.12 "Cover Standard" is amended by deleting the words "(or an alternate fuel if elected by Buyer and replacement Gas is not available)" in the third line of Section 2.12 ("Cover Standard").

Section 2.16 is deleted in its entirety and replaced with the following:

- "Delivery Point" shall mean city gate interconnection between the Buyer's utility and the upstream Transporter or such other delivery point(s) as are agreed in a Transaction. However, to the extent that Buyer owned transportation capacity is used to deliver Gas to the Buyer, the Delivery Point shall be the receipt point of such capacity.
- Section 2.21 "Guarantor" shall be deleted in its entirety and replaced with the following: "Guarantor shall mean any entity or individual that has provided a guaranty or surety of the obligations of a party hereunder.
- Section 2.32 "Transaction Confirmation" shall be amended by adding the following at the end of the section: "The term "Transaction Confirmation" shall have the same meaning as "TC".
- 2.35 Section 2.35 "Transporter" is amended by adding the word ", storage," after the word "gathering" on the first line of Section 2.35 and adding the words "or storing" after the word "transporting" on the second line of Section 2.35.

The following sections shall be added to Section 2:

- "Costs" means, with respect to the Non-Defaulting Party, brokerage fees, commissions and other similar transaction costs and expenses reasonably incurred by such party as a result of the Event of Default except for such amounts already included in the Net Settlement Amount.
- "Market Price" is a price comprised of (i) the spot commodity cost of Gas as determined by CESNG in its reasonable discretion, plus (ii) all related Interstate and Intrastate pipeline charges required to deliver Gas to the Delivery Point, plus (ili) a reasonable market based margin. Market Price does not include the cost of utility imposed charges, including but not limited to utility or pipeline balancing charges unless otherwise agreed upon.
- "Secondary Firm" means deliveries and receipts of Gas under this Contract will be on a best-efforts basis up to Buyer's maximum daily quantity as set forth in a Transaction Confirmation, if any. Seller may interrupt performance without liability to the extent that one or more of the following conditions are present: (i) Force Majeure, (ii) curtailment by Buyer's local distribution company, (iii) curtailment of supply by a Gas supplier, (iv) curtailment of storage by a storage provider, (v) curtailment of transportation by a Transporter, including, but not limited to, transportation between secondary firm points, (vi) recall of transportation capacity release by its releaser, (vii) curtailment of Gas production behind a specific meter. If Seller Interrupts for any other reason, Buyer's exclusive remedy shall be that Buyer may recover damages as provided in Section 3.2 of the Base Contract.

SECTION 3. PERFORMANCE OBLIGATION

- Section 3.1 shall be amended by inserting *, Secondary Firm, * before "Firm" in the second sentence. 3.1
- Section 3.2 shall be amended by inserting " or Secondary Firm" after "Firm" in the first sentence, 3.2

SECTION 6. TAXES

Section 6 is deleted in its entirety and replaced with the following:

6. Buyer shall pay all federal, state, municipal and local taxes, dulies, fees, levies, premiums or other charges imposed by any governmental authority, directly or indirectly, on or with respect to the Gas and related products and services provided under the Contract, including without limitation any production, severance or ad valorent taxes, and including any taxes enacted after the Effective Date (collectively, "Taxes"). The term "Taxes" shall include any amounts imposed on Buyer directly or on CESNG in its function as Buyer's supplier, and that are associated with the supply of Gas to Buyer (in which case the Buyer shall be responsible to relimburse CESNG for such amounts). If Buyer is exempt from any Taxes, Buyer shall provide CESNG with any state and/or local examption cartificate prior to the legal date of Buyer's first invoice.

All Taxes invoiced to Buyer under this Contract will be with any state and/or local exemption certificate prior to the issue date of Buyer's first invoice. All Taxes invoiced to Buyer under this Contract will be included on the invoice or in the applicable fixed price as allowed by law.

SECTION 7. BILLING, PAYMENT AND AUDIT

Section 7.1 is deleted in its entirety and replaced with the following:

- 7.1 Buyer will be billed for natural gas usage and related products and services supplied under the Contract in one of the following ways based on 7.1 Buyer will be billed for natural gas usage and related products and services supplied under the Contract in one of the following ways based on availability and eligibility of each Facility, which may change from time to time: (i) Dual Billing: Buyer will receive two invoices, one from CESNG for the natural gas supply and one from the utility for the amounts payable by Buyer for services provided by the utility ("Delivery Charges"); (ii) CESNG Consolidated Billing: Buyer will receive one invoice from CESNG that includes both the natural gas supply charges and the Delivery Charges. In the case of CESNG Consolidated Billing, Buyer agrees that (i) Buyer remains exclusively liable to the utility for all Delivery Charges, (ii) CESNG has no obligation to review Delivery Charges for accuracy, (iii) should Buyer dispute Delivery Charges, that is a matter for Buyer and the utility to resolve without any involvement or obligation on the part of CESNG, and (iv) CESNG may withhold any payments due to the utility if Buyer falls to pay CESNG invoices in accordance with this Contract. If the actual quantity of Gas delivered is not known by the invoice date. Invoices will be prepared based on estimated in accordance with this Contract. If the actual quantity of Gas delivered is not known by the invoice date, invoices will be prepared based on estimated meter data or other estimated information. The invoiced quantity will then be adjusted to the actual quantity on the following Month's invoice or as soon thereafter as actual delivery information is available. Section 7.2 is deleted in its entirety and replaced with the following:
- 7.2 All amounts set forth in an invoice are payable to the party issuing the invoice and will include, in addition to the natural gas supply charges, Delivery Charges, and Taxes, all other amounts related to the purchase and delivery of natural gas. CESNG's invoices will be sent to Buyer in accordance with CESNG's normal billing cycle, as adjusted from time to time. CESNG's invoices are due and payable on the Payment Date set forth on the cover page of the Base Contract, or such other date as set forth in a Rider, without offset or reduction of any kind, to the address set forth on the Invoice; provided that if the Payment Date is not a Business Day, payment is due on the next Business Day following that date.
- Section 7.4 shall be deleted in its entirely.

SECTION 8. TITLE, WARRANTY, AND INDEMNITY

The last sentence of Section 8.1 is deleted in its entirely and replaced with the following:

"Buyer shall have responsibility for and assume any liability with respect to said Gas at and after its delivery to Buyer at the Delivery Point(s)."

SECTION 10. FINANCIAL RESPONSIBILITY

Section 10.1 will be deleted in its entirety and replaced with the following:

If CESNG has reasonable grounds: (i) to believe that Buyer's creditworthiness has become unsatisfactory; or (ii) for insecurity with respect to Buyer's performance under the Agreement, CESNG may demand, in writing, adequate assurance of future performance from Buyer in a form, in an amount, from an issuer, and for a term, all as reasonably satisfactory to CESNG ("Adequate Assurance of Performance"). To satisfy a amount, from an issuer, and for a term, all as reasonably satisfactory to CESNG ("Adequate Assurance of Performance"). To satisfy a amount, from an issuer, and for a term, all as reasonably satisfactory to CESNG ("Adequate Assurance of Performance"). If at any time CESNG requires Buyer to prepay for gas, then (i) CESNG shall be under no obligation to deliver gas if Buyer falls to pay any prepayment by its due date, and (iii) in the event the aggregate cost of any quantities of gas required by Buyer in a month in excess of the scheduled nominated quantitles exceeds \$15,000.00, Buyer shall pay CESNG the aggregate cost of such quantitles within three (3) Business Days of CESNG's request for

The following sections shall be amended as follows:

Section 10.2 shall be amended by defeting the word "or" before "(ix)" in such Section; and adding the following immediately after the ";" in

"(x) make any representation or warranty herein which is false or misleading in any material respect when made or when deemed made or repeated; (xi) fall to perform any material covenant or obligation set forth in this Contract (except to the extent such fallure constitutes a separate Event of Default, and except for such party's obligations to deliver or receive Gas (the exclusive remedy for which is provided in Section 3)) if such failure is not remedied within five (5) Business Days after receipt of written notice thereof; (xil) or consolidate or amalgamate with, or merge with or into, or transfer all or substantially all of its assets to, another entity and, at the time of such consolidation, amalgamation, merger or transfer, the resulting, surviving or transferee entity fails to assume all of the obligations of such party under the Contract; then"

Section 10.2 shall further be amended by adding the following immediately before the "." at the end of such Section:

"provided, that no suspension shall continue for more than ten (10) Business Days unless an Early Termination Date has been declared and the Defaulting Party has been given notice thereof in accordance with Section 10.3."

10.3.1 The following sentence shall be added at the end of Section 10.3.1: "In calculating its early termination damages hereunder, the Non-Defaulting Parly may include its Costs."

SECTION 11. FORCE MAJEURE

Section 11.6 shall be deleted in its entirety.

SECTION 13. LIMITATIONS

Section 13 shall be deleted in its entirety and replaced with the following:

THIS SECTION DOES NOT WAIVE ANY LIMITATION ON INDEMNITY OR DAMAGES IMPOSED ON BUYER OR SELLER BY WISCONSIN STATE THIS SECTION DOES NOT WAIVE ANY LIMITATION ON INDEMNITY OR DAMAGES IMPOSED ON BUYER OR SELLER BY WISCONSIN STATE STATUTES. NOTWITHSTANDING ANY OTHER PROVISION OF THE CONTRACT TO THE CONTRARY, THE ENTIRE LIABILITY OF EACH PARTY FOR ANY AND ALL CLAIMS WILL BE LIMITED TO DIRECT ACTUAL DAMAGES ONLY, SUBJECT IN ALL CASES TO AN AFFIRMATIVE OBLIGATION ON THE PART OF EACH PARTY TO MITIGATE ITS DAMAGES, AND NEITHER PARTY WILL BE LIABLE FOR ANY CONSEQUENTIAL, EXEMPLARY, SPECIAL, INCIDENTAL OR PUNITIVE DAMAGES, INCLUDING, WITHOUT LIMITATION, LOST OPPORTUNITIES OR LOST PROFITS NOT CONTEMPLATED BY SECTION 10 ABOVE. The rights, powers, remedies and privileges provided in this Contract are cumulative and not exclusive of any rights, powers, remedies and privileges provided by law. Buyer acknowledges and agrees that the utility is exclusively responsible for the gas distribution and delivery system, that CESNG has no independent control over their systems and will have no liability for any of their sets or emissions. for any of their acts or omissions.

SECTION 15. MISCELLANEOUS

The following sections shall be amended as follows:

Section 15.4 shall be deteted in its entirety and replaced with the following:

- 15.4 Supersedes Prior Agreements. The Contract embodies the parties' entire agreement and understanding and may not be contradicted by any prior or contemporaneous oral or written agreement. Notwithstanding any provision to the contrary in a prior agreement, the parties agree that this Base Contract shall supersede and replace all prior written and oral agreements or arrangements between the parties with respect to the purchase and sale of Gas and related services and that all transactions under any such prior agreement(s) are, as of the Effective Date, now governed solely by the terms of this Base Contract and shall be Transactions hereunder and a part of the single integrated agreement between the parties. This Contract may be amended only by a writing executed by both parties.
- 15.8 The following shall be added to the end of Section 15.8:

"On the effective date and the date of entering into each transaction, each party represents and warrants that: (i) It is duly organized, validly operating and in good standing under the laws of the jurisdiction of its formation; (ii) execution, delivery and performance of the Contract are duly authorized and do not violate any governing documents or any of its contracts or any applicable Law; (iii) It has the knowledge and experience to evaluate the merits and risks associated with the Contract; (iv) It is an "eligible contract participant" as that term is defined in the Commodity Exchange Act, as arrended; (v) this Contract, each transaction hereunder, and each other document executed and delivered in accordance with this Contract constitutes a legally valid support provider, if applicable, is not bankrupt and there are no proceedings pending or being contemplated by it, its credit support provider, if any, or, to its knowledge, threatened against it which would result in it being or becoming bankrupt and there is not pending or, to its knowledge, threatened against it, or its credit support provider, if any, any legal proceedings that could materially adversely affect its ability to perform its obligations under this Contract and each transaction hereunder; and (vii) no Event of Default with respect to it has occurred and is continuing and no such event or circumstance would occur as a result of its entering into or performing its obligations under this Contract and each transaction hereunder. The parties agree that this Contract constitutes a "qualified financial contract" as that term is defined in N.Y.G.O.L. §5-701(b) and a "master netting agreement" as defined in 11 U.S.C. 101(38A) or any successor provisions.

To the extent, if any, that a transaction does not qualify as a "first sale" as defined by the Natural Gas Act and §§ 2 and 601 of the Natural Gas Policy Act, each Party irrevocably waives its rights, including its rights under §§ 4-5 of the Natural Gas Act, unilaterally to seek or support a change to any terms and conditions of the Contract, including but not limited to the rate(s), charges, or classifications set forth therein. By this provision, each Party expressly waives its right to seek or support, either directly or indirectly, and by whatever means: (i) an order from the U.S. Federal Energy Regulatory Commission ("FERC") seeking to change any of the terms and conditions of the Contract agreed to by the Parties; and (ii) any refund from the other Party with respect to the Contract. Each Party further agrees that this waiver and covenant shall be binding upon it notwithstanding any regulatory or market changes that may occur after the date of the Contract or any transaction entered into between the Partles. Absent the agreement of both Partles to the proposed change, the standard of review for changes to any terms and conditions of the Contract proposed by (a) a Party, to the extent that the waiver set forth in this Section 15.8 is unenforceable or ineffective as to such Party due to a final determination being made under applicable law that precludes the Party from waiving its rights to seek or support changes from the FERC to the terms and conditions of this Contract, (b) a non-party, or (c) the FERC acting sua sponte, shall solely be the "public interest" application of the "just and reasonable" standard of review set forth in United Gas Pipe Line Co. v. Mobile Gas Service Corp., 350 U.S. 332 (1956) and Federal Power Commission v. Sierra Pacific Power Co., 350 U.S. 348 (1956) (the "Mobile-Sierra Doctrine"), as the MobileSierra Doctrine has been clarified by Morgan Stanley Capital Group, Inc. v. Public Uiil. Dist. No. 1 of Snohomish 128 S.Ct. 2733 (2008).

Each party represents to the other party (with such representations deemed to be repeated by each party on each date on which a relevant transaction is entered into) that, with respect to any commodity option transaction, it is: (1) a producer, processor, or commercial user of, or a merchant handling the commodity that is the subject of the option transaction; (2) offering or has been offered or is entering into the transaction solely for purposes related to its business as such; and (3) intends to make or take, as applicable, physical delivery of the commodity if the option is exercised.

Further, Buyer warrants, represents and covenants that: (a) the data given and representations made concerning its Facility(ies) are true and correct; (b) it is entering into this Contract to purchase its Gas requirements only and not for speculative or resale purposes, and that the natural gas purchased under this Contract will be consumed at the Facility(ies); (c) if any transaction hereunder gives Buyer the right to adjust the Contract Quantity, or to require CESNG to provide some other quantity of Gas, Buyer's election to exercise such right is based predominantly on supply and demand factors related to Buyer's business; (d) it is the party of record of the Facility(ies), or if it is not the party of record, it has the authority to enter into and blnd its principal to the Contract; and (e) if Buyer is a Governmental Entity, it will not claim immunity on the grounds of sovereignly or similar grounds from enforcement of the Contract. For purposes of this Section, defenses identified in Wis. Stats. §§ 893.80 and 895.46 are not considered similar defenses to sovereign immunity. "Governmental Entity" means a municipality, county, governmental board, governmental department, commission, agency, bureau, administrative body, joint action agency, court or other similar political subdivision (including a public school district or special purpose district or authority), or public entity or instrumentality of the United States or one or more states."

- 15.10 Section 15.10 is amended by adding the words "and such party's affiliates" after the word "party" in the third line of such Section and by adding the parenthetical "(excluding the name of the other party)" after the word "information" in the sub-section (v).
- 15.11 Section 15.11 shall be deleted in its entirety and replaced with the following:

This Contract, and the rights and duties of the parties arising therefrom, shall be governed by, and interpreted and construed in accordance with, the law of the jurisdiction identified on the cover page of the Base Contract (without reference to choice of law doctrine). EACH PARTY HEREBY IRREVOCABLY WAIVES ANY AND ALL RIGHT TO TRIAL BY JURY

IN ANY PROCEEDING ARISING OUT OF OR RELATING TO THIS CONTRACT.

The following sections shall be added to Section 15:

- 15.13 Information and Authorization. Buyer shall designate CESNG to the utility and any other applicable Transporter(s) as an authorized recipient of Buyer's current and historical Gas billing and usage data. Buyer authorizes CESNG to take such actions it deems necessary to enroll the Buyer's accounts with the Utility as accounts to be served by CESNG solely for fulfilling CESNG's obligations under this Contract, including executing on Buyer's behalf any documents necessary to effectuate any account enrollment or election, undertaking the management of any storage or transportation capacity allocated to Buyer by the utility or other Transporter(s), disposing of storage balances, adding or deleting accounts as necessary, receiving usage nominations from the utility relating to Buyer's Gas requirements and other similar documents. Buyer shall take any actions and execute any documents as CESNG shall reasonably request in this regard.
- 15.14 Change in Law. CESNG may pass through or allocate, as the case may be, to Buyer any increase or decrease in CESNG's costs related to the natural gas and related products and services sold to Buyer that results from the implementation of new, or changes (including changes to transportation rates) to existing, Laws, or other requirements or changes in administration or interpretation of Laws or other requirements. "Law" means any law, rule, regulation, ordinance, statute, judicial decision, administrative order, Transporter business practices or protocol, Transporter tariff, or rule of any commission or agency with jurisdiction in the state in which the Facilities are located. Such adjusted amounts will be included in subsequent invoices to Buyer. 15.15 Transporter Restrictions. If either Party receives an operational flow order or any other usage or operating instructions, restrictions or similar notice (each an "OFO") from a Transporter requiring action to be taken in connection with the flow and/or consumption of gas covered by this Agreement, such party will use commercially reasonable efforts to notify the other party of such event in a timely

manner. Such notice may be given by electronic means. Each Parly will take all commercially reasonable actions required by the OFO within the time prescribed. If OFO penalties, charges, fees, costs or expenses result from the actions or inactions of one Party, then such Party shall be solely responsible for any such penalties, charges, fees, costs or expenses. Both parties agree that OFOs may require one or both parties to buy or sell Gas quantities in the then-current market conditions, which may be appreciably higher or lower than the original pricing. If CEGS is responsible for nominations and balancing on a Transporter imposing an OFO or similar restriction, CEGS may (but is not required to) increase or decrease nominations, as appropriate, to avoid penalties.

IN WITNESS WHEREOF, the parties hereto have executed these Special Provisions as of the Effective Date.

NL MAD By:	Dond	J	Oona	<u> </u>
Name:	David	T	Donat	

CONSTELLATION ENERGY SERVICES - NATURAL GAS, LLC

MILWAUKEE BOARD OF SCHOOL DIRECTORS ON BEHALF OF MILWAUKEE PUBLICATIONS

Name: Darienne B. Driver, Ed D

Title: Superintendent of Schools

Name: Mark A. Sain, President

Title: Milwaukee Board of School Directors

Client#: 82241

EXELON

ACORD.

CERTIFICATE OF LIABILITY INSURANCE

DATE (MMIDDIYYYY)

6/03/2016

THIS CERTIFICATE IS ISSUED AS A MATTER OF INFORMATION ONLY AND CONFERS NO RIGHTS UPON THE CERTIFICATE HOLDER. THIS CERTIFICATE DOES NOT AFFIRMATIVELY OR NEGATIVELY AMEND, EXTEND OR ALTER THE COVERAGE AFFORDED BY THE POLICIES BELOW. THIS CERTIFICATE OF INSURANCE DOES NOT CONSTITUTE A CONTRACT BETWEEN THE ISSUING INSURER(S), AUTHORIZED REPRESENTATIVE OR PRODUCER, AND THE CERTIFICATE HOLDER. IMPORTANT: If the certificate holder is an ADDITIONAL INSURED, the policy(les) must be endorsed. If SUBROGATION IS WAIVED, subject to the terms and conditions of the policy, certain policies may require an endorsement. A statement on this certificate does not confer rights to the certificate holder in lieu of such endorsement(s). CONTACT Elizabeth Rodriguez

PHONE
(AC, No, 5xt):

EMAIL
ADDRESS:
erodriguez@mesirowfinancial.com PRODUCER FAX No): 312-595-8169

P&C Chicago Mesirow Insurance Services NAIC# 353 N. Clark Street INSURER(S) AFFORDING COVERAGE INSURER A: American Guarantee & Liability 26247 Chicago, IL 60654 INSURER B: National Fire Ins Co of Hartfor 20478 INSURED Constellation NewEnergy, Inc. INSURER C: 100 Constellation Way INSURER D : Baltimore, MD 21202 INSURER E : INSURER F :

REVISION NUMBER: **CERTIFICATE NUMBER:** COVERAGES THIS IS TO CERTIFY THAT THE POLICIES OF INSURANCE LISTED BELOW HAVE BEEN ISSUED TO THE INSURED NAMED ABOVE FOR THE POLICY PERIOD INDICATED. NOTWITHSTANDING ANY REQUIREMENT, TERM OR CONDITION OF ANY CONTRACT OR OTHER DOCUMENT WITH RESPECT TO WHICH THIS CERTIFICATE MAY BE ISSUED OR MAY PERTAIN, THE INSURANCE AFFORDED BY THE POLICIES DESCRIBED HEREIN IS SUBJECT TO ALL THE TERMS, EXCLUSIONS AND CONDITIONS OF SUCH POLICIES. LIMITS SHOWN MAY HAVE BEEN REDUCED BY PAID CLAIMS. POLICY EFF POLICY EXP LIMITS ADDL SUBR TYPE OF INSURANCE POLICY NUMBER **EACH OCCURRENCE** COMMERCIAL GENERAL LIABILITY DAMAGE TO RENTED PREMISES (Ea occurrence) CLAIMS-MADE OCCUR MED EXP (Any one person) PERSONAL & ADV INJURY GENERAL AGGREGATE GEN'L AGGREGATE LIMIT APPLIES PER: PRODUCTS - COMPJOP AGG POLICY 11/01/2015 11/01/2016 COMBINED SINGLE LIMIT OTHER: \$1,000,000 AUTOMOBILE LIABILITY 5082786753 В BODILY INJURY (Per person) ANY AUTO BODILY INJURY (Per accident) SCHEDULED PROPERTY DAMAGE (Per eccident) AUTOS NON-OWNED Х HIRED AUTOS EACH OCCURRENCE UMBRELLA LIAB OCCUR AGGREGATE **EXCESS LIAB** CLAIMS-MADE RETENTION \$ DED 01/05/2016 01/05/2017 X PERTUTE WORKERS COMPENSATION AND EMPLOYERS' LIABILITY WC931275104 \$1,000,000 E.L. EACH ACCIDENT ANY PROPRIETOR/PARTNER/EXECUTIVE OFFICER/MEMBER EXCLUDED? N EL. DISEASE - EA EMPLOYEE \$1,000,000 (Mandatory in NH) E.L. DISEASE - POLICY LIMIT \$1,000,000 if yes, describe under DESCRIPTION OF OPERATIONS below DESCRIPTION OF OPERATIONS / LOCATIONS / VEHICLES (ACORD 101, Additional Remarks Schedule, may be attached if more space is required) RE: Professional Services Contract.

CERTIFICATE HOLDER	CANCELLATION
Milwaukee Board of School Directors	SHOULD ANY OF THE ABOVE DESCRIBED POLICIES BE CANCELLED BEFORE THE EXPIRATION DATE THEREOF, NOTICE WILL BE DELIVERED IN ACCORDANCE WITH THE POLICY PROVISIONS.
1124 N. 11th St. Milwaukee, WI 53233	AUTHORIZED REPRESENTATIVE
	Dear P. Horamany no

This page has been left blank intentionally.

			7 .
A	ccc)R	'D"
1		ラ ー	

CERTIFICATE OF LIABILITY INSURANCE

DATE (MM/DD/YYYY)

6/3/2016

THIS CERTIFICATE IS ISSUED AS A MATTER OF INFORMATION ONLY AND CONFERS NO RIGHTS UPON THE CERTIFICATE HOLDER. THIS CERTIFICATE DOES NOT AFFIRMATIVELY OR NEGATIVELY AMEND, EXTEND OR ALTER THE COVERAGE AFFORDED BY THE POLICIES BELOW. THIS CERTIFICATE OF INSURANCE DOES NOT CONSTITUTE A CONTRACT BETWEEN THE ISSUING INSURER(S), AUTHORIZED REPRESENTATIVE OR PRODUCER, AND THE CERTIFICATE HOLDER.

IMPORTANT: If the certificate holder is an ADDITIONAL INSURED, the policy(les) must be endorsed. If SUBROGATION IS WAIVED, subject to the terms and conditions of the policy, certain policies may require an endorsement. A statement on this certificate does not confer rights to the certificate holder in lieu of such endorsement(s).

certificate noider in neu or such endorsements	>)-		
PRODUCER		contact NAME: Jenna Hagy	
Mike Hogue		PHONE (AJC, No. Ext):405-235-6633	FAX (A/C, No):405-235-6634
Arthur J. Gallagher Risk Management Services 1230 North Robinson Avenue	, Inc.	E-MAIL ADDRESS:Jenna_Hagy@ajg.com	
Oklahoma City OK 73103-4820		INSURER(S) AFFORDING COVERAGE	NAIC#
onanoma org ott / 0 / 0 / 0 / 0 / 0 / 0 / 0 / 0 / 0 /		INSURER A: Federal Insurance Company	20281
INSURED		INSURER B:	
Constellation Energy Services - Natural Gas, Li	LC	INSURER C:	
clo Exelon Corporation	• • • • • • • • • • • • • • • • • • • •	INSURER D:	
1409 Tangier Drive, Sulte A Baltimore MD 21220		INSURER E:	
Darrillole MD 21220		Insurer f :	
ADVEDAGES CEDICAT	TE MI IMPEDI 4040004407	REVISION NUI	MBER:

VVI	EUVOE9	Z111 3	~~::	= 130 (CED 11) (CED 110)				
IN	IS IS TO CERTIFY THAT THE POLICIE DICATED. NOTWITHSTANDING ANY F IRTIFICATE MAY BE ISSUED OR MAY ICLUSIONS AND CONDITIONS OF SUCI	EQUII PER1	REME TAIN.	NT, TERM OR CONDITION OF AN' THE INSURANCE AFFORDED BY	y contract The policie	OR OTHER I S DESCRIBE	DOCUMENT WITH RESPE D HEREIN IS SUBJECT T	CI TO WHICH THIS
INSR LTR	Type of Insurance		SUBR	POLICY NUMBER	POLICY EFF (MM/DD/YYYY)		LIMI	rs
		1		1	01480046	01410046		P4 000 000

LTR	TYPE OF INSURANCE	INSR	WVD	POLICY NUMBER	(MM/DD/YYYY)	(MM/DD/YYYY)	riwi i	0
A	GENERAL LIABILITY	Y	1112	35816976	8/1/2015	8/1/2016	EACH OCCURRENCE	\$1,000,000
	X COMMERCIAL GENERAL LIABILITY		ļ				DAMAGE TO RENTED PREMISES (Ea occurrence)	\$1,000,000
1	CLAIMS-MADE X OCCUR						MED EXP (Any one person)	\$10,000
							PERSONAL & ADVINJURY	\$1,000,000
							GENERAL AGGREGATE	\$2,000,000
	GEN'L AGGREGATE LIMIT APPLIES PER:		l				PRODUCTS - COMP/OP AGG	\$2,000,000
	X POLICY PRO- LOC							\$
A	AUTOMOBILE LIABILITY	Y		74999277	8/1/2015	8/1/2016	COMBINED SINGLE LIMIT (Ea accident)	\$1,000,000
	X ANY AUTO					,	BODILY (NJURY (Per person)	\$
	ALL OWNED SCHEDULED						BODILY INJURY (Per accident)	\$
	NON-OWNED						PROPERTY DAMAGE (Per accident)	\$.
	HIREO AUTOS AUTOS	ĺ	١.		į			\$
A	X UMBRELLA LIAB X OCCUR	Y	<u> </u>	79834313	8/1/2015	8/1/2016	EACH OCCURRENCE	\$1,000,000
	EXCESS LIAB CLAIMS-MADE						AGGREGATE	\$
	DED RETENTION\$,				s
-	WORKERS COMPENSATION						WC STATU- OTH- TORY LIMITS ER	
	AND EMPLOYERS' LIABILITY ANY PROPRIETOR/PARTNER/EXECUTIVE						E.L. EACH ACCIDENT	\$
	OFFICER/MEMBER EXCLUDED? (Mandatory in NH)	N/A					E.L. DISEASE - EA EMPLOYEE	\$
	If yes, describe under DESCRIPTION OF OPERATIONS below		1		'		E.L. DISEASE - POLICY LIMIT	\$
 	DESOURT HOM OF OF ELECTRONIS BRIDGE							•
1								
					1			

DESCRIPTION OF OPERATIONS / LOCATIONS / VEHICLES (Attach ACORD 101, Additional Remarks Schedule, if more space is required)

RE: Professional Services Contract
Certificate holder is included as Additional Insured on the General Liability, Automobile and Umbrella policies, as per GL endorsement #42-02-0038 (8-89), Auto endorsement #16-02-0292 (4-11), and Umbrella pursuant to and subject to the policy's terms, definitions, conditions and exclusions.

CERTIFICATE HOLDER	CANCELLATION
Milwaukee Board of School Directors 1124 N. 11th St Milwaukee WI 53233	SHOULD ANY OF THE ABOVE DESCRIBED POLICIES BE CANCELLED BEFORE THE EXPIRATION DATE THEREOF, NOTICE WILL BE DELIVERED IN ACCORDANCE WITH THE POLICY PROVISIONS.
·	AUTHORIZED REPRESENTATIVE

© 1988-2010 ACORD CORPORATION. All rights reserved.



CERTIFICATE OF LIABILITY INSURANCE

DATE (MM/OD/YYYY) 06/04/2016

THIS CERTIFICATE IS ISSUED AS A MATTER OF INFORMATION ONLY AND CONFERS NO RIGHTS UPON THE CERTIFICATE HOLDER. THIS CERTIFICATE DOES NOT AFFIRMATIVELY OR NEGATIVELY AMEND, EXTEND OR ALTER THE COVERAGE AFFORDED BY THE POLICIES BELOW. THIS CERTIFICATE OF INSURANCE DOES NOT CONSTITUTE A CONTRACT BETWEEN THE ISSUING INSURER(S), AUTHORIZED REPRESENTATIVE OR PRODUCER, AND THE CERTIFICATE HOLDER. IMPORTANT: If the certificate holder is an ADDITIONAL INSURED, the policy(les) must be endorsed. If SUBROGATION IS WAIVED, subject to the terms and conditions of the policy, certain policies may require an endorsement. A statement on this certificate does not confer rights to the certificate holder in lieu of such endorsement(s). CONTACT NAME: PRODUCER Marsh USA Inc. FAX (AIC, No): PHONE (A/C, No, Exi): E-MAIL 1717 Arch Street Philadelphia, PA 19103-2797 ADDRESS: NAIC # INSURER(S) AFFORDING COVERAGE INSURER A : Lexington Insurance Company 19437 S27339-NewEn-PROF-16-17 INSURED INSURER B: Constellation Energy Services-INSURER C: Natural Gas, LLC 124 W Broadway INSURER D Madison, Wi 53716 INSURER E: INSURER F : CLE-005103281-01 **REVISION NUMBER:2** CERTIFICATE NUMBER: COVERAGES THIS IS TO CERTIFY THAT THE POLICIES OF INSURANCE LISTED BELOW HAVE BEEN ISSUED TO THE INSURED NAMED ABOVE FOR THE POLICY PERIOD INDICATED. NOTWITHSTANDING ANY REQUIREMENT, TERM OR CONDITION OF ANY CONTRACT OR OTHER DOCUMENT WITH RESPECT TO WHICH THIS CERTIFICATE MAY BE ISSUED OR MAY PERTAIN, THE INSURANCE AFFORDED BY THE POLICIES DESCRIBED HEREIN IS SUBJECT TO ALL THE TERMS, EXCLUSIONS AND CONDITIONS OF SUCH POLICIES, LIMITS SHOWN MAY HAVE BEEN REDUCED BY PAID CLAIMS. ADDL SUBR LIMITS TYPE OF INSURANCE . POLICY NUMBER EACH OCCURRENCE DAMAGE TO RENTED PREMISES (Ea occurrence) COMMERCIAL GENERAL LIABILITY CLAIMS-MADE OCCUR MED EXP (Any one person) PERSONAL & ADV INJURY GENERAL AGGREGATE GEN'L AGGREGATE LIMIT APPLIES PER: PRODUCTS - COMP/OP AGG PRO-PULICY OTHER: COMBINED SINGLE LIMIT (Ea accident) AUTOMOBILE LIABILITY BODILY INJURY (Per person) OTUA YAA ALL OWNED AUTOS SCHEDULED AUTOS NON-OWNED AUTOS BODILY INJURY (Per accident) PROPERTY DAMAGE (Per accident) \$ HIRED AUTOS UMBRELLA LIAB EACH OCCURRENCE OCCUR AGGREGATE **EXCESS LIAB** CLAIMS-MADE DED RETENTION \$ PER WORKERS COMPENSATION AND EMPLOYERS' LIABILITY E.L. EACH ACCIDENT ANY PROPRIETOR/PARTNER/EXECUTIVE OFFICER/MEMBER EXCLUDED? (Mandatory in NH) N E.L. DISEASE - EA EMPLOYEE \$ If yes, describe under DESCRIPTION OF OPERATIONS below E.L. DISEASE - POLICY LIMIT | \$ 1,000,000 05/19/2017 Limit 023058104 05/19/2016 Professional Liability SIR 200,000 Architects / Engineers DESCRIPTION OF OPERATIONS / LOCATIONS / VEHICLES (ACORD 101, Additional Remarks Schedule, may be attached if more space is required) CANCELLATION CERTIFICATE HOLDER Milwaukee Board of School Directors SHOULD ANY OF THE ABOVE DESCRIBED POLICIES BE CANCELLED BEFORE THE EXPIRATION DATE THEREOF, NOTICE WILL BE DELIVERED IN ACCORDANCE WITH THE POLICY PROVISIONS. 1124 N. 11th SL Milwaukee, WI 53233 AUTHORIZED REPRESENTATIVE Manashi Mukherjee Mariaoni Mulcherfer